# Development of Techniques for the Isolation of Genes Expressed in the Shoot Apex of Sugar Beet (*Beta vulgaris*).

A thesis submitted to the University of Glasgow for the degree of Doctor of Philosophy

By John R. Bray

January 1993 Re-submitted February 1994

Department of Biochemistry

University of Glasgow

© John R.Bray 1993

ProQuest Number: 13833763

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 13833763

Published by ProQuest LLC (2019). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

#### Acknowledgements

I must first thank both Dr Gareth Jenkins of the Department of Biochemistry, the University of Glasgow, for his help and support during this project, and Dr Wolfgang Schuch of the Department of Exploratory Plant Sciences, ICI, Jealotts Hill, for providing financial support and allowing me to use the facilities of his department. I would also like to thank Prof. Richard Cogdell for allowing me to use the facilities of the Botany Department and the S.E.R.C. for providing the funding for this project.

In addition I am indebted to Dr Keith Edwards for his patience and advice in development of the PCR library procedure, to all of the members of lab B4a of this department, to those of the Department of Botany who helped to make the course of this study run smooth despite the many setbacks and to NormanTait for photographic work.

Finally I would like to thank my parents for their constant support and encouragement.

# List of Abbreviations

A 260	Absorbance at a wavelength of 260 nm
A 280	Absorbance at a wavelength of 280 nm
A 600	Absorbance at a wavelength of 600 nm
ABA	Abscisic acid
AMV	Avian myeloblastosis virus
ATT	Aurintricarboxylic acid
BSA	Bovine serum albumin
CCC	2-Chloroethyltrimethylammonium chloride
cDNA	Complementary deoxyribonucleic acid
COR	Cold regulated
cpm	Counts per minute
dATP	2'-Deoxyadenosine 5'-triphosphate
dCTP	2'-Deoxycytidine 5'-triphopsphate
DEPC	Diethyl pyrocarbonate
dGTP	2'-Deoxyguanosine 5'-triphosphate
DN	Day neutral of the propage subploting acid
DNA	Deoxyribonucleic acid
DNP	Day neutral plant
dNTP	2'-Deoxynucleoside 5'-triphosphate
DTT	Dithiothreitol
dTTP	2'-Deoxythymidine 5'-triphosphate
EDTA	Ethylenediaminetetra-acetic acid
5-FU	5-Fluoro uracil

i

GA	Gibberellic acid
GAs	Gibberellins
GAn	Specific numbered giberellin
HEPES	N-2-hydroxyethylpiperazine-N'-2-ethanesulphonic acid
IPTG	Isopropylthio-β-D-galactoside
kb	Kilo base
kbp	Kilo base pairs
kD	Kilo daltons
LD	Long day
LDP	Long day plant
μCi	Microcuries
μg	Microgrammes and Masson and Andreas
μΜ	Micromolar HCI (pH 2.5), 0.015, ediation
M	Molar ni saline cirate
mg	Milligrammes marshale + Electrication
min	Minute made to acid
mm	Millimetres ACI (pH 7.6). 1 mM EDTA (pH8.0) Second
MOPS	3-[N-Morpholino] propane sulphonic acid
mRNA	Messenger ribonucleic acid
NaPP	Sodium pyrophosphate
NB	Night break ministration of the second s
ngoat	Nanogrammes and index of D-saladoode
nm	Nanometres
nM	Nanomolar
PCR	Polymerase chain reaction

Pfu	Plaque forming units
pg	Picogrammes
Poly (A <sup>+</sup> ) RNA	mRNAs carrying polyadenylated sequences at their 3' termini
RNA	Ribonucleic acid
rpm	Revolutions per minute
SD: of Abbreviation	Short day
sddH <sub>2</sub> O	Sterile double distilled water
SDP	Short day plant
SDS	Sodium dodecyl sulphate
sec General lat	Second
SLDP Howering	Short long day plant
SM Fectors Aff	100 mM NaCl, 10 mM MgSO <sub>4</sub> ,
	50 mM Tris-HCl (pH 7.5), 0.01% gelatin
SSC Percention	Standard saline citrate
SSPE Evidence 0	Standard saline phosphate + EDTA
TCA Allomative	Trichloroacetic acid and of Florad School of State
TE con The Effects	10 mM Tris-HCl (pH 7.6), 1 mM EDTA (pH8.0)
Tris Introductio	Tris [hydroxymethyl] aminomethane
2-TU Venditati	2-Thio uracil
v/v Association	Volume per unit volume
w/v Perception	Weight per unit volume
X-Gal doventio Pr	5-bromo-4-chloro-3-indolyl-β-D-galactoside

Evidence from Grafibing Experiments for a selected of Piors' Institution.

17

# Table of Contents

1.7	Events at the Apical Meristern, page	number
Title p	age totacular Events Upon Induction	
Ackno	wledgements	
List of	Abbreviations	i
Table	of Contents	iv
Summ	ary	1
Chapte	er 1: Introduction	4
1.1	General Introduction	4
1.2	Flowering	6
1.3	Factors Affecting Flowering	7
1.3.1	Photoperiod	7
1.3.2	Perception of Photoperiod	9
1.3.3	Evidence from Grafting Experiments for a Model of Photoperiodic Induction	
1.3.4	Alternative Hypotheses for the Control of Floral Induction	12
1.4	The Effects of Temperature	13
1.4.1	Introduction	13
1.4.2	Vernalization	14
1.4.3	Association of Vernalization with Photoperiod	15
1.4.4	Perception of Cold Temperatures	17
1.4.5	Juvenile Phase	18
1.4.6	Reversion to Vegetative Growth	
1.4.7	Evidence from Grafting Experiments for a Model of Floral Initiation	20
	by Low Temperatures	

1.5	Gibberellic Acid	22
1.6	Models for the Combined Effects of Vernalization and Photoperiod	24
1.7	Events at the Apical Meristem	26
1.8	Molecular Events Upon Induction	29
1.8.1	Early Biochemical Studies of Floral Initiation	29
1.8.2	Genetic Control of Flowering	33
1.9	Sugar Beet	
1.9.1	Introduction	. 35
1.9.2	Effects of Photoperiod and Temperature	
1.9.3	Juvenile Phase	39
1.9.4	Genetics of Sugar Beet Annuality	40
1.9.5	Events at the Apex	41
1.9.6	Mechanisms Involved in the Regulation of Vernalization	
1.9.7	Grafting Experiments	44
1.10	AimsDNA.Framerat	.45
2.13		
Chapter	2: Materials and Methods	
2.1	General Laboratory Methods	49
2.2	Sources of General Laboratory Reagents	49
2.3	Sources of Enzymes and Related Products	49
2.4	Growth and Maintenance of Bacterial and Phage Stocks	50
2.5	E. coli Strains Used in this Study	.51
2.6	Bacteriophage Strain Used in this Study	. 51
2.7	Maintenance of Plants	. 52
2.7.1	Light and Temperature Regimes	. 52

2.7.2	Growth of Plants	52
2.7.3	Sampling of Plant Tissue	52
2.7.4	Microscopy of Apical Tissue	53
2.8	General Molecular Biology Methods	54
2.8.1	Estimation of Nucleic Acid Concentration	54
2.8.2	Estimation of RNA Purity	54
2.8.3	Phenol Extraction	.54
2.8.4	Ethanol Precipitation	55
2.8.5	Agarose Gel Electrophoresis	55
2.8.6	Isolation of DNA from Agarose Gels	56
2.8.7	Denaturing Polyacrylamide Gel Electrophoresis	57
2.9	Preparation and Purification of Oligonucleotides	57
2.10	Oligonucleotides used in this Study	58
2.11	End Labelling of Double Stranded DNA	59
2.12	Radiolabelling of DNA Fragments	. 59
2.13	Non-Radioactive Labelling of DNA	60
2.14	Hybridisation of Radiolabelled Probes	60
2.14.1	Plaque Hybridisation	60
2.14.2	Southern Analysis of DNA Fragments	62
2.14.3	Dot Blot Analysis of DNA Fragments	.62
2.14.4	Northern Analysis of RNA samples	63
2.15	Polymerase Chain Reaction	. 63
2.16	Preparation of Total Plant RNA	65
2.16.1	Large Scale RNA Prep	.65
2.16.2	Small Scale RNA Prep	.67

2.17	Synthesis of Double Stranded cDNA	68
2.18	Preparation of Sugar Beet cDNA LIbraries	71
2.18.1	PCR Amplified cDNA	71
2.18.2	Preparation of Plating Cells	73
2.18.3	Packaging of Cloned cDNA into Bacteriophage Lambda	73
2.18.4	Estimation of Phage Concentration	. 74
2.19	Determination of the status of PCR Amplified cDNA Libraries	.74
2.19.1	Examination of Random Inserts	74
2.19.2	Screening with Specific cDNAs	75
2.19.3	Screening with Non-PCR Amplified cDNA	76
2.20	Isolation of Apex-Specific cDNA Clones	76
2.20.1	Differential Screening	. 76
2.20.2	Screening of 'Cold' Plaques	. 77
2.20.3	Production of a Subtracted Probe	78
3.2.3	Removal of Unincorporated Radiolabelled Nucleotides	
Chapter	3: Results tion of Apical Merideen cDNA After Lightion of Linkers sectors	
3.1	Development of Methods to Produce PCR Amplified cDNA Libraries from	.80
3.2.6	Beta vulgaris	
3.1.1	RNA	80
3.1.2	cDNA Synthesis	81
3.1.3	Fractionation of Leaf cDNA on a Sephadex G-50 Column	84
3.1.4	Development of Techniques for PCR Amplification of cDNA Suitable	
	for Cloning	85
3.1.4.1	Oligonucleotide Linkers	85
3.1.4.2	Ligation of Linkers to Control DNA	87

3.1.4.3	PCR Amplification of $\lambda$ /Hind III Marker DNA, Blunt-Ended	
	and Ligated to JB Linkers	90
3.1.4.4	Ligation of Linkers to the cDNA	92
3.1.5	Development of the Cloning Procedure for PCR Amplified cDNA	95
3.1.5.1	Eco R1 Digestion of the leaf cDNA	95
3.1.5.2	Removal of Small Sized cDNA Prior to Cloning	95
3.1.5.3	Ligation of PCR Amplified cDNA to $\lambda$ ZAP II Phage Vector	98
3.1.6	Characterisation of the Leaf PCR Amplified cDNA Library	98
3.1.6.1	Examination of Random Plaques	. 100
3.1.6.2	Screening of the Leaf Library with Known DNA Probes	102
3.2	Preparation of a PCR Amplified cDNA Library from Shoot Apical Meristem	
	Tissue	110
3.2.1	Isolation of RNA from Annual Beta vulgaris Shoot Apical Meristem Tissue	110
3.2.2	Preparation of Annual Shoot Apical Meristem cDNA	112
3.2.3	Removal of Unincorporated Radiolabelled Nucleotides	. 112
3.2.4	Amplification of Apical Meristem cDNA After Ligation of Linkers	. 116
3.2.5	Removal of Small Sized cDNA Prior to Cloning	118
3.2.6	Amplification of Inserts from the Apical Meristem cDNA LIbrary	118
3.2.7	Screening the Apical Meristem cDNA Library with Known cDNA Probes	122
3.2.8	Screening the PCR Amplified Apical Meristem cDNA Library with Single	
	Stranded Apical Meristem cDNA	127
3.3	Methods of Isolating Apex-Specific cDNA Clones from the Apical Meristem	
	cDNA Library	129
3.3.1	Differential Screening	129
3.3.2	Secondary Screening of Putative Apex-Specific cDNA Clones	130

3.3.3	Investigation of Putative Positive Clones to Establish the Abundance of their	
	Transcripts Within Leaf and Apical Meristem RNA	133
3.3.4	'Cold' Plaque Screening of the Apical Meristem Library	133
3.3.5	Development of Methods for Subtractive Screening of the Apical Meristem	
	cDNA Library	136
3.3.5.1	Incorporation of Biotin Labelled dATP in to PCR Amplified Leaf cDNA	.137
3.3.5.2	Removal of Biotin Labelled Leaf cDNA Using Streptavidin	139
3.3.5.3	Establishing the Number of Rounds of Subtraction Required to Remove the	
	Driver cDNA	141
3.3.5.4	Subtraction of Leaf Driver cDNA with Apical Meristem cDNA	141

Chapte	er 4: Discussion	145
4.1	Development of the PCR Cloning Technique	. 145
4.2	Techniques for the Isolation of Apex-Specific cDNA Clones	.148
4.3	Concluding Remarks	153

References	160
NCICI CIICCS	

induction of flowering. Concrating seed from these plants, and thus maintaining these lines of blennial beens, can be very difficult. To understand, and be able to constrol, flowering in sugar beet crops at the molecular level would, therefore, be of advantage to both the augar beet prower and to the breater.

It is known that the annual beet plant possesses a gene, B, which coolers annuality upon the best plants. This gene shows almont complete Mendelian dominance over the gene for bienniality. Since the biennial requires vernalization to induce theseeing it is possible the

# Summary

Flowering plants are the most complex and highly evolved organisms within the plant kingdom. The process of flowering is regulated by a combination of intrinsic and environmental factors which signal the switch from vegetative to floral development. The molecular mechanisms involved in this process are poorly understood, and although there have been many models put forward to try to explain the way in which the environment interacts with these intrinsic factors to bring about floral induction at the shoot apex, no one model has been shown to satisfy every case.

The sugar beet plant (*Beta vulgaris*) is commercially important due to its ability to store a large concentration of sucrose in its expanded tap root (the beet). *Beta vulgaris* exists in two growth forms, annual and biennial. Both are obligate long day plants (LDP) but the biennial requires an additional period of cold treatment (vernalization) prior to exposure to the long days. A substantial amount of the sugar beet crop is lost each year due to a phenomenon referred to as bolting. Here the biennial beet flowers in its first growing season as a consequence of exposure to low temperature early in that growing season and, therefore, loses its supply of sucrose to the formation of the flower stalk. There are also several biennial genotypes of sugar beet that require very specific environmental growth conditions for the induction of flowering. Generating seed from these plants, and thus maintaining these lines of biennial beets, can be very difficult. To understand, and be able to control, flowering in sugar beet crops at the molecular level would, therefore, be of advantage to both the sugar beet grower and to the breeder.

It is known that the annual beet plant possesses a gene, B, which confers annuality upon the beet plants. This gene shows almost complete Mendelian dominance over the gene for bienniality. Since the biennial requires vernalization to induce flowering it is possible that the vernalization process induces changes within the biennial that allow the expression of this B gene, or some other gene specific to the biennial plant. Alternatively the biennial may not contain a functional B gene product. The process of vernalization is known to be perceived at the shoot apex. Hence, in this study the shoot apices of annual sugar beet plants that had experienced non-floral-inductive conditions (short days, 20<sup>o</sup>C) were isolated to look for genes expressed specifically in the shoot apex, and possibly involved in the process of vernalization.

To do this the Polymerase Chain Reaction (PCR) was employed to amplify cDNA copies of transcripts expressed in such a small amount of tissue. To develop the techniques required for the cloning of PCR amplified cDNA the more abundant leaf RNA was used. Double stranded cDNA was prepared using total RNA, thus eliminating the step of enriching for poly (A<sup>+</sup>) RNA. This did not appear to affect the final quality of the cDNA libraries produced, as a ribosomal DNA clone hybridised at a very low frequency to the Beta vulgaris cDNA clones studied. The screening of the leaf library with a light harvesting complex chlorophyll a/b binding protein (LHCP) cDNA clone demonstrated that the leaf PCR amplified library was of plant tissue, and the LHCP clone was represented at a frequency within the expected order of magnitude. A number (35) of annual apices were isolated and a small amount of RNA extracted from them. This RNA was treated in the same manner as the leaf RNA and a PCR amplified apical meristem cDNA library was successfully prepared from this material. The frequency of LHCP clones within this library was approximately 10 fold less than that of the leaf library. Thus, the apical library did contain clones of plant origin, and the frequency of the LHCP clones was again of the expected order of magnitude. Therefore, the apical library was differentially screened to look for clones representing apex-specific transcripts. On secondary screening a number of clones several putative positives were isolated. However, the most promising of these putative apexspecific clones was shown to be most highly expressed in leaf tissue, as shown by hybridisation to a northern blot. Subsequently, over 200 'cold' plaques were isolated to explore the possibility that the apex-specific mRNAs may be present at too low a frequency to be detected by differential screening. However, none of the 'cold' plaques isolated was seen to be apex-specific on first screening. It was, therefore, decided to attempt to eliminate all cDNA species common to the apex and leaf cDNA populations using a subtraction procedure. This proved a relatively simple procedure to develop, but it did not prove possible to obtain a subtracted probe that contained sequences hybridising only to an apical cDNA probe. Therefore, using this technique it was not possible to isolate apex-specific cDNA clones. There may be several reasons for this result. It is unlikely that the number of cDNA clones screened was too small for the isolation of apex-specific cDNA clones, but this possibility must not be ignored. The amplified apical cDNA may have contained apexspecific cDNA represented at too low a frequency to be detected by the methods used in this study. They may also not have been present in the small sample of cDNA cloned into the vector, and would therefore never have been isolated. Finally it is possible that the apical tissue of Beta vulgaris does not possess any mRNA species specific to itself, and therefore it would never have been possible to isolate a cDNA clone representing such a mRNA from this library.

parental species. Although sexual hybrids have not been pursued to their limits, e.s. African violet, and the minacle rates like and likes (Cocking, 1990), the intervention of more sophisticated genetic engineering techniques whereby novel genetic information is directly introduced into the desired species, does have many advantages over conventional breading programmer. Genetic engineering offers are a direct changes in a shorter period of tens. The genetic information transferred may be totally novel, and as a bares extent the property is soperandom. CHAPTER 1 INTRODUCTION

# 1.1. General Introduction

It has long been the wish of many agronomists to introduce desirable traits into species of economic importance. Since the earliest farming practices began the realisation that man could manipulate species to his own advantage has given agronomists the incentive to break down the barriers between species, and to select for novel combinations of desired characters.

Conventional breeding programmes, though still essential to the improvement of certain crop species, are dogged by factors such as the long generation times of plants, the limited variability of genetic information, and the fact that recombination is random. Hence, the isolation of new varieties requires selection on a large scale. Therefore, one major problem in using interspecific hybridisation in plant breeding programmes is the low probability of obtaining in one individual the desired genetic information from the parental species. Although sexual hybrids have not been pursued to their limits, e.g. African violet, and the miracle rices IR8 and IR36 (Cocking, 1990), the intervention of more sophisticated genetic engineering techniques whereby novel genetic information is directly introduced into the desired species, does have many advantages over conventional breeding programmes. Genetic engineering offers more direct changes in a shorter period of time. The genetic information transferred may be totally novel, and to a large extent the process is non-random.

The introduction of genes for resistance to viruses, insects, and herbicides, has been successfully achieved using the tools of genetic engineering (Newbury *et al.*, 1989, De Block *et al.*, 1987). These advances have been achieved in a much shorter time, and with much greater efficiency than would have been possible using conventional techniques. As more problems are overcome by the use of molecular biology, greater, and more complex challenges are put before the molecular biologist. The science of genetic engineering to date has focussed on the introduction of monogenic traits. The problem faced by the genetic engineer now is one of introducing multigenic traits into new host plants. One example where advances are being made in this area is that of the nitrogen fixation process. Most commonly this process occurs with the association of leguminous plants and *Rhizobium* bacteria. Genetic information from both plant and bacterium is essential for the development of nitrogen fixing root nodules (Sprent, 1989). A degree of success has been obtained in this field (Al-Mallah *et al.*, 1989), but there is still a long way to go before multigenic traits may be introduced into new hosts as easily as single genes.

One important issue for crop improvement is a better understanding of how environmental stresses (such as light, temperature, and humidity) affect the development of plant species (Hughes and Pearce, 1988). Light and temperature in particular, are known to affect the reproductive development of many flowering plant species and, as breeding programmes are hindered in many cases by a long generation time, the ability to control flowering in plant species is of great importance to the breeder. Therefore the study of flowering in plants plays a key role in the acquisition of knowledge important to the improvement of crop species.

5

## 1.2 Flowering

The development of flowering plants depends on meristems, the groups of dividing cells that are the source of new plant structures. Both the root and shoot apical meristems are present in the developing seed as the radicle and plumule respectively. The shoot apical meristem gives rise to a proliferation of vegetative leaf and stem tissue until intrinsic or environmental factors signal the meristem to begin floral development. Although the process of flowering is complex and covered by a large area of research, it can be divided into three major phases:

1) Induction, the exposure to favourable conditions for flowering which result in the production of a floral stimulus.

2) Evocation, synonymous with initiation, and described as the events occurring at the shoot apex as a result of the arrival of the floral stimulus from the leaf, committing it to develop as a floral meristem.

3) Morphogenesis, the development of differentiated floral organs.

The vegetative shoot meristem is organised into several zones containing cells with distinct histological characteristics and fates. By contrast the floral meristem contains only two distinct zones, designated the core and the mantle (Sussex, 1989). A major consequence of floral induction is the loss of vegetative meristem zones, and the concentration of floral meristem cells into the region of the mantle. The molecular and cellular events which occur during this meristematic reorganisation are not clearly understood, and therefore a vast amount of energy is being channelled into research on this subject.

In general floral meristems develop in whorls, or helical rows, from meristematic

cells in a progressive order, i.e. sepals, petals, stamens, and carpels or pistils. Experimental manipulation of the floral meristem has shown that there is a progressive and generally irreversible commitment of the primordia to develop into specific organ systems (Steves and Sussex, 1989). At early stages of flower development the floral meristem is able to give rise to all floral organ systems. However, as flower development proceeds, primordia are committed to differentiate into specific organ systems. The molecular events required for the specification of rows of floral meristem cells into specific organ system primordia are not yet known, but an increasing number of floral homeotic mutants and genes are being isolated and are beginning to provide new insights into this process.

### **1.3 Factors Affecting Flowering**

1.3.1 Photoperiod and species already mentioned are of the pollutate photoperiod

The effects of changes in daylength (photoperiod) upon the flowering of plants have been studied from as early as the middle of the nineteenth century. Sachs (1863, 1866, cited in Murneek and Whyte, 1948) showed an interest in the process of floral induction, but it was not until the early twentieth century that a French botanist by the name of Turnois established the association of the relative period of light and dark with the ability of plants to flower. More extensive studies by Garner and Allard (1920) with Maryland Mammoth variety of tobacco (*Nicotiana tobaccum*) and soybean (*Glycine max*) established the need for daylengths of less than a critical maximum number of hours to allow the development of flowering in these species. Plant species with such a requirement are referred to as short day plants (SDP). Soon after this discovery it became apparent that many plants display a photoperiodic requirement to flower and a number of different categories were formed to accommodate them. Plants requiring more than a critical minimum number of daylight hours to induce flowering (e.g. *Hyoscyamus niger*, *Lolium temulentum*) are referred to as long day plants (LDP). A few species assumed to be SDP were later observed to have a dual photoperiodic requirement in which a period of short days (SD) did not initiate flowering unless it was followed by a period of long days (LD). These species are referred to as short long day plants (SLDP). Conversely it is true that some plant species require exposure to LD prior to exposure to SD for flower development to continue. There are, of course, plant species that will flower irrespective of the photoperiodic conditions. These are known as day neutral plants (DNP). There is a degree of overlap in the photoperiodic requirements of different classes of plants. For example the SDP *Dactylis glomerata* will flower when exposed only to daylengths of 14 h, whereas the LDP *Hyoscyamus niger* will flower in a photoperiod of as little as 12 h.

Most of the plant species already mentioned are of the obligate photoperiodic group of plants, having an absolute requirement for favourable photoperiodic conditions to initiate flowering. Many other plant species will display an enhanced ability to flower under SDs or LDs, but given time they would eventually flower under non-inductive conditions. These are known as facultative photoperiodic plants and include such species as *Digitalis purpurea*, *Dianthus barbatus*, *Cherianthus cheiri*, and *Teucrium scorodonia* (Vince-Prue, 1975). It was soon established that the length of the light period was less critical than the length of the exposure to darkness. For example in the SDP *Pharbitis nil*, induction occurs when a single period of darkness longer than a critical minimum duration is interposed into continuous light, thus establishing that in common with most SDP elapsed time in darkness is the critical factor for floral induction. However, exposure to light is not unnecessary but is an integral part of the overall mechanism.

### 1.3.2 Perception of Photoperiod.

Experimental evidence from several plant species indicates that the daylength requirement of photoperiodically induced plants is perceived by the leaf tissue (Vince-Prue, 1975). For many years the sensitivity of leaves to photoperiod was suggested to increase as the leaves expanded until they reached full size, gradually decreasing as they senesce. Work by Moskov (1936) on Chrysanthemum; Borthwick and Parker (1938 a) on Biloxi soybean; and Naylor (1941) with Xanthium indicated that in these SDPs the first expanded leaf, rather than the mature expanded leaves, showed the greatest sensitivity to photoperiodic induction. This led Khudairi and Hamner (1954) to assess the relative sensitivity of leaves of different ages from the SDP Xanthium to photoperiodic induction. They concluded that the sensitivity of the Xanthium leaves to photoperiodic induction increases with age of the expanding leaves reaching a maximum when they are half expanded. Sensitivity then decreases with an increase in age of the expanding leaves. This viewpoint was reinforced by Salisbury (1955). However, Zeevart (1958) showed that the LDP Perilla retained sensitivity in its leaves for a considerable time after full expansion, and suggested that the position of the leaves on the plant was of greater importance. The LDP Lolium tumulentum may also retain sensitivity within its fully expanded leaves as the response of plants with only their basal leaves remaining was similar to that of young plants with only a few young leaves remaining. This observation overlooks the fact that plant age plays a major role in sensitivity of leaves to photoperiod (Borthwick and Parker 1938a; Moskov 1936). Therefore, in considering the response to

photoperiod of leaves of different ages, it is essential to use plants of the same age and not to introduce new variables which may complicate interpretation of the experiment in hand.

The work of Gressel *et al.* (1980) demonstrated using fibre optics that the apical tissues may also perceive inductive photoperiods. However, the light perceived by the apex is believed only to modify the responsiveness of the apex to the arrival of the floral stimulus. Therefore the apical meristem is not believed to influence the production of the floral stimulus under inductive conditions directly, but to alter its responsiveness to the stimulus produced. Similar results have been obtained by Baldev (1959) with exposed apical buds of *Cuscuta*. It is possible, therefore, that tissue containing rapidly dividing cells plays a major role in the perception of photoperiod. This does not rule out all other tissue types as being able to induce flowering in plant species, and Nitsch (cited in Wareing and Phillips, 1978) has shown stem tissue alone is able to induce flowering in cultured explants of *Plumbago indica*. However, this is an exception to the rule, based on extensive evidence, that young unexpanded leaves are the most sensitive to photoinductive conditions.

1.3.3 Evidence From Grafting Experiments for a Model of Photoperiodic Floral Induction.

It was Sachs who, as early as 1882, is reported to have postulated the occurrence of flower producing substances (Blüstoffe) formed in the leaves and translocated to the meristematic regions (cited in Jaffe *et al.*, 1987). Chailakhyan later (1938) obtained evidence that a floral inductive agent was synthesised in the leaves under favourable photoperiodic conditions. He found that leaves from induced *Perilla* plants grafted onto

uninduced plants caused the latter to flower. The agent was assumed to be hormonal in nature and was termed florigen. Gardner and Loomis (1953) obtained evidence for the presence of a floral inductive hormone with their work on the SDP Dactylis glomerata, and stated that flowering was induced by a chemical or hormonal differentiation in response to favourable conditions. Because an induced SDP will stimulate a non-induced LDP it has been suggested that the floral stimulus is identical in the two types of plant (Zeevart, 1976). This theory of a flower-inducing hormone led to the belief that inhibitors of flowering may exist, and it is the balance of the agonist and antagonist within the shoot apical meristem which determines the status of the plant, either vegetative or floral (Hamner, 1940). Earlier work by Borthwick and Parker (1938b) on some SDPs had also implied that inhibitors exist within the leaves of the plant. They used plants with two branches of nearly equal size, one was exposed to SDs whilst the other was maintained in LDs. The donor branch was reintroduced to LDs and it was observed that the receptor branch did not flower unless it was defoliated. The belief here is that the presence of the leaves on the receptor stalk supplying photosynthates to the shoot apex dilutes the effect of the stimulus arriving from the donor stalk. Withrow et al. (1943) reported that noninduced leaves positioned between the induced leaves and the apex have an inhibitory effect in spinach. Lang et al. (1952) reported that Hyoscyamus niger and Chenopodium amaraticolor will flower in non-inductive conditions if all the leaves are continually removed. Later work by Lang (1965) suggests that an inhibitory graft transmissible agent, termed antiflorigen, is present in a variety of photoperiodic and non-photoperiodic plants, and is not specific in a taxonomic sense as scions from non-induced Nicotiana sylvestris and Hyoscyamus niger are able to inhibit flowering in Nicotiana tobaccum. There is also evidence from work with mutants of pea (Pisum sativum) that genes which code for inhibitors of flowering exist in this species (Reid, 1986). To date, however, nobody has

succeeded in isolating or characterising the proposed hormones, and so alternative models have been suggested by several workers to accommodate the requirements of the control of flowering.

1987), indicates the need for assimilates in the developing tissue prior to floral induction

### 1.3.4 Alternative Hypotheses for the Control of Floral Induction.

It has been suggested that the translocation of such a specific hormone as "florigen" by mass flow with the photosynthetic assimilates would be the most inefficient way of having it arrive at the shoot apex, since very few molecules would ever arrive at their destination (McDaniel, 1985). Considering the many different types of plant responses it seems unlikely that a single mechanism will control flowering in all plants. This has led people to look in other directions for a suitable hypothesis. Evidence from work by Sachs and Hackett (1983) on Xanthium and Nicotiana sylvestris of the need for transport of the photoperiodic stimulus to the roots led to the belief that something more complex was in operation in the stimulation of flowering. Therefore the nutrient diversion hypothesis was put forward postulating that induction of flowering resulted from modifying the source-sink relationship within the plant such that the shoot apical meristem receives a higher concentration of assimilates than under normal conditions. The inhibitory effect of non-induced leaves may be interpreted on the basis that the leaves provide an alternative source of assimilates (Lang, 1965; Zeevart et al., 1977). Non-induced leaves close to the apical bud have been shown to prevent flowering, but removal results in flower formation. The use of <sup>14</sup>CO<sub>2</sub> has shown that assimilates accumulate in the leaves when present (Zeevart, 1976), suggesting that the presence of

the leaves close to the apical meristem may affect the ability of the plant to flower by the retention of necessary assimilates. The fact that starch levels in leaves grown under SDs are 2-3 times higher than under LDs, and that isolated shoot tips or embryos may be induced to flower only when an external supply of sugars is provided (Atherton *et al.*, 1987), indicates the need for assimilates in the developing tissue prior to floral induction.

Application of a variety of chemicals, including plant growth regulators and carbohydrates, promotes flowering. This led Bernier (1988) to propose the multifactorial hypothesis of floral induction. This hypothesis takes into account the fact that several factors are involved in the control of flowering, promotors and inhibitors. Floral induction is believed to take place only when all factors required are present in the shoot apical meristem at the appropriate concentrations and at the right times. Martinez -Zapater and Sommerville (1990) argue that differences in phenotypes of various *Arabidopsis* mutants suggest there are several factors involved in the control of flowering, and the effect of one of these factors can be offset in many cases by the action of others. This is in agreement with the belief of Bernier and his co-workers. Thus, there are several hypotheses available which attempt to explain floral induction, but further experimental evidence is required before it will be clear which is most likely to be correct.

# 1.4 The Effects of Temperature

1.4.1 Introduction

Photoperiodic responses cannot be considered in isolation from temperature,

because it is rare to find a plant in which the daylength response is wholly independent of temperature. Many plants which are day neutral (DN) at one temperature are sensitive to daylength at another. Therefore the effects of temperature on flowering plants have been studied extensively (Chroboczek, 1934; Murneek and Whyte, 1948; Lang, 1965).

## 1.4.2 Vernalization.

The requirement for certain varieties of plants to be exposed to cool temperatures to allow the induction or promotion of flowering was termed by Lysenko as vernalization in 1928 (cited in Murneek and Whyte, 1948). This term originates from the Latin for spring (vernum) and relates to the action of low temperatures converting winter varieties into spring varieties (Chouard, 1960). As early as 1857 the effects of low temperatures upon the initiation of flowering were noted by Kilppart whilst working on winter wheat (cited in Murneek and Whyte, 1948). He stated that to convert winter wheat to spring wheat you need only allow the winter wheat to germinate slightly and maintain it at near freezing temperatures until it is sown out in the spring. The realisation of the importance of the near freezing temperatures to induce flowering, and not just to suspend the plants in a state of readiness to germinate, was not realised until the early twentieth century. Gassner (1918) exposed plants of Pektus winter rye, and Pektus spring rye to temperatures of 1-2°C, 5-6°C, 12°C and 24°C. He noticed that flowering in the spring rye was not influenced by temperature, whereas the winter rye flowered most successfully if it had been exposed to 1-2°C. Flowering at 5-6°C only occurred if the plants were removed from the constraints of the greenhouse at the end of March during the cold of the early spring, and thus experienced continued cold treatment at temperatures below that to which they had been exposed in the greenhouse. Winter rye

plants grown under conditions in which they never perceived temperatures below 12°C showed little, if any, floral induction. Since that time many other species have been observed to require vernalizing temperatures as an obligatory step before flowering can be initiated (e.g. *Beta vulgaris*, *Hordeum vulgare* and *Hyoscyamus niger*).

The process of vernalization is, like photoinduction, not a simple one. It has been observed that some plant varieties which do not have an obligate requirement for vernalization to flower, will flower in a shorter time if they are exposed to vernalizing conditions. Those plants which show an obligate requirement for vernalization to initiate flowering will flower at a faster rate the longer the cold period and the cooler the temperature. Certain species may posses both facultative and obligate vernalization requiring genotypes within them. Some species may be vernalized after a lengthy period of exposure to temperatures as high as 15°C. Above this temperature they will not flower unless other important factors are favourable (e.g. light, water, and nutrients). In general, however, temperatures between 2-10°C are considered best for vernalization as these low temperatures are not low enough to cause freezing damage to the plant tissue but have been shown to be highly successful in the induction of flowering in many plant species, provided an adequate supply of  $O_2$  and carbohydrates is available at the shoot apex (Chouard, 1960). After vernalization is completed the differentiation of flower initials does not generally occur until the plant is later returned to higher temperatures, and in many cases a particular photoperiodic regime (Vince-Prue, 1975).

### 1.4.3 Association of Vernalization with Photoperiod

The cold requirement for flower formation is often linked with a particular photoperiodic condition. Most frequently it is the association of LD with a vernalization

requirement (Vince-Prue, 1975). However, vernalization in LD is ineffective in inducing flower formation unless the LD are perceived during the latter part of the vernalization period (Magara, 1964). The effective temperature for vernalization under these conditions is increased in species such as sugar beet.

Exposure to SD after vernalization will have little or no effect on the vernalized state of the plant unless the SD conditions are maintained for an extensive period of time. Even as many as 200 SD after vernalization of the LDP *Hyoscyamus niger* has no effect on the vernalized state of the plant (Bernier *et al.*, 1981). Therefore, the temperature effect on the growing tip is a self perpetuating physiological change. All the tillers of vernalized winter rye will flower when exposed to the right photoperiodic conditions, even though they were not formed prior to the cold period (Bernier *et al.*, 1981). However, the tillers must have developed from tissue that was exposed to the cold, and they are, therefore, able to respond to inductive daylength conditions by flowering.

The vernalization requirement of many plants may be replaced, or modified, by exposure to non-inductive photoperiodic conditions. SD may substitute partly or wholly for the low temperatures in some LDP. Wellensiek (1958) showed that certain plants became sensitive to LD one month sooner when exposed to SD than they did when exposed to low temperature. Whilst the minimum cold requirement decreased with age, the SD requirement remained constant. Also fewer LD were necessary in LDP after exposure to SD than to cold (Vince-Prue, 1975). Wellensiek (1958) thus concluded that exposure to low temperatures and SDs are different mechanisms of obtaining the same result.

The partial or total replacement of chilling by LD is exemplified only by certain *Chrysanthemum* species, but not by any other plants (Vince-Prue *et al.*, 1975).

### 1.4.4 Perception of Cold Temperatures.

In sugar beet, Chrysanthemum, celery, cauliflower and olive exposure of the shoot apical meristem to low temperatures results in vernalization. Successful reciprocal graft transfers of shoot tips between vernalized and non vernalized Hyoscyamus niger, and Althea rosea, and vernalization of excised shoot tips of carrot and cabbage, demonstrate the shoot apex to be the perceptive tissue (Bernier et al., 1981). Curtis and Chang (1930) demonstrated that cooling coils around the shoot tip of celery induced flowering in these plants when the rest of the plant was exposed to temperatures of 20°C. It is generally accepted, therefore, that most cold-requiring plants perceive low temperatures at the shoot tip (Vince-Prue, 1975). However, chilling of the apical bud of carrot after removal of 4/5's of the root had no effect (Chouard, 1960) and the author suggested that the sugar supply to the stem from the roots seems to play some part in the vernalization process, as has also been suggested by Bernier (1988). Vernalization has been reported in detached leaf tissue of Streptocarpus and Lunaria biennsis (Zeevart, 1963), but later evidence indicated that merisematic tissue at the base of the petiole was responsible. This is suggested to occur because vernalization takes place only when dividing cells are present during cold treatment (Wellensiek, 1964). Apical meristems contain a large proportion of dividing cells, and are therefore likely to give the greatest response to vernalizing temperatures. It is possible, however, that all dividing cells are potential sites for vernalization and experiments with Lunaria show the base of the petiole to be vernalizable (Vince-Prue, 1975). It seems to be the case that meristematic activity is important in the process of vernalization, however winter rye is vernalizable at temperatures of -2°C and below and the same is observed in Cherianthus seeds. At these temperatures mitosis is most likely to be suspended. Therefore, vernalization only seems

to occur in cells able to divide, or those about to undergo cell division, and is not solely confined to the apical meristem.

## 1.4.5 Juvenile Phase

In many plant species there is a requirement for the plants to reach a certain age before they are able to respond to vernalizing temperatures. Juvenility is most often observed in obligate vernalization plants, many of which are biennial in habit (Bernier *et al.*, 1981). This rule, like many associated with flowering, has its exceptions, and in certain plant species vernalization of the germinating seed, or even the developing seed on the mother plant, is possible to a degree. However, the younger the plants are when they receive the cold temperatures, the longer is the time to flowering. In the case of seeds developing on the mother plant, in almost every instance a subsequent period of low temperature is required to vernalize the plants completely, but the cold treatment necessary is reduced in length. Therefore, partial vernalization has been attained (Lexander, 1985).

The apical meristem in certain vernalization requiring species with a juvenile phase is reported to be less active in young plants than in those possessing a more mature apical meristem (Williams and Atherton, 1990). This may be an important factor in the determination of juvenile incompetence to perceive low temperatures and may be due to growth elsewhere on the plant exerting priority demands upon the limiting nutrient supply. Evidence from Williams and Atherton (1990) suggests that inadequate carbohydrate availability for juvenile cauliflower plants is a consequence of allocation of assimilates to the leaf sink. This adds to the evidence stated in section 1.3.4.supporting the theory that depletion of assimilates influences competence to respond to floral inductive conditions.

The duration of the juvenile phase varies from plant to plant, being 10-30 days for *Hyoscyamus niger* (Sakar 1958), and 8 weeks for *Lunaria biensis* (Wellensiek 1958). Sensitivity to cold treatment increases with time after this initial juvenile period up to a threshold (Vince-Prue, 1975). There is no correlation between the size of the seed and the duration of the juvenile phase.

### 1.4.6 Reversion to Vegetative Growth

Sussex (1989) stated that plants that flower in response to inductive photoperiods can usually be transferred to non-inductive conditions once flowering is initiated, where flower formation continues to completion. The implication here is that floral meristems are stably determined. McDaniel et al. (1991) established clearly that the floral stimulus acts within a few hours to make the apex florally determined. The commitment to flowering in plants exposed to inductive daylengths is usually, therefore, only reversible under unusual growth conditions and is dependent on the time at which such unfavourable conditions were experienced. There is extensive evidence that the effect of the vernalizing temperatures experienced by many plants may be reversed if unusually high temperatures are experienced. This process is referred to as devernalization. It was first reported by Gregory and Purvis (1936) where the effect of high temperatures for a few days was to nullify the effect of the cold treatment. For devernalization to occur chilling treatment should be sub-optimal as optimally vernalized species, in general, cannot be devernalized. Winter wheat exposed to temperatures of 35°C for 3 days can be completely devernalized provided the duration of the vernalizing temperatures does not exceed about 12 weeks (Purvis, 1948). Exposing vernalized peas, however, to 35°C for

18 -24 h can completely devernalize them (Chouard, 1960). The temperature at which the peas were vernalized was not reported, however, and so it is possible that vernalization was not optimally achieved. In general heat treatment must immediately follow the chilling treatment. An intervening period of non-devernalizing temperatures stabilises the effect of the chilling (Bernier *et al.*, 1981). One exception to this rule is *Cheiranthus allionii*, which is better devernalized 1-2 months after the chilling period.

Devernalization has also been observed to be caused by CO<sub>2</sub> depletion and SD (Chouard, 1960). SD devernalization is of particular interest as many plants, such as *Beta vulgaris* and *Oenothera* species have demonstrated this phenomenon. The processes involved in SD devernalization and devernalization by high temperatures are probably of different origin. The process of SD devernalization has, however, been shown to be a true photoperiodic response as a night break (NB) has been demonstrated to nullify the effect. Given the range of responses of plant species to temperature a unifying principle underlying devernalization is difficult to obtain (Zeevart, 1963).

# 1.4.7 Evidence from Grafting Experiments for a Model of Floral Initiation by Low Temperatures

The transmissibility of a floral stimulus produced during low temperature treatment has been demonstrated in grafting experiments with a number of related and unrelated families, between different species and genera, and in plants of different response types (Vince-Prue, 1975). It is suggested that the presence of mature leaves, and light, impair the establishment of the necessary sucrose gradient from the donor to the receptor and, as the stimulus is transmitted with the photosynthates, removal of mature leaves or growth in the dark allows the correct flow of stimulus. In the majority of grafting experiments most low temperature graft partners were LDP with or without the low temperature requirement, and experiments were usually carried out under LD conditions. Under these conditions the donor plant would always be competent to flower having first perceived a period of low temperature. Therefore, the receptor plant may have been induced by the production of the correct floral stimulus in the donor plant, and as such the receptor plant would have bypassed the need for vernalization. The belief is, therefore, that vernalization may be a strictly localised state that enables the plant to produce a transmissible substance under favourable conditions at a later stage. The question remains as to whether a transmissible substance specific for vernalization, and distinct from that stimulus produced by photoinduction, is produced after low temperature treatment. Evidence for the existence of such a transmissible substance, termed vernalin (Melchers and Lang, 1948) comes from work with the SDP Nicotiana tobaccum (Maryland mammoth) grafted to a stock of Hyoscyamus niger. The Maryland mammoth, after being exposed to low temperatures, was able to induce flowering in the non-induced stock plant under LD conditions whilst the Maryland mammoth donor remained vegetative. In this case photoinduction of the floral state in the stock was not a possibility as the stock plant has an absolute requirement for low temperature prior to photoinduction. Floral induction has also been observed under SD conditions after grafting annual Hyoscyamus niger donor tissue to unvernalized biennial Hyoscyamus niger stocks, but only after the removal of the donor. Under these conditions flowering took place 12 h later (Vince-Prue, 1975). Only a few plants have, therefore, been shown to be graft inducible to flower under non-floral inductive conditions.

21

### 1.5 Gibberellic Acid

Gibberellins (GAs) are a class of plant hormones which have been shown to induce the flowering of many LDP and cold requiring plants under non-inductive conditions (Zeevart, 1976). Applied GAs are usually ineffective in triggering flowering in SDP but there are a few exceptions such as *Chrysanthemum* (Pharis and King 1985). Nanda *et al.* (1967), however, report that GAs have an inhibitory effect on flowering in some SDP under inductive conditions.

GAs most often stimulate cell elongation. However, there are many different GAs that exist, some of which are more effective at inducing floral development in some plant species than others.  $GA_7$  is active in the majority of plants, but  $GA_3$  has the greatest effect on a small number of plants, including *Beta vulgaris* (Zeevart, 1976).  $GA_3$  was, however, only effective in promoting flowering in *Beta* species at temperatures slightly above the vernalization temperature range, and was ineffective at higher temperatures (Vince-Prue, 1975). This implies that some degree of the low temperature effect may be necessary prior to the application of  $GA_3$  for the initiation of flowering in these plants.  $GA_3$  was also shown to overcome the cold requirement of biennial *Hyoscyamus niger* plants to flower under LD conditions (Chouard, 1960). It was observed, however, that application of  $GA_3$  to *Hyoscyamus niger* induced shooting prior to shooting. Lang (1965) cites many more examples where stem elongation only is observed under non-inductive conditions including *Oenothera biennsis, Scrofularia alta,* and *Scabiosa* varieties. GAs, therefore, do not replace totally the cold requirement of these plants to flower.

The ability of plants to be vernalized as imbibed seed has often been reported (Chouard, 1960). Endogenous applications of GA were generally ineffective at these

early stages. This lack of response of seed-vernalizable plants to treatment with GA, together with the work on *Beta* species and *Hyoscyamus niger*, suggests that GA does not directly control, or substitute for vernalization. It is possible that GA is not a direct product of thermoinduction, but the vernalization process may lead to the synthesis of GA, which may be required for processes leading to flowering. This hypothesis may help to explain why only partial replacement of the cold requirement is observed in many plants.

The endogenous concentration of GAs of plants is often found to increase after low temperature treatment (Wareing and Phillips, 1978). In wheat, it is reported that the amount of GA-like substances appears greater in vernalized than in non-vernalized specimens (Vince-Prue, 1975). The GA was not, however, thought to be involved in floral induction, but only in the elongation response to low temperature. Exogenous applications of extracts from *Xanthium* plants, thought to be GA-like in substance, have been reported to promote flowering under non-inductive conditions (Wareing and Phillips, 1978). It would be wrong to assume, however, that GA is the sole causative agent in floral induction as other factors within the extract may also have acted to stimulate floral induction.

Since GA has no stimulatory effect on the flowering of most SDP Chailakyan (1937) suggested that the causative agent in SDP for floral induction was a separate entity termed "Anthesin". This agent was suggested to be active in limiting floral induction in SDP during LD and GA was suggested to be active in limiting floral induction in LDP during SD. If this were the case then a graft union of non-induced LDP and SDP under non inductive conditions should induce flowering in both plant types due to the inductive effect of the non-induced graft partner. This has been demonstrated not to be the case (Zeevart, 1976). It is not now generally accepted that GA is not the sole flower inducing

agent in all plant species. It is suggested that GAs are probably involved in vernalization and may be required as a precursor to a co-factor, or the synthesis of a hormone (Pharis and King, 1985). This is consistent with most evidence to date which suggests that GAs are only one of many factors necessary to induce flowering in LD and cold requiring plants.

## 1.6 Models for the Combined Affects of Vernalization and Photoperiod

Salisbury (1963) and Lang (1965) proposed that low temperatures could act by suppressing the action of a flowering inhibitor relatively more than the activity of a promoter. This could occur via a differential effect on the synthesis of the promotor/inhibitor substances due to different temperature coefficients of enzymes involved in their synthesis, such that production of the inhibitor is suppressed at lower temperatures. Over the years various models have been constructed to try to account for the observed effects on vernalization of different environmental stimuli. Lang and Melchers (cited in Chouard, 1960) formulated a model (see Fig 1.1) to account for the observations they made on *Hyoscyamus niger*. The model incorporates a precursor (A) used in a reaction at low temperatures to give rise to a thermolabile substance (B) which is lost at high temperatures, but which at intermediate temperatures can be used to form a floral stimulus (C). Later models were fashioned along the same lines but involved more intermediates.

Alternatively the physiological effects of growth at low temperatures, for instance the depression of leaf growth, and associated altered assimilate levels at the shoot apex may result in vernalization. The accumulation of carbohydrates at the shoot apex being a
# Fig 1.1Model to try to Account for the Observed Effects on Vernalization of<br/>Different TemperaturesTaken from Gregory and Purvis, 1936

A is used in a reaction at low temperatures which gives rise to B. B is thermolabile. At high temperatures it is used in a reaction to produce D. At intermediate temperatures it is used to produce the floral stimulus C. requirement for floral meriation forantion could also account for the effect of other factors which alter assimilate availability being able to substitute for low temperatures (e.g. SD, high light intensity, and high pressure) (Williams and Atherton, 1990).

McDaniei (1985) discussed the effects of vernatization in terms of the ormpetence of tissues to respond to inductive conditions. In photoperiodic plants with a vernalization requirement, where daylength is perceived by the leaves, a floral stimulus of some sort is produced in response to the inductive daylength. The lack of response of non-vernalized plants must mean that either the leaves are not competent to respond to

Low temperatures

A



High

Intermediate temperatures



1) the meristern starts producing leaves competent to respond to the

2) an incompetent contemporatures as competent to respond to the floral stimulus. The response type a plan belongs to c De distinguished b) grafting experiments (see Fig 1.2). Fig 1.2(a) shows the accepted scheme for the way in which low temperatures and inductive photoperiods act to bring about floral induction in many flowering plants. Those requiring low temperatures must be exposed to such conditional prior to exposure to the inductive photoperiod for a flowering response to be observed.

К

#### 1.7 Events at the Apical Meristern

One of the major problems in the study of flowering is to discover the underlying cause of the relatively sudden transition of a vegetative growing point from the production of leafy organs to the formation of floral organs. There are several distinct requirement for floral meristem formation could also account for the effect of other factors which alter assimilate availability being able to substitute for low temperatures (e.g. SD, high light intensity, and high pressure) (Williams and Atherton, 1990).

McDaniel (1985) discussed the effects of vernalization in terms of the competence of tissues to respond to inductive conditions. In photoperiodic plants with a vernalization requirement, where daylength is perceived by the leaves, a floral stimulus of some sort is produced in response to the inductive daylength. The lack of response of non-vernalized plants must mean that either the leaves are not competent to respond to the daylength or the apex is not competent to respond to the floral stimulus.

Thus McDaniel proposed that vernalization may affect the meristem in either of two ways:

1) the meristem starts producing leaves competent to respond to the inductive daylength with the production of the floral stimulus, or

2) an incompetent meristem becomes competent to respond to the floral stimulus.

The response type a plant belongs to can be distinguished by grafting experiments (see Fig 1.2). Fig 1.2(a) shows the accepted scheme for the way in which low temperatures and inductive photoperiods act to bring about floral induction in many flowering plants. Those requiring low temperatures must be exposed to such conditions prior to exposure to the inductive photoperiod for a flowering response to be observed.

#### **1.7** Events at the Apical Meristem

One of the major problems in the study of flowering is to discover the underlying cause of the relatively sudden transition of a vegetative growing point from the production of leafy organs to the formation of floral organs. There are several distinct

## Fig 1.2 Model to Explain the Differences Between Different Vernalization Response Types

In a photoperiodic plant which requires vernalization flowering will only occur when both the cold and daylength requirements have been met.

#### a) Distinction Between Cold Requiring and Non-Cold Requiring Genotypes.

Plants not requiring vernalization will flower in an inductive photoperiod without prior cold treatment (upper line). Plants which require vernalization (lower lines) will only flower after cold treatment followed by the appropriate daylength

#### b) Hypothetical Vernalization Response Types

In plants with response type A the cold requirement is eliminated by grafting on a flowering shoot. In this type of plant the unvernalized meristem is evidently able to respond to the floral stimulus translocated from the graft, but the leaves are unable to produce it. In plants with response type B grafting on a flowering shoot does not induce flowering. In this type of plant the unvernalized meristem is unable to respond to the floral stimulus

Chapter 1: Introduction



types of floral development; apices ending in a single flower (e.g. Viscaria) apices terminated by a single inflorescence (e.g. the terminal spikelet of the Triticum aestivum ear), and apices ending in a compound inflorescence (e.g. the Compositae) (Schwabe, 1987). Despite these differences amongst flowering plants many similarities are observed at the apical meristem during the transition to flowering. There is a notable increase in mitotic activity and related cell elongation (Gonthier et al., 1987). These events occur in cells of the central and peripheral zones of the shoot apical meristem (Bernier et al., 1981). The vegetative shoot meristem is organised into several zones containing cells with distinct histological characters and fates. By contrast the floral meristem contains two distinct zones; the inner core and the outer few layers of cells termed the mantle (Sussex, 1989). A major consequence of floral induction is the loss of the vegetative meristem zones, and a concentration of floral meristem cells into the outer mantle region (Drews and Goldberg, 1989). The general shape of the apical meristem changes from being relatively small and pointed, to becoming a larger, more dome-shaped arrangement. Subsequent changes lead not only to the generation of the germline and organs of sexual reproduction, but also to the different spatial relationships of these organs, often differing from the spatial characteristics of the vegetative organs (Meeks-Wagner et al., 1989). In DN plants the transition from a vegetative to a floral apical meristem occurs when the plant is a certain age, or has produced a determined number of internodes. For example the DN variety of tobacco plant, Wisconsin 38, produces approximately 35 internodes before the transition to a flower producing plant. For plants requiring a specific photoperiod to induce flowering the apical meristem will generally remain vegetaive until the correct photoperiod is perceived. Nougaréde et al. (1989) state that under noninductive conditions quantitative photoperiodic plants undergo a transition from vegetative to intermediate before achieving a pre-floral state. They interpret this

intermediate phase as being a preparatory step, and noted that obligate photoperiodic plants generally miss out this intermediary phase. It may be the case that when the correct photoperiod is perceived by such plants they pass through this preparatory step to the prefloral stage.

One effect of low temperature on the apical meristem of vernalization requiring plants is to alter the rate of cell metabolism, and this change is suggested not to be directly related to vernalization (Zeevart, 1976). Most evidence for the role of low temperature on floral induction points to an improvement of the supply of assimilates to the apex. (Bernier *et al.*, 1981; Salisbury, 1963). Under warm non-inductive conditions rapid growth of young leaves could deprive the shoot apices of certain growth factors such as carbohydrates limiting further development of the apex (Williams and Atherton, 1990). Chilling could, therefore, inhibit leaf growth and increase the availability of growth factors to the apex by reducing the requirement for a supply to the leaves. Hussey (1963) showed that there was a decrease in the growth rate of leaf initials in tomatoes at low temperatures whilst there was an increase in dry weight of the apex and accelerated flowering. Atherton *et al.* (1987) indicate that carbohydrates accumulate in the shoot tips of cauliflower during chilling and suggest that suppression of leaf development in mature plants may be essential for vernalization to occur.

## 1.8 Molecular Events Upon Induction

1.8.1 Early Biochemical Studies of Floral Induction

Floral induction and subsequent development is considered to be mediated

(Vince-Prue, 1975) More recent work with mutants of Pitum sativary has demonstrated

through the control of gene expression. The nature of the changes which take place in the transition to the vernalized state is unknown, but may involve the expression of previously inactive genes or the inhibition of previously active genetic information (Vince-Prue, 1975). Tran Thahn Van et al. (1985) give evidence that oligosaccharide fragments from plant cell walls play a role in flowering, although since that time they have been unable to confirm or deny this. Many inhibitors of flowering have been found in the search for that one molecule which may bring about the switch from vegetative to floral development. Actinomycin D is an inhibitor of RNA synthesis which is known to suppress flowering in Pharbitis nil cv violet when applied to the cotyledons prior to a single inductive long night (Arzee et al., 1970). The inhibitor of protein synthesis, cycloheximide, has been shown to inhibit floral induction in Xanthium (Ross, 1970), whilst Zeevart (1962) has demonstrated that the process inhibited by 5- fluro uracil (5-FU) in the bud of Xanthium plants during the first part of the inductive dark period is RNA synthesis. Zeevart also demonstrated that 5-FU inhibited flowering in Pharbitis nil by suppressing DNA replication in the shoot apex. The ability to develop flowers was restored by the addition of thymidine, thymidylic acid, deoxycytidine, and 5-methyl deoxycytidine, all of which are precursors of DNA (Zeevart, 1962). Hess found that 2thio uracil (2-TU) inhibits RNA synthesis and flower formation (cited in Zeevart, 1963). 2-TU incorporated into RNA also inhibits floral initiation in Streptocarpus wendlandii (Vince-Prue, 1975). More recent work with mutants of Pisum sativum has demonstrated that two loci, Sn and Dne, confer the ability of Pisum sativum plants to respond to photoperiod (Murfet and Reid, 1987). Reid (1986) has demonstrated by grafting experiments that the gene combination Sn, Dne acts by regulating the production of an inhibitor which delays flower initiation. Using the evidence from such experiments with inhibitors it was suggested that nucleic acid synthesis was involved in the control of floral induction. Direct observations of differences in protein and nucleic acid synthesis of cells of induced and non-induced plants have shown promising results. Cold treated Arabidopsis of a variety requiring vernalization showed an increase in RNA content whilst devernalization brought about a decrease in RNA synthesis by 40% (Zeevart, 1976). Under the same conditions, varieties of Arabidopsis not requiring vernalization did not show these results. Embryos of seeds of winter wheat have also been shown to contain increased RNA contents upon exposure to cold, but so too have embryos of spring wheat, although to a lesser extent (Vince-Prue, 1975). Finch and Carr (1956, cited in Zeevart, 1963), however, found no differential expression of nucleic acids between vernalized and unvernalized winter wheat. This is contradicted by Danyluk and Sarhan (1990) who indicate that low temperature induces differential gene expression in winter wheat, and suggest that protein kinase is involved in the regulation of transcription. Meza-Basso et al. (1986) showed preferential protein synthesis under cold temperature conditions in Brassica napus, of which 14 new proteins were shown to accumulate after 6 h. Numerous reports have therefore clearly established that plants exposed to low temperature synthesise a new set of proteins (Zeevart, 1976; Meza Basso et al., 1986; Danyluk and Sarhan, 1990). Terekota (cited in Vince-Prue, 1975) reported that in winter wheat the histone pattern is much more complex than in spring wheat. Furthermore, it becomes less complicated on exposure to low temperatures eventually resembling the simpler pattern found in spring wheat. A histone mediated suppression of genetic information in unvernalized winter wheat was therefore suggested.

Warm (1984) showed there to be qualitative and quantitative changes in the mRNA population in leaves of the LDP *Hyoscyamus niger* on exposure to LD. However it is not possible to distinguish between changes associated with floral induction and those caused by a change in photoperiod. It must be established whether the increase in

RNA synthesis is due to an increase in synthesis of RNA species already present, or the production of new RNA species. Lay-Yee *et al.* (1987) showed substantial quantitative and qualitative differences in RNA translation products from cotyledons of *Pharbitis nil* maintained under different photoperiodic conditions. The ability of red light to inhibit flowering in *Pharbitis* when given in the middle of the dark period was utilised in their experiments, and the effect of a NB yielded one specific 28 kD polypeptide difference compared to that of a SD induced cotyledon. The more specific effect of the NB on floral induction, and demonstrates that induction of flowering may involve changes in gene expression. The fact that only one protein change was detectable with the system used by Lay-Yee *et al.* indicates, also, that very few genes may be involved in floral induction in *Pharbitis nil*, and this may also be the case in other species.

With the above information in mind McDaniel (1991) posed the question " Does an inductive photoperiod lead to the production of new mRNAs in the apical meristem that are related to the transmission of the floral stimulus from the leaves to the apex ?" The inaccessibility of the shoot apical meristem and the small amount of material retrieved makes such a study very difficult. Studies on the protein composition of vegetative and floral apices of *Pharbitis nil* and *Xanthium* show no consistent differences in types of proteins (Kohli *et al.*, 1980), which is probably due to the difficulty of reproducing data with such small amounts of material. The introduction of the Polymerase Chain Reaction (PCR) to the field of molecular biology has meant that it may now be possible to work with minute amounts of tissue (e.g. a single cell) and look for changes in gene expression within that tissue. Therefore, it appears that the inaccessibility of the shoot apical meristem of many plant species, due to its small size, can be overcome. Kelly *et al.* (1990) and Meltzer *et al.* (1990) have used PCR to identify differences correlated with the onset of floral induction in the apical meristem.

Meeks-Wagner *et al.* (1989) note that the majority of mRNAs from tobacco floral structures are common to leaf mRNAs. To isolate genes involved in the transition to flowering differential screening (as described in section 1.10) of cDNA libraries is one option. This still requires large amounts of starting material, but using the property of tobacco that its cells can be grown in liquid culture as thin cell layers, and that these cells may be induced to floral or vegetative growth, means that large amounts of meristematic tissue is available for the study of gene expression during the transition to flowering in tobacco plants. Using differential screening Meeks-Wagner *et al.* (1989) isolated many families of cDNAs showing specificity to tissue induced to floral growth. Two transcripts corresponding to cDNAs from one family (PB 7-1 and PB 7-2) were shown to be present in pre-floral and floral apices, but not in juvenile apices. The levels of RNA corresponding to these two, and one other cDNA clone (PB 7-5) from the same family, was high in the roots of plants possessing immature influoresences, but in the roots of seedlings the level was very low, again indicating the importance of the roots in the transition to flowering.

#### 1.8.2 Genetic Control of Flowering

Genetic control of photoperiod insensitivity has been cited as being directed by a single gene in Maryland mammoth tobacco (Murfet, 1977) and the Ld mutant of Arabidopsis (Réide, 1975) or by several genes, as in Sorghum (Poa and Morgan, 1986) and Triticum aestivum (Law, 1987). In Sorghum the response to daylength is dominant over insensitivity, whereas in tobacco and wheat the opposite is found. The most extensive advances in this area of study have been with the work on mutants of Pea (*Pisum sativum*). Seven major loci have been identified that confer or modulate the ability to perceive or respond to daylength changes (Murfet, 1989). Of these genes, two (Sn and Dne) acting together confer sensitivity to daylength and the end product of their combined effect is a graft transmissible inhibitor (Murfet, 1977). Two other genes (Lf and Veg) confer sensitivity of the apex to the floral stimulus, and the most recent finding is that of the locus (G) which confers synthesis of the floral stimulus. The joint presence of dominant alleles of Sn and Dne confer a requirement for either LD or vernalization, which here act on the same process (Bernier, 1988). However it is generally accepted that the photoperiod (Ppd) and vernalization (Vrn) genes have proved to be separate and independent genes, probably controlling different component processes (Bernier 1988; Law, 1987). Purvis (1939) makes the assumption that the allele for winter habit of Petkus rye is dominant, but the progeny of F2 crosses between winter and spring rye do not follow the normal Mendelian pattern of inheritance (Pollock and Eagles, 1988). This is also true for Hyoscyamus niger, of which there are two races, annual and biennial. The biennial includes two forms, one with an obligate and one with a facultative vernalization requirement. This shows that the dominance of the biennial habit is incomplete. Genes controlling the response of plants to vernalization and photoperiod act independently in wheat suggesting that they may influence different component processes (Law, 1987). However there appears to be a degree of diversity amongst the known examples of plant genes produced in response to vernalization and photoperiod making a single unifying hypothesis for the effect of vernalization and photoperiod upon genes of flowering plants difficult.nth century all sugar came from sugar cane, but today half of the world

Besides the role of genes involved in the perception of light and low temperature there is a great deal of interest in general in those genes which govern pattern formation and cell differentiation in the meristematic tissue of flowering plants. Of particular interest are genes that play a role in the reorganisation of the vegetative meristem into the floral meristem and those required for establishing morphological patterns of petals, sepals, stamens and pistils. This is now developing into a vast topic of interest, and is beyond the scope of this report. However, it should be mentioned that the isolation of homeotic mutants (plants that transform one of the floral organ systems into another), has been successful in *Arabidopsis* (Schultz *et al.*, 1991) and *Antirrhinum majus* (Sommer *et al.*, 1990). Those cDNA clones isolated to date correspond to transcripts which are involved in determination of the identity of floral organ systems. One exception is the *leafy* mutant gene of *Arabidopsis* which is similar to the *floricula* gene isolated from *Antirrhinum*. They are expressed in the young flower primordia, but not in the inflorescence meristem, consistent with the proposal that they are involved in floral meristem identity.

#### 1.9 Sugar Beet

#### 1.9.1. Introduction

Sugar beet (*Beta vulgaris*, shown in Fig 1.3) is of particular interest commercially and physiologically because of its ability to store sugar at high concentrations, mainly as sucrose with a small concentration of reducing sugars (Terry, 1968). Before the seventeenth century all sugar came from sugar cane, but today half of the world production is refined from sugar beet. Cultivated beets are derived from the genus *Beta* which possesses useful characteristics such as disease resistance (Campbell, 1953).

## Fig 1.3 Mature Annual Beta vulgaris Plant

A 24 week old annual *Beta vulgaris* plant grown under short day conditions at a temperature of 20<sup>o</sup>C. The expanded tap root indicates that an excess of sugars is being stored in the root of this plant

#### Chapter 1: Introduction

par brei plants exist in two growth less DP but the biennist has, is addition, a to DD for Boral industics. Therefore, a



The daylength is known to be personally been been and the second second

#### 1.9.2 Effects of Photoperiod and Temperature

Sugar beet plants exist in two growth forms, annual and biennial. Both are obligate LDP but the biennial has, in addition, a vernalization requirement prior to exposure to LD for floral induction. Therefore, all sugar beet varieties may be maintained as vegetative rosette plants indefinitely if they are grown in SDs at warm temperatures (Stout, 1945). The cultivated sugar beet plant is biennial in growth habit (Lexander, 1985), and it is in the cold period of the winter months that reserves built up during the first growing season are stored in the expanded tap root (referred to as the beet). It is the expanded tap root that is harvested, and the sugar then extracted. However, different genotypes have different requirements for daylength and temperature (Lexander, 1985). The most susceptible biennial genotypes do not need very long days, or very long periods of low temperature exposure, to undergo floral initiation. They may, therefore, be induced to develop inflorescence stalks in their first growing season. This behaviour is very undesirable for the sugar producer as the tap roots of plants that have undergone this bolting process are very thin and highly lignified, making it difficult to retrieve a significant amount of sugar. It is, therefore, desirable for the sugar producer to suppress this bolting tendency. For the seed producer the biennial habit of the cultivated beet is, however, an obstacle to the rapid improvement of certain genotypes as they often have very specific requirements for daylength and temperature (Lexander, 1987).

The daylength is known to be perceived by the leaves of the sugar beet plant (Stout, 1945). The optimum light conditions required to induce flowering depend on the quality of the light (wavelength), the quantity of the light (fluence rate), and the periodicity (photoperiod). For sugar beet continuous illumination gives rise to flowering earliest, although some beet plants have been seen to flower in as little as 12 h of light

37

(Lexander, 1985). In general 16 h of light in a 24 h photoperiod induces flowering in the majority of sugar beet genotypes. The fluence range of light is required to be between 1000-2000 pmol m<sup>-2</sup> s<sup>-1</sup> for optimum flowering (Lasa and Péres-Peña, 1978). As for the wavelength requirement, it is only necessary that the appropriate amount or R/FR light is perceived at the right time to influence the action of phytochrome (Lane *et al.*, 1965). The extension of the daylength with fluorescent white light (Heide, 1973), or even low intensity red light (Lexander, 1975), was demonstrated to increase the bolting tendency of sugar beet plants. The intensity of light used to extend the daylength is, therefore, not required to be very strong to have an inductive effect, but the higher the light intensity, and the longer the daylength, the earlier the bolting of the sugar beet plant (Lexander, 1987).

The low temperature requirement of the biennial sugar beet plant has been demonstrated to be perceived by the shoot apex (Curth, 1960). After induction by LDs the floral stimulus produced is transported to the shoot apex. The apex of the biennial beet is only competent to respond to the floral stimulus once it has been vernalized. The belief is that the leaves produced by the unvernalized biennial plant are not competent to respond to the LDs and, therefore, do not produce the floral stimulus in LDs. It is only after vernalization that leaves competent to respond to LDs are produced at the apex, and in subsequent LD conditions the leaves will respond by producing the floral stimulus, which is perceived by the apex (Bernier *et al.*, 1981). Some biennial genotypes of sugar beet require as little as a few weeks of very mild temperatures for floral induction to occur, but, in general, plants of 4-8 weeks old at the start of the vernalization period require 10-14 weeks of vernalization at temperatures between 2-10°C for a total commitment to floral development (Lexander, 1985). Some sugar beet genotypes, after exposure to a period of low temperatures, will flower even in SDs (Vince -Prue, 1975). However, the time to flowering is dramatically decreased in LDs compared with SDs, and this phenomenon is only observed in a few bolting sensitive genotypes (Lexander, 1985). Fife and Price (1953) observed that certain genotypes of sugar beet bolted in continuous darkness after vernalization, provided they possessed a large amount of reserves in their tap root. This indicates that thermal induction may be more important than photoinduction in the initiation of reproductive growth in sugar beet. However, the importance of daylength in bringing seed formation to completion was demonstrated in their work, as seed set was only successful after vernalization and LDs. In the majority of genotypes the vernalization effect may be reversed if a sufficiently long period of SDs, or a period of high temperatures is perceived immediately after the vernalization period (Lexander, 1985). High temperature devernalization has been observed at lower temperatures in SDs than in LDs (perhaps due to the additive devernalizing effect of SDs). SD devernalization has been observed at temperatures above 18°C (Magara 1960) and at 24°C and above some devernalization also occurs in LDs, particularly if the plants were chilled in SD. Subsequent bolting is arrested until a further low temperature treatment is supplied (Heide, 1973).

The flowering of the cultivated genotypes of sugar beet plants, therefore, requires low temperatures prior to LDs, and can be promoted by artificial light used to extend the daylength, and counteracted by high temperatures and SDs prior to vernalization. There are also many different genotypes of cultivated sugar beet plants, all of which have specific individual requirements for daylength and temperature.

## 1.9.3 Juvenile Phase

Sugar beet possesses no juvenile phase (Chouard, 1960; Heide, 1973; Lexander,

1985). The responsiveness of the plant to environmental factors does increase with age but vernalization, or at least partial vernalization, is possible whilst the seeds are developing on the mother plant. For complete successful vernalization the tap root has to have developed to supply the nutrients required for vernalization (Chouard, 1960), therefore the observed vernalization of sugar beet embryos is incomplete and requires a further period of low temperature treatment to induce flowering. This partial vernalization means that plants originating from these seed are more easily induced to flower by treatment at low temperatures (Lexander, 1987). Therefore, seed produced in colder climates will yield a higher percentage of undesirable bolters than those produced in warmer climates. Also, if the seed is planted early in the spring and a late frost is experienced, a large percentage of the crop may be lost to bolting. Therefore, it is seen as a great disadvantage to the sugar beet industry that vernalization may be induced at any age.

#### 1.9.4 Genetics of Sugar Beet Annuality

Cultivated sugar beet varieties show a wide range of genetic variability in respect of the annual/biennial growth habit. In 1876 Rimpau stated that the annual habit is dominant (cited in Abegg, 1936). In controlled crosses between annual wild beet (*Beta patula*) and pollen from biennial commercial stock all progeny proved to be annual. Munerati (cited in Abegg, 1936) contributed the first significant advance in the genetics of bolting character in 1931. Crossing uniformly annual and biennial strains all the F1 progeny were shown to be annual, although the time required from germination to flowering was slightly longer than in the annual parent. The F2 progeny showed the typical 3:1 ratio of annuals to biennials indicating that annuality was controlled by a simple Mendelian factor. The non-requirement for a low temperature period to induce flowering in annual sugar beet plants has therefore been assigned to a gene B (for bolting) which is nearly completely dominant over b. In addition the existence of the gene B' is supposed which would confer the requirement for a short period of vernalization to induce flowering. Cultivated sugar beet plants do not contain the gene B for annuality, but by hybridisation with wild beets it can be introduced and cause unwanted bolting (Lexander, 1985).

The hypothesis for incomplete dominance is that the heterozygotes (Bb) are slower at inflorescence stalk development than the homozygotes (BB). Detailed examinations of various hybrids has shown that dominance is not complete (Abegg, 1936). The annual habit is, however, strongly dominant and the F2 ratios are always approximately 3:1 annual:biennial. Although the annual habit is determined by this single dominant gene (B) several genes must be involved in the bolting susceptibility of biennial sugar beets. The implication from the relationship between characteristics of non-induced plants and bolting-susceptible plants is that some genes are active before bolting inducing conditions occur (Lexander, 1987).

1.9.5 Events at the apex

The vegetative shoot apical meristem of sugar beet produces leaf primordia close to the terminal point of the apex. Consequently the axial zone is short and the apical dome is flat. After floral induction the apex elongates and the leaf primordia arise at greater distances from each other. When the apical meristem begins divisions leading to flower primordia and their bracts, these structures originate more superficially at the apex than do the vegetative primordia. Bracts are formed by the tunica cells, and the floral primordia are produced by the outer corpus layer. The axial portion of the apex proliferates and the rib meristem gives rise to longitudinal rows of cells, thus initiating the inflorescence axis. The flower primordia and bracts are formed successively closer to the tip, and finally a terminal flower primordium is laid down. Buds of the second order are produced at the sides of each flower primordium. The second tunica layer gives rise to the floral parts. Flowering proceeds from the base to the top in the main stem before the bracts. Within a cluster the bud in the middle is the first to open (Lexander, 1985).

#### 1.9.6 Mechanisms involved in the Regulation of Vernalization

Lexander (1975) suggested that membrane proteins were involved in the regulation of bolting susceptibility on the basis of observations of the sulphydryl content of the membrane proteins under low temperature conditions. Many exceptions were observed during this work and so to state that membrane permeability is affected in all sugar beet genotypes under vernalizing conditions is not possible. Sugars have also been observed to be limiting for flower initiation and an optimal concentration of sugars at the apex is possibly one requirement for flowering in this species as sucrose utilisation accompanies flower formation in sugar beet. Although many factors may be optimal for bolting, the requirement for one or more additional factors could be lacking as it is believed that several genes must be involved in bolting susceptibility in sugar beet (Lexander, 1975.) Some genes are assumed to be active before bolting induction occurs, but whether others exist that are activated only when such conditions are experienced is not known.

The theory of a hormone, referred to as vernalin, produced under low temperature conditions was abandoned since it was only possible to demonstrate transfer of a floral

stimulus, under non-photoinductive conditions, after vernalization in very few species (Vince-Prue, 1975). Flowering can be induced in non-vernalized scions of sugar beet by grafting them to vernalized stocks. This implies that a transmissible substance produced in the stocks under vernalizing conditions is transported to the scion, but the plants used as the donor in these grafts had been induced to flower by exposure to LDs. Therefore, it is probable that the unvernalized receptor plant receives the floral stimulus from the vernalized, and flowering, donor through the production of a photoinduced stimulus by the donor (Lexander, 1987).

GA3 has been shown to replace part of the cold requirement for extension of the inflorescence stalk of many plants, including sugar beet (Magara, 1964; Zeevart, 1976; Lexander, 1985). It is therefore assumed that GA is involved in the process of vernalization (Pharis and King, 1985). It is improbable that GA constitutes the flowering hormone, however, as in all but a few exceptional cases exogenous application of several different GAs only results in stem elongation and not flowering (Lexander, 1985). However, it is reported that the seeds of these exceptional individuals may have previously experienced a period of low temperature treatment (Gaskill, 1957; Stout, 1959). Application of GA does not, however, substitute for the LD requirement of sugar beet as such photoperiodic conditions are still required by plants after application of GA (Snyder and Wittwer, 1959). An increase in the endogenous concentration of GA has been observed in sugar beet plants after a period of vernalization (Lexander, 1985), but the quantity extracted from bolting and vegetative plants appeared to be the same according to Magara (1960). This was contradicted by Radley (1975) who suggested that there is an increase in the concentration of endogenous GA during the start of bolting in sugar beet. I have malized and flowering steeks were addred to hower. This was also

Some sugar beet genotypes could be highly sensitive to GA even before the cold

period, whereas others attain a high degree of sensitivity when experiencing low temperatures. The degree of sensitivity may be a function of the state of the cell membrane, or the abundance of the receptors (Lexander, 1975). Many chemicals have been shown to inhibit unwanted bolting in sugar beet (Lexander, 1985). The most interesting have been those that decrease the endogenous concentration of GA, such as 2chloroethyltrimethylammonium chloride (CCC) and Ethrel. However, in most cases flowering was not inhibited, only the process of stem elongation (Magara, 1967; cited in Lexander, 1985). This indicates that the major role of GA is in the control of stem elongation during floral development, and the production of floral organs is not dependent on the presence of GA. This was emphasised by the work of Reid (1986) on dwarf mutants of pea that are lacking in GA biosynthesis, in which the process of flowering is not inhibited.

The process of vernalization is believed to consist of two phases, and it is during the second phase that plants become sensitive to daylength and GA (Magara, 1964). This is in agreement with the suggestion of Vince-Prue (1975) and Bernier (1988) that low temperature makes the plant sensitive to other factors which, in turn, induce flowering. There is, therefore, evidence for many factors associated with the induction of flowering in sugar beet, all of which may be equally important, and none that have an effect on flowering in the absence of one or more other factors.

## 1.9.7 Grafting Experiments

Curtis and Hornsey (1964) showed that unvernalized biennial scions of sugar beet grafted onto unvernalized and flowering stocks were induced to flower. This was also demonstrated by Magara (1960). However, it was noted that defoliation of the scion may be necessary for floral induction. Naylor (1941) grafted annual and biennial sugar beet plants but failed to provide conditions necessary for translocation between them. Therefore the annual donor plant bolted whilst the biennial remained vegetative.

From grafting experiments concerned with the photoperiod alone it was shown that 3-6 weeks of continuous light are required to produce the floral stimulus in the leaves, translocate this to the apical bud, and induce the formation of the seed stalk (Stout, 1945). It was demonstrated in this work that vegetatively growing plants do not produce any substance antagonistic to reproductive development in sugar beet, but merely lack the substance in sufficient amounts that causes flowering.

## 1.10 h Aims cloning procedure involved the enrichment for poly (A ') RNA. Since the

#### ount of RNA available for this cloning procedure was very small it was d

Since the low temperature requirement of the biennial *Beta vulgaris* plant has been demonstrated to be perceived at the shoot apex (Curth, 1960), and since the annual *Beta vulgaris* plant is known to possess a gene which confers annuality upon the plant (Lexander, 1985), the principle aim of this study was to develop techniques for the isolation of cDNA clones corresponding to genes expressed in the shoot apex of the annual *Beta vulgaris* plant grown under non-floral-inductive conditions (short days, 20°C). Once such techniques were developed the aim was to characterise apex-specific cDNA clones to investigate whether any of these might be connected with vernalization by studying expression in the annual and biennial genotypes.

The amount of RNA available from small amounts of tissue such as the apical meristem of plants is limiting. Therefore, the intention was to develop techniques involving the Polymerase Chain Reaction (PCR). Since its introduction in 1985 (Saiki *et* 

al., 1985) PCR has transformed the way DNA analysis is carried out, and nucleic acid amplification has rapidly become one of the most important tools for the molecular biologist. The reaction is based on repeated thermal denaturation of double stranded DNA and extension by a thermostable DNA polymerase (from the bacterium Thermus aquaticus) of two primers, each annealed to a complementary template, to achieve amplification of greater than one million fold. PCR has recently lent itself to the field of cDNA cloning where small amounts of starting material are to be cloned. Using defined DNA adaptors with known restriction sites small amounts of cDNA may be amplified via PCR to build up enough material to allow cloning of that material into a suitable vector. This has been successfully demonstrated by Akowitz and Manuelidis (1989), Belyavsky et al. (1989), Timblin et al.(1990) and Domec et al. (1990). In all but the last reference cited here the cloning procedure involved the enrichment for poly(A<sup>+</sup>)RNA. Since the amount of RNA available for this cloning procedure was very small it was decided to eliminate this step, and develop a technique for producing cDNA libraries from small amounts of total RNA. A major disadvantages of using PCR is that amplification is prone to mismatching where a nucleotide substitution occurs. Therefore, the sequence of the cDNA clones produced by PCR may not represent the true sequence of the original mRNA. There is also the possibility that in a population of cDNA sequences PCR will preferentially amplify cDNAs of small size giving rise to a cDNA library which does not truly represent the population of cDNA clones that would have been produced using conventional cloning methods. Therefore, the representitiveness of any cDNA libraries produced was ascertained using several cDNA clones representing known plant mRNAs. The ultimate aim of this study was then to isolate cDNA clones representing mRNAs specific to the annual Beta vulgaris apical meristem tissue conferring annuality upon the genotype, and also involved in the process of vernalization. Therefore, the intention was

to prepare cDNA libraries from annual *Beta vulgaris* apical meristem tissue which had not been exposed to floral inductive conditions. These plants must, therefore, be maintained at warm temperatures, and exposed to short days only. On successful production of a PCR amplified cDNA library from such annual, warm, short day apical tissue it was hoped that clones specific to the apical meristem may be isolated using the techniques of differential screening. Here cDNA is prepared from mRNA isolated from two distinct cell types, plus and minus. cDNA clones specific to the plus cell type can be identified using a plus-minus hybridisation technique. Typically, duplicate nylon filters carrying cDNA clones derived from plus cells are screened with radiolabelled cDNA from plus and minus cells respectively. Clones shown to hybridise to the plus and not the minus cDNA probe are selected and characterised (Boll et al., 1986). This procedure requires vast amounts of cDNA. However, since the intention was to use PCR to prepare the cDNA for making the apical library, and since leaf tissue was used to develop the techniques involved in the PCR library production, an ample supply of cDNA should be available for this purpose. Differential screening does not, however, lend itself to the isolation of cDNA clones representing mRNAs present at a low abundance (<0.1%).

Subtractive hybridisation is a technique recently employed to try to overcome just such a problem. This process removes cDNA species common to both the plus and minus cDNA populations used in the differential screen by the use of an agent such as biotin, which is bound to the minus cDNA species present at a 1000 fold excess over the plus cDNA species. The two populations of cDNA are denatured and then incubated together in the presence of a high salt concentration at a temperature that will facilitate annealing of common cDNA species but not those specific to one cDNA population. The minus cDNA, whether it is single stranded, or double stranded, or a double stranded hybrid of plus and minus cDNA, is removed using avidin, a protein with a high affinity for the biotin moiety. The remaining cDNA should be specific to the plus population, and after several rounds of PCR amplification and subtraction low abundance cDNA species specific to the plus population of cDNA should be represented to a degree at which they are easily detectable. Such subtractive screening and library construction is possible without using PCR, but when starting material is limiting this procedure is aided by the amplification of plus and minus species. This procedure has recently aided the identification of many novel and important genes (Sive and St John, 1988. Timblin *et al.*, 1990).

All general laboratory reagents were of Analar parity and obtained from eather signator BDH unless otherwise indicated.

Sources of Envymes and Other Related Products.

Sources of General Laboratory Reagents

All restriction and modification enzymes were obtained from BRL and used according to their instructions, unless otherwise indicated. Deoxynucleoside triphosphases and Sephadex G-50 were obtained from Pharmacia UK. Bio-gel A 150 was obtained from Bio-Red. *Taq* Polymerase was obtained from Promega. All tadiochemistics were obtained from Amersham International.

Chapter 2: Materials and Methods

## CHAPTER 2

## MATERIALS AND METHODS

Loria broth solutions containing agar or agarose to give a solid support to the growth in Loria broth 1% (w/v) Bacto-Tryptone (Difco).

#### 2.1 General Laboratory Methods

General molecular techniques were carried out essentially as described in Sambrook et al. (1988).

#### 2.2 Sources of General Laboratory Reagents

All general laboratory reagents were of Analar purity and obtained from either Sigma or BDH unless otherwise indicated.

#### 2.3 Sources of Enzymes and Other Related Products

All restriction and modification enzymes were obtained from BRL and used according to their instructions, unless otherwise indicated. Deoxynucleoside triphosphates and Sephadex G-50 were obtained from Pharmacia UK. Bio-gel A 150 was obtained from Bio-Rad. *Taq* Polymerase was obtained from Promega. All radiochemicals were obtained from Amersham International.

#### Growth and Maintenance of Bacterial and Phage Stocks 2.4

The bacterial strains used in this study were grown in solutions of Luria broth or Luria broth solutions containing agar or agarose to give a solid support to the growth media.

Luria broth

1% (w/v) Bacto-Tryptone (Difco), 0.5% (w/v) Yeast extract (Difco) 1% (w/v) NaCl

pH 7.2

L-agar contained the above plus 1.5% (w/v) Bactoagar (Difco). Top agarose contained the above plus 0.6% (w/v) agarose (Gibco/BRL)

Bacterial strains were stored at -70°C as a frozen stock, prepared by mixing 1ml of an overnight culture with an equal volume of sterile glycerol. Stocks of bacteriophage were stored at 4°C in SM phage buffer containing 1% (v/v) chloroform.

#### SM Buffer;

100 mM NaCl 10 mM MgSO<sub>A</sub> 50 mM Tris-HCl (pH 7.5) 0.01% gelatin

#### 2.5 E.coli Strains Used in this Study

Strain of Tempera	Relevant Genotype	Source
XL-1 Blue	supE, hsdR, lac	I Jepson, ICI Seeds, Bracknell.
Warm, short day gr	$F'$ , proAB <sup>+</sup> , lac <sup>q</sup> , lacZ $\Delta M$	15
DP-50	supE, supF, hsdS	Stratagene
HB101	hsdRk <sup>-</sup> , hsdSk <sup>-</sup> , recA13,	Gibco/BRL
quantam seasor.	thi <sup>-</sup> , supE44.	

*E.coli* strains were routinely cultured at 37°C for between 18 and 20 h. Where bacteria containing plasmids were cultured the appropriate antibiotic was included at the following concentrations:

Ampicillin	150 μg/ml final concentration
Tetracycline	50 μg/ml final concentration
All antibiotics we	re stored as filter sterilised concentrate at -20°C.

2.6 Bacteriophage Strain Used in this Study

Strain	Relevant Genotype	Source
λZAPII	λsbhIλ1 <sup>o</sup> chiA131 srIλ3 <sup>o</sup> cIts857	Stratagene
Semales of	$srI\lambda 4 nin5 srI\lambda 5^{\circ} Sam100$	f RNA were disact

immediately. Samples of leaf tissue were stored at 70°C for later use. For the dissection of apical tissue the large leaves were removed from the region of

#### 2.7 Maintenance of Plants

## 2.7.1 Light and Temperature Regimes

#### Warm, short day growth conditions.

Temperature at plant height, 20-23°C; 8 h fluorescent light (Osram 45W warm white light ); light intensity 100-120  $\mu$ mol m<sup>-2</sup>s<sup>-1</sup>, (400-700 nm) as measured with a Li COR LI 185b quantum sensor.

## 2.7.2 Growth of Plants

Both diploid annual (7S 31660) and tetraploid biennial (5P 606) sugar beet (*Beta vulgaris*) seeds were supplied by Dr Wolfgang Schuch, ICI Seeds, Bracknell, UK. The seeds were soaked overnight in distilled water to leach out germination inhibitors, then sown in flat trays containing autoclaved compost and left to germinate in the warm SD growth room. When the first pair of true leaves emerged (after approximately 2 weeks) the seedlings were potted into 4 inch pots and grown to at least the 6 leaf stage.

## 2.7.3 Sampling of Plant Tissue

Samples of leaf and apical tissue for the isolation of RNA were dissected as described below and then frozen in liquid nitrogen (BOC). Samples of the apical tissue were used immediately. Samples of leaf tissue were stored at -70°C for later use.

For the dissection of apical tissue the large leaves were removed from the region of the shoot apex. The remaining leaves were removed with fine microdissection scissors under a dissecting microscope. When the apical meristem was visible it was dissected out and placed in a 1.5 ml microcentrifuge tube immersed in liquid nitrogen.

#### 2.7.4 Microscopy of Apical Tissue

Apices were fixed in 3% (w/v) glutaraldehyde (Aldrich Chemical Company Ltd. Dorset) for a minimum of 48 h. Mr Eoin Robertson, Department of Chemistry, University of Glasgow, embedded, sectioned and stained the tissue. Briefly, after fixing in the glutaraldehyde the tissue was rinsed in distilled water and dehydrated through immersion in solutions of increasing concentration of ethanol. The tissue was then cleared in xylene and after decanting most of the xylene, covered in wax pellets and placed in a 60°C incubator overnight. This was repeated twice more, and the tissue was then embedded in wax in preshaped plastic containers.

Sections of 10  $\mu$ m thickness were cut on a rotary microtome, separated on water and collected on standard glass slides. The sections were dewaxed in xylene, hydrated with water and mordanted in 1% (w/v) ferric ammonium sulphate for 40 min, rinsed in water and then transferred to 0.5% (w/v) haematoxylin and left overnight. Slides were regressively destained in 1% (w/v) ferric ammonium sulphate and the destaining process was stopped by a 1 h wash in tap water. Finally, the sections were dehydrated through an ethanol series and counterstained in 0.5% (w/v) Fast Green in xylene and mounted in DPX mounting medium.

After the various enzymatic treatments that DNA was subjected to, proteins and other contaminants which could inhibit further reactions were removed by phonol extraction using an equal volume of ultra pure phonol (BRL) containing 0.1% (w/v) 8-hydroxyquinoline (BDH Chemicals Limited). Phonol-extracted DNA was further extracted with phenol

#### 2.8 General Molecular Biology Methods

### 2.8.1 Estimation of Nucleic Acid Concentration

DNA and RNA concentrations were calculated by measuring the absorbance (A) of a solution at 260 nm and 280 nm in a Pye Unicam Model SPB-400 spectrophotometer, using a quartz cuvette with a 1 cm path length. A DNA solution with an  $A_{260}$  of 1.0 was taken to have a concentration of 50 µg/ml. An oligonucleotide with the same  $A_{260}$  was taken to have a concentration of 20 µg/ml. An RNA solution with the same  $A_{260}$  was taken to have a concentration of 40 µg/ml. (Sambrook *et al.*, 1988).

#### 2.8.2 Estimation of RNA Purity

The  $A_{260}/A_{280}$  ratio was used to assess the purity of a given RNA sample, with an acceptable ratio being within the range of 1.8-2.0. Samples having a ratio outside this range were subjected to a further round of phenol/chloroform extraction to remove proteins, carbohydrates, and unwanted salts. All RNA samples were also examined on a 1.5% (w/v) denaturing agarose gel to establish the level of degradation.

### 2.8.3 Phenol Extraction

After the various enzymatic treatments that DNA was subjected to, proteins and other contaminants which could inhibit further reactions were removed by phenol extraction using an equal volume of ultra pure phenol (BRL) containing 0.1% (w/v) 8-hydroxyquinoline (BDH Chemicals Limited). Phenol-extracted DNA was further extracted with phenol

saturated with chloroform, and finally twice with chloroform to remove residual phenol before ethanol precipitation. RNA was extracted using the protocol described in Jepson *et al.* (1991).

#### 2.8.4 Ethanol Precipitation

DNA in aqueous solution was precipitated in the presence of 0.15 M ammonium acetate with 2.5 volumes of ethanol which had been stored at  $-20^{\circ}$ C. This solution was kept at  $-20^{\circ}$ C for at least 1 h before recovering the DNA by centrifugation at 5,000 xg for 15 min in a cold room at 4°C. After removal of the supernatant the pelleted DNA was vacuum dried (or left to air dry for 1 h) before being resuspended in either 1xTE (10 mM Tris-HCl pH 7.6, 1 mM EDTA pH 8.0) or sddH<sub>2</sub>O.

2.8.5 Agarose Gel Electrophoresis

DNA was separated in agarose gels (1.0 - 2.5% (w/v)) using a Tris borate buffer as described (Sambrook *et al.*, 1988: pp 6.9-6.13). Before loading on to the gel, the DNA was mixed with 0.2 volumes of 5x loading buffer ( 80% (v/v) glycerol, 0.01% (w/v) bromophenol blue in 1x TE ) (Sambrook *et al.*, 1988).

RNA was separated in agarose gels containing 1.5% (w/v) agarose in a formaldehyde gel buffer consisting of the following;

2.2 M formaldehyde 20 mM MOPS (pH 7.5) 8 mM Na Acetate (pH 5.5) 1 mM EDTA (pH 8.0) Final pH: 7.0

All the solutions were added and heated to  $65^{\circ}$ C before adding the formaldehyde. The gel was then poured and allowed to set for 30-60 min in a fume hood. RNA in solution was made up to a concentration of 10-30 µg in 4.5 µl of DEPC treated sddH<sub>2</sub>O. To the RNA samples the above solutions were added to the same final concentrations with the addition of 12.5 M formamide to maintain the RNA in a denatured form at low temperatures (37-42°C). The samples were then denatured at 65°C for 15 min. Immediately 2 µl of formaldehyde gel loading buffer was added to each sample before loading onto the gel.

Formaldehyde Gel Loading Buffer

50% (v/v) glycerol 1 mM EDTA (pH 8.0) 0.25% (w/v) bromophenol blue 0.25% (w/v) xylene cyanol FF

Once loaded the gel was run at 25 V overnight.

2.8.6 Isolation of DNA from Agarose Gels

DNA was isolated from agarose gels by microcentrifugation of the gel slice through siliconised glass wool (Heery et al., 1990). A hole was pierced in a 1.5 ml microcentrifuge tube with a needle before adding the glass wool and the gel slice. This tube was inserted into a second 1.5 ml microcentrifuge tube to collect the solution containing the DNA. Both tubes were centrifuged at 5,000 xg for 20 min and the liquid collected was sequentially extracted with phenol and chloroform before being ethanol precipitated as described in section 2.8.4.

#### 2.8.7 Denaturing Polyacrylamide Gel Electrophoresis

Denaturing polyacrylamide gel electrophoresis was set up according to the described method utilising urea as a denaturing agent (Sambrook *et al.*, 1988: pp 6.49). These gels were used for purifying oligonucleotides and monitoring ligation reactions. The usual running conditions were as follows:- 17 watts (approximately 350 V, 40 mA) for 2 h on an Atto preparative gel apparatus.

#### 2.9 Preparation and Purification of Oligonucleotides

Oligonucleotides for use as PCR primers were synthesised on an Applied Biosystems 380b Synthesiser. Oligonucleotides were routinely purified by denaturing polyacrylamide gel electrophoresis on a 20% (w/v) gel as previously described. Oligonucleotides were first deblocked by incubation in 40% (v/v) ammonium hydroxide at  $55^{\circ}$ C for 7 h before being concentrated by freeze drying. Approximately 100 µg of crude deblocked oligonucleotide in 50 µl of 1X TE was mixed with 10 µl of 5X gel loading buffer and run on a 20% (w/v) gel. Oligonucleotides were visualised by UV shadowing using a short wave UV light (260nm) and a TLC plate with fluorescent indicator (Kodak Eastman 13252 TLC plate with fluorescent indicator N<sup>Q</sup> 6063). The product was then excised from the gel with a sterile
scalpel and placed into a 1.5 ml microcentifuge tube. The oligonucleotide was eluted from the gel by crushing the gel with a sterile disposable inoculation loop and incubating at  $65^{\circ}$ C with 500 µl of 1X TE for between 5 and 16 h. The contents of the tube were transferred to a fresh microcentrifuge tube which had been prepared by piercing a hole in it with a sterile needle. This tube was packed with siliconised glass wool and placed in to a second microcentrifuge tube. This was centrifuged for ten sec at 5,000 xg, which was sufficient to force the liquid through the glass wool but retain the acrylamide. The lower microcentrifuge tube was recentrifuged to remove the final traces of polyacrylamide particles. The oligonucleotide solution was made up to 10 mM MgCl, 0.1 M NaCl and ethanol precipitated. Oligonucleotides were resuspended in 1X TE at a final concentration of 100 µg/ml.

### 2.10 Oligonucleotides Used in This Study

Two major Oligonucleotides were used in this study for the purpose of PCR amplification of double stranded cDNA. These were as follows: Primer 1 (JB1): (5'-ATGCTTAGGAATTCCGATTTAGCCTCATA-3') Primer 2 (JB2): (3'-AAATCGGAGTAT-5')

Two further oligonucleotides were used in this study to PCR amplify cDNA inserts from purified bacteriophage plaques. These were as follows: Primer 3 (PUC1): (5'-ATGCGGATCCGTTTTCACACAGCAAACAGCTATGAC-3') Primer 4 (PUC2): (5'-TGGGGGATTCGTGCGACGTTGTAAAACGACGGCCAGT-3')

Primer 1 (JB1) was extensively used in the PCR amplification of double stranded cDNA ligated to annealed primers 1 and 2, as well as for the PCR amplification of cDNA inserts from selected plaques to show the number of inserts present in a single recombinant.

## 2.11 End Labelling of Double Stranded DNA

Double stranded cDNA was end labelled with <sup>32</sup>P dCTP to ensure that all cDNA was blunt-ended prior to ligation to annealed primer 1 and primer 2 oligonucleotides. The end labelling reaction (50 µl) contained:

10  $\mu$ g DNA 125 nM each of dATP, dTTP, dGTP 50 mM NaH<sub>2</sub>PO<sub>4</sub> (pH 7.5) 3 mM MgCl<sub>2</sub> 1 mM 2-mercaptoethanol 5  $\mu$ Ci [ $\alpha$ -<sup>32</sup>P] dCTP (3000 Ci/mmol) 50 units Klenow fragment H<sub>2</sub>O to 50  $\mu$ l

The reaction mixture was incubated at room temperature (22°C) for 20 min. Finally cold dCTP was added to a final concentration of 125 nM to ensure that dCTP was not limiting, enabling the reaction to go to completion. This was left at room temperature for 10 min before extracting sequentially with phenol and chloroform (as described in section 2.8.3) and being passed down a Sephadex G-50 column.

## 2.12 Radiolabelling of DNA Fragments

In almost every case cDNA fragments were radiolabelled using an Amersham random priming kit according to their instructions. After labelling the unincorporated nucleoside triphosphate was removed by chromatography through a Sephadex G-50 column. The column was a Pasteur pipette plugged with siliconised glass wool. Two column volumes of 1X TE was passed down the column prior to loading the radiolabelled DNA sample. 200  $\mu$ l fractions were collected and the radioactivity within them estimated using a scintillation counter. The void peak was pooled (fractions 3-6 in most instances) and stored on ice before use. When labelled DNA was to be used in a hybridisation experiment it was first heated in a boiling water bath for 2-3 min to denature the DNA duplex, before adding it to the hybridisation solution as quickly as possible.

## 2.13 Non-Radioactive Labelling of DNA

For the purpose of producing subtracted probes, cDNA from sugar beet leaf material was labelled with biotin by nick translation using the BRL kit (cat No 8160SB) containing biotin-7-dATP. Unincorporated biotin was removed by chromatography on a Sephadex G-50 column as previously described (section 2.12). 5  $\mu$ l aliquots from each of the twenty 200  $\mu$ l fractions were analysed for biotin incorporation by dot blotting and visualising with the BRL blue GENE non-radioactive nucleic acid detection system.

## 2.14 Hybridisation of Radiolabelled Probes

2.14.1 Plaque Hybridisation (Benton and Davies, 1977)

Plaque hybridisation was carried out by placing 10 cm<sup>2</sup> nylon membrane filters onto

16-20 h. The filters were removed from the invubator and weshed in 3X SSC pH 7.5, 0.195

agar plates containing the phage plaques. These plates were placed at 4°C for at least 1 h prior to applying the filters. The filters were allowed to contact the plate for 30 sec (60 sec, 120 sec, and 180 sec for lifts 2, 3, and 4 respectively). After orientating the filters with India ink, they were placed plaque side up on 3MM paper soaked in the following solutions;

i) 0.5 M NaOH, 1.5 M NaCl for 5 min ii) 0.5 M Tris-HCl pH 7.5, 1.5 M NaCl for 5 min iii) 2X SSC pH 7.5 for 5 min

The filters were allowed to air dry for 10 min before being baked at 80<sup>o</sup>C in a vacuum oven for 1-2 h. The filters were then pre-hybridised for at least 4 h in the following buffer;

6X SSPE pH 7.5 0.25% (w/v) milk powder (Marvel) 0.1% (w/v) SDS

In general 10 ml of pre-hybridisation buffer were used for every filter being prehybridised. The pre-hybridisation took place in 10 cm<sup>2</sup> sandwich boxes capable of holding 200 ml of solution. For the hybridisation procedure a maximum of 10 ml of pre-hybridisation solution was added to a 10 cm<sup>2</sup> petri dish to which was added the radiolabelled probe DNA. Each filter to be treated was added individually to eliminate air bubbles, and ensure all filters were exposed to the probe. The petri dish lid was held on with two lengths of Nesco film bound tightly around the edge of the dish. The petri dish containing the filters and the probe was placed inside a 20 cm<sup>2</sup> petri dish, acting as a spill tray, and incubated at 65<sup>o</sup>C for 16-20 h. The filters were removed from the incubator and washed in 3X SSC pH 7.5, 0.1% (w/v) SDS, at 65<sup>o</sup>C for 90 min, with three changes of wash solution during that time, unless otherwise indicated. Autoradiography was carried out at -70°C for 24-72 h as indicated, with or without an intensifying screen.

## 2.14.2 Southern Analysis of DNA Fragments (Southern, 1975)

Complementary strand DNA and DNA produced by PCR amplification were subjected to electrophoresis on agarose gels at 25-30 V overnight. The gel was then placed into 1 1 of 0.5 M NaOH, 1.5 M NaCl for 45 min , followed by 5 min in 1 1 of 0.5 M Tris-HCl pH 7.5, 1.5 M NaCl. The gel was then placed onto the blotting apparatus as described (Sambrook *et al.*, 1988: pp 7.46-7.48) and blotted overnight on to a nylon membrane. After blotting the membrane was placed into 2X SSC for 5 min, after which time it was allowed to air dry for 10 min before being baked at 80°C for 1-2 h. The membrane was then used immediately, or stored at 4°C in prehybridisation solution for up to 12 months. Hybridisation and autoradiography were carried out essentially as previously described.

# 2.14.3 Dot Blot Analysis of DNA Fragments

Aliquots of DNA in solution were heated to 95°C for 2-3 min and immediately chilled on ice. An equal volume of ice-cold 20X SSC was added to the samples. A section of Hybond-N (Amersham) filter was cut to 120 x 80 mm and immersed in distilled water for 5 min, followed by 10X SSC for 20 min. The filter was then set up in a BRL HYBRI-DOT 96hole manifold apparatus (No. 580-1050 MM) and suction sufficient to pull 1 ml of water up a 2 ml glass pipette applied. Each well was washed through with 200 µl of 10X SSC and the required number of samples were loaded into the appropriate wells. Following loading of DNA solutions, the wells were washed through with two further applications of 200 µl of 10X SSC. The manifold apparatus was then dismantled and the filter subjected to denaturing conditions and baking as previously described (section 2.14.1).

## 2.14.4 Northern Analysis of RNA Samples

Denatured RNA samples were subjected to electrophoresis through 1.5% (w/v) agarose gels containing 2.2 M formaldehyde at 25 V overnight. Before blotting the lane containing the RNA markers was excised with a sterile scalpel. The remainder of the gel was blotted as previously described. Once blotted the filter was baked at 80°C for 1-2 h before pre-hybridising for 4-6 h at 65°C in 6X SSC, 2X Denhardts solution (0.04% (w/v) Polyvinylpyrolidone, 0.04% (w/v) Ficoll (type 400, Pharmacia), 0.04% (w/v) Bovine Serum Albumin (fraction V, Sigma)), 0.1% (w/v) SDS.

The filter could be probed immediately, or stored at 4<sup>o</sup>C for up to 12 months. Denatured radiolabelled probe was added to a new hybridisation bag containing the filter to be probed. The denatured DNA probe was added and the bag was sealed, removing all air bubbles in the process. The filter was then incubated at the recommended temperature for 16-20 h in a shaking water bath before washing in a range of salt washes at 65<sup>o</sup>C depending on the size and relative homology of the probe being used.

# 2.15 Polymerase Chain Reaction

For the duration of this study two separate PCR amplification protocols were used. For the PCR amplification of double stranded cDNA a protocol devised by Keith Edwards

### Chapter 2: Materials and Methods

and colleagues (at ICI Research Station, Jealotts Hill) was used. PCR reactions were set up to contain the following in a final volume of 50  $\mu$ l

10 mM Tris-HCl (pH 8.3) 50 mM KCl 1.5 mM MgCl<sub>2</sub> 0.1% (w/v) gelatin 200 μM of each of the four dNTP's 600 ng of the primer JB1 Template DNA 2.5 units of *Taq* DNA polymerase

The protocol for PCR amplification using primer 1 (JB 1) was as follows;

i) The reaction mixture was heated to  $72^{\circ}$ C for 2 min to allow the smaller primer (JB2) to melt off, and *Taq* polymerase enzyme filled in the overlap remaining.

ii) The DNA was then denatured at 94°C for 1 min.

iii) The primer (JB1) was allowed to anneal to its complementary DNA by cooling to 68°C for 0.8 min.

iv) Taq polymerase was allowed to extend the product by heating to 72°C for 3 min. Steps ii), iii) and iv) were repeated 35 times.

v) Taq polymerase was allowed to extend all unfinished DNA products by maintaining the temperature at  $72^{\circ}$ C for a further 10 min.

The cycling process was carried out on a Techne PHC-1 machine pre-set to complete the desired cycle.

In the second protocol primers PUC 1 and PUC 2 were used instead of the single JB 1 primer, and steps iii) and iv) were altered as follows;

iii) The primers (PUC 1 and PUC 2) were allowed to anneal to their complementary DNA strands by cooling to 65<sup>o</sup>C for 1 min.

iv) Taq Polymerase was allowed to extend the product by heating to 72°C for 1.2 min.

## 2.16 Preparation of Total Plant RNA

## 2.16.1 Large Scale RNA Prep

All plasticware used throughout this procedure was autoclaved at 20 psi for 15 min and glassware was treated with 0.1% (v/v) diethylpyrocarbonate (DEPC) before baking for 16-24 h at 200°C. Solutions other than those containing Tris were rendered RNAse free by treating them overnight with 0.1% (v/v) DEPC and autoclaving as for the plasticware. Stock solutions of Tris-HCl were filter sterilised before use.

Between 5 and 10 g of plant tissue was frozen in liquid nitrogen and then ground to a fine powder using a mortar and pestle which had been sterilised by autoclaving. Once a fine powder was obtained 25 ml of phenol-cresol solution was added along with 50 ml of homogenising buffer.

The plant material and solutions within the mostar were allowed to make white continuing to grind until a paste formed. The passe was doen transferred to 30 ml Corex tube and contrifuged at 22,000 xg, 4°C for 15 minutes. The supersulant was transferred to 50 ml Falcon tribes (Elkay Industries Ltd) containing an equal volume of phenol-chleroform and

#### **Chapter 2: Materials and Methods**

Homogenising Buffer400 mM NaCl50 mM Tris-HCl (pH 9.0)5 mM EDTA1% (w/v) SDS10 mM Dithiothreitol (DTT)1 mM Aurintricarboxylic acid (ATT)10 units Heparin

## Phenol-Cresol Solution

100 g phenol
0.1 g 8-hydroxyquinoline
10 ml Cresol
Homogenising buffer added to saturation.

Phenol-Chloroform Solution 100 g phenol

0.1 g 8-hydroxyquinoline

50 ml chloroform

sdd H<sub>2</sub>O added to saturation

The plant material and solutions within the mortar were allowed to thaw whilst continuing to grind until a paste formed. The paste was then transferred to 30 ml Corex tubes and centrifuged at 22,000 xg, 4°C for 15 minutes. The supernatant was transferred to 50 ml Falcon tubes (Elkay Industries Ltd) containing an equal volume of phenol-chloroform and

centrifuged at 5,000 xg, 4°C for 30 min. This last step was repeated before adding 12 M LiCl to the supernatant to a final concentration of 2 M and incubating the tubes at 4°C overnight. The tubes were then centrifuged at 5,000 xg, 4°C for 30 min as before. The supernatant was discarded and the pellet resuspended in a minimal volume of 5 mM Tris-HCl (pH 7.5) and precipitated with LiCl as previously described. The tubes were centrifuged once more at 5,000 xg and the pellet remaining was washed in ice cold 70% ethanol and left on ice for 10 min. The tubes were centrifuged once more and the pellet dried under vacuum before being resuspended in a small volume (100-500  $\mu$ l) of DEPC treated sdd H<sub>2</sub>O. The concentration of RNA was then determined as described in section 2.8.1.

# 2.16.2 Small Scale RNA Prep

To isolate RNA from apical meristem tissue from sugar beet it was necessary to scale down the above procedure to accommodate such small amounts of tissue. The volumes used in the large scale procedure were reduced 100 fold to allow the extraction of apical meristem RNA to be performed in 1.5 ml microcentrifuge tubes. It was assumed that the weight of the 35 sugar beet apical meristems would be approximately 50  $\mu$ g which is at least 100 fold less than the amount of starting material used in the large scale RNA prep. Measuring the weight of such a small amount of tissue without thawing that material proved to be extremely difficult and so the assumption was made by measuring the weight of a similar volume of small sized pieces of sugar beet young leaves. The small amount of tissue was ground to a fine powder under liquid nitrogen in a microcentrifuge tube using a small glass rod. 250  $\mu$ l of phenol-cresol solution was then added along with 500  $\mu$ l of homogenising buffer. The plant material and solutions within the microcentrifuge tube were allowed to thaw whilst continuing to grind. The resultant paste was centrifuged for 15 min at 5,000 xg and the supernatant transferred to a second microcentrifuge tube. To the solution remaining in the first microcentrifuge tube a further 250  $\mu$ l of homogenising buffer was added to ensure that as much RNA as possible was extracted from the small amount of starting material. This first microcentrifuge tube was shaken by hand for 10 min before being centrifuged as before. The supernatant from this second centrifugation was added to the original supernatant in the second microcentrifuge tube and an equal volume of phenol-chloroform was added before centrifugation as before. This last step was repeated before adding 12 M LiCl to the supernatant at a final concentration of 2 M and incubating the microcentrifuge tube at 4°C overnight. Essentially from here on the small scale RNA prep followed the method of the large scale RNA prep except that microcentrifuge tubes were used and the final pellet of RNA was resuspended in approximately 10  $\mu$ l of DEPC-treated sdd H<sub>2</sub>O.

# 2.17Synthesis of Double Stranded cDNA(Gubler and Hoffman, 1983;<br/>Lu and Werner, 1988)

cDNA was synthesised in all cases using an Amersham cDNA synthesis kit. The method used is a development of the method of Gubler and Hoffman (1983). In brief, the first strand cDNA is primed with oligo dT and synthesised using AMV (avian myeloblastosis virus) reverse transcriptase. Second strand cDNA synthesis is performed using the mRNA/cDNA hybrid as a substrate as follows;

- 1) *E.coli* ribonuclease H (an endoribonuclease which will digest an RNA/DNA hybrid only) is utilised to produce nicks and gaps in the mRNA strand.
- 2) This provides efficient primers with 3'OH for *E.coli* DNA polymerase I to replace the mRNA strand by a nick translation type reaction
- 3) T4 DNA polymerase is used to remove any small 3'-overhangs on the first strand

## cDNA

A variety of different RNA concentrations were used to establish the best conditions for cDNA synthesis using the Amersham cDNA synthesis kit. Since the eventual intention was to prepare cDNA from small quantities of apical meristem RNA it was decided to use total RNA and not to enrich for poly (A<sup>+</sup>) RNA. The sample of RNA to be used was dissolved in a total volume of 10 ml of DEPC treated sdd H<sub>2</sub>O. The first strand cDNA synthesis was set up using the following;-

4μl first strand buffer (contents unknown) 1 µl Na pyrophosphate (concentration unknown) 1 µl human placental RNAse inhibitor (20 units/µl) 2 µl dNTPs (10 mM dATP, dGTP and dTTP, 5 mM dCTP) 1  $\mu$ l oligo dT (1.6  $\mu$ g/ $\mu$ l) 5 μCi [α-<sup>32</sup>P] dCTP (3000 Ci/mmol) X $\mu$ g of RNA \* denatured at 65<sup>o</sup>C for 5 min included at 37°C for 10 15 sdd H<sub>2</sub>O to 20  $\mu$ l as storned by adding LDTA (pH 8.0) to a

\* The amount of RNA used was dependent on the purity of the RNA. 100 X more RNA was used than would have been if  $poly(A^+)$  RNA and not total RNA was used.

The reactants were added in the order shown above and mixed gently. 20 units of AMV reverse transcriptase were added and the solution mixed and centrifuged for 3 sec. The reactants were then incubated at 42°C for 40 min.

A 2 µl aliquot was removed from the first strand reaction mixture to be subjected to denaturing agarose gel electrophoresis. The remainder of the first strand mixture was

#### Chapter 2: Materials and Methods

combined with the following:

37.5  $\mu$ l second strand buffer (contents unknown) 20  $\mu$ Ci [ $\alpha$ -<sup>32</sup>P] dCTP (3000 Ci/mmol) 1  $\mu$ l *E.coli* ribonuclease H (0.8 units/ $\mu$ l) 6.5  $\mu$ l *E.coli* DNA polymerase (3.5 units/ $\mu$ l) H<sub>2</sub>O to 100 $\mu$ l

The reactants were added in that order, mixed quickly and centrifuged for 3 sec before incubating at:-

12<sup>o</sup>C for 60 min 22<sup>o</sup>C for 60 min 70<sup>o</sup>C for 10 min

At this point 4.0 units of T4 DNA polymerase I were added and the reactants incubated at  $37^{\circ}$ C for 10-15 min. The reaction was stopped by adding EDTA (pH. 8.0) to a final concentration of 1 mM. A 10 µl sample of the second strand reaction mixture was removed to be subjected to denaturing agarose gel electrophoresis along with the first strand mixture and a sample of radiolabelled *Hind* III digested lambda DNA. The gel was washed in 7% (w/v) trichloroacetic acid for 30 min and then blotted according to the procedure described in section 2.14.2.

The double stranded cDNA was size fractioned on a Pasteur pipette column containing Sephacryl 300 (Pharmacia) added to 0.5 cm below the top of the pipette. The column was washed with 2 column volumes of 1X TE before the radiolabelled cDNA was added. A further 100  $\mu$ l of 1X TE was added before collecting 200  $\mu$ l fractions from the column. A 10  $\mu$ l aliquot was taken from every fraction containing any radioactivity. These samples were subjected to non denaturing agarose gel electrophoresis and blotted as previously described (section 2.14.2). Those fractions containing unincorporated radioactivity were discarded and the remaining fractions were pooled together and ethanol precipitated. The resulting pellet was resuspended in a small volume (10-20 µl) of 1X TE. To ensure that the cDNA was blunt ended prior to ligation of the linkers the Klenow fragment of DNA polymerase 1 was used as described in section 2.11.

## 2.18 Preparation of Sugar Beet cDNA Libraries

## 2.18.1 PCR Amplified cDNA

To establish the feasibility of PCR amplification of blunt ended cDNA using primers JB 1 and JB 2  $\lambda$  DNA digested with *Hind* III was used as a control. Primers JB 1 and JB 2 were annealed by incubating 10 µg of JB 1 with 5 µg of JB 2 in 5 µl of 1X TE containing 100 mM NaCl at 70°C and allowing them to cool, over a period of 4-5 h, to room temperature. The annealed oligonucleotide primers were subjected to ligating conditions with a sample of  $\lambda$  DNA digested with *Hind* III ( $\lambda$ /*Hind* III DNA) which had been blunt-ended using the Klenow fragment of DNA polymerase I. Several control ligations using  $\lambda$ /*Hind* III DNA were set up as follows;

# Non Self Ligation Control

i)  $\lambda$ /Hind III DNA incubated overnight at 12<sup>o</sup>C in ligation buffer but in the absence of any ligase enzyme

ii)  $\lambda$ /Hind III DNA incubated overnight at 12°C in ligation buffer in the presence of

T4 DNA ligase.

iii)  $\lambda$ /Hind III DNA which had been blunt-ended using the Klenow fragment of DNA polymerase I was incubated overnight at 12<sup>o</sup>C in ligation buffer in the presence of T4 DNA ligase

iv)  $\lambda$ /Hind III DNA which had been blunt-ended using the Klenow fragment of DNA polymerase I was incubated overnight at 12°C in ligation buffer in the presence of T4 DNA ligase and the annealed JB 1 and JB 2 primers.

# PCR Amplification Control

 $\lambda$  DNA digested with *Hind* III and ligated to the annealed primers was subjected to agarose gel electrophoresis and individual bands were removed from the gel with a sterile scalpel. The DNA from each gel slice was eluted by centrifugation at 5,000 xg for 20 min as described in section 2.8.6. PCR amplification of the DNA using the cycling procedure designed for use with the JB 1 primer, described in section 2.15, was performed on an undiluted sample eluted from the gel and a 1:10 dilution of this sample.

Leaf cDNA was next used to confirm that the amplification procedure would be successful with a sample of multiple mixed cDNA species. Blunt-ended leaf cDNA was ligated to the annealed primers and passed down a BIO-GEL A-150 column, prepared in a similar manner to the sephadex G-50 column described in section 2.12. A 1  $\mu$ l aliquot was removed from all fractions off the column containing any radioactivity and subjected to PCR amplification directly. A further 10  $\mu$ l aliquot was removed from each fraction and subjected to agarose gel electrophoresis and blotted as described in section 2.14.2. The remainder of the fractions of interest were pooled together and ethanol precipitated. This sample of cDNA was subjected to electrophoresis through a 1.5% (w/v) agarose gel for a short time (20 min at 40 mA) and several slices of agarose were removed at 4 mm intervals down the gel, and the DNA eluted as previously described (section 2.8.6). Each sample was diluted and PCR amplified. The PCR amplified products were size fractioned on a 1.5% (w/v) agarose gel by removing all DNA above 200-300 base pairs. The larger DNA was eluted from the gel as previously described (section 2.8.6) and digested with *Eco* RI. Once the digestion was completed the PCR amplified cDNA was ethanol precipitated and resuspended in 10 to 20  $\mu$ l of 1X TE. The concentration of the cDNA was estimated as described in section 2.8.1. To establish that the cDNA had been digested with the *Eco* RI enzyme it was incubated under ligating conditions and an increase in size of the cDNA, when subjected to electrophoresis through a 1.5% (w/v) agarose gel, demonstrated that the *Eco* RI digested leaf cDNA was estimated as previously described (section 2.8.1) and 50 to 250 ng of the cDNA was ligated to 1  $\mu$ g of  $\lambda$  ZAP II DNA before packaging.

## 2.18.2 Preparation of Plating Cells

Cells used to plate lambda phage were prepared freshly as required by inoculating 20 ml of L-broth containing 0.2% (w/v) maltose and 100 mM MgCl, with 200  $\mu$ l of an overnight culture of the desired strain. The cells were grown to an A<sub>600</sub> of 0.2 and centrifuged at 3000 xg. for 20 min. The cells were resuspended in 8 to 10 ml of ice cold 10 mM MgSO<sub>4</sub> and stored on ice until required.

2.18.3 Packaging of Cloned cDNA into Bacteriophage Lambda

Between 50 and 250 ng of cDNA was ligated to 1  $\mu$ g of  $\lambda$  ZAP II DNA as previously described (section 2.18.1). The ligated DNA was then packaged into  $\lambda$  bacteriophage from

the invitrogen kit supplied by Stratagene, following their instructions. The packaged DNA was diluted with 500  $\mu$ l of SM phage buffer (described in section 2.4) and 10  $\mu$ l of chloroform was added. Serial dilutions were made from 10  $\mu$ l of this stock and incubated with 200  $\mu$ l of plating cells. The inoculated cells were plated onto L-agar with top agarose containing 40  $\mu$ g/ml IPTG and 40  $\mu$ g/ml X-Gal and incubated at 37°C for 16 to 24 hours.

#### 2.18.4 Estimation of Phage Concentration

The number of phage within a sample was determined by preparing serial ten-fold dilutions of the phage in SM buffer (described in section 2.4). 20  $\mu$ l aliquots of these dilutions were spotted onto a lawn of the appropriate host strain of bacterium, prepared by mixing the 200  $\mu$ l of plating cells with 3 ml of top agarose. The spots were allowed to dry for 20 min at room temperature, before being incubated at 37°C for 10-12 hours. The titre of the phage was then determined.

Gpal 2 (Phenylalanine ammonia-lyase): 1.8 kbp cDNA tragment from

# 2.19 Determination of the Status of the PCR Amplified Libraries

## 2.19.1 Examination of Random Inserts

Plugs of phage plaques were taken at random from plates showing the greatest number of recombinant clones, using the narrow end of a Pasteur pipette. The plugs of agar containing the phage were suspended in 200  $\mu$ l of SM phage buffer (described in section 2.4) and incubated at room temperature for 60 min. They were then stored at 4°C for 12 to 18 months, or until needed. The inserts from these plaques were obtained by 1 of 2 ways; i) The insert was PCR amplified directly from the SM buffer solution containing the phage plaque using primer 1 (JB 1) in a PCR reaction containing 28  $\mu$ l of the phage in solution.

ii) The above was repeated using primer 3 (PUC 1) and primer 4 (PUC 2).

## 2.19.2 Screening with Specific cDNAs

Several cDNA clones were used to establish whether the PCR amplified cDNA libraries were representative of standard, non PCR amplified cDNA libraries. The cDNA clones used for screening were chosen because of their various relative abundances, and are listed below;

cDNA Clone	Source
LHCP (Light harvesting complex chlorophyll <i>a/b</i> binding protein):	Keith
1.1 kbp cDNA fragment from Zea mays cloned into pWreck vector	Edwards
Gpal 2 (Phenylalanine ammonia-lyase): 1.8 kbp cDNA fragment from	Keith
Zea mays cloned into PUC 8	Edwards
α-Tubulin: 500 bp cDNA fragment from Zea mays cloned into pBR 322	Keith Edwards
ALS (Acetolactate synthase): 600 bp fragment from Pisum sativum	Ian
cloned into $\lambda ZAP$	Jepson
PTA 71: 700 bp genomic DNA fragment of wheat rDNA	Simon
TTATIL TOO OP BOILOIMO DI LI DI	Brown

Phage plaques from the PCR amplified cDNA libraries were plated as previously described (section 2.18.4), to a density of 250 plaque forming units (pfu) per plate, and probed as described in section 2.14.1 with the different cDNA clones. Plaques which showed

a positive signal were removed using a Pasteur pipette and replated to a lower density until a single plaque could be identified as a positive clone. Several purified positive clones screened with a single cDNA probe were PCR amplified using primer 1 (JB 1) and subjected to agarose gel electrophoresis.

## 2.19.3 Screening with Non-PCR amplified cDNA

To establish that the PCR amplified cDNA libraries contained cDNA clones representing transcripts found within plant tissue single stranded sugar beet apical meristem cDNA was prepared as described in section 2.17 and used to probe the PCR amplified cDNA libraries in the manner described in section 2.14.

performed with a number of plaques which initially did not show a signal with either probe.

## 2.20 Isolation of Apex-Specific cDNA Clones

#### 2.20.1 Differential Screening

The PCR amplified cDNA libraries were titred as previously described (section 2.18.4) and phage plaques were plated on L-agar in 0.7% (w/v) top agarose (without IPTG and X-Gal) to a relatively low density (1000 pfu per plate) and incubated for 8-10 h. After this time plaques had begun to form but were not overlapping. Four separate lifts were taken from each plate as described in section 2.14.1. After incubation in the pre-hybridisation solution for 2-4 h two of the filters from each plate were incubated with PCR amplified leaf cDNA probe, and the other two were incubated with PCR amplified apical meristem cDNA. The specific activity of each probe was measured using a scintillation counter and an

approximately equal number of cpm of each probe were used to screen the appropriate filters. The filters were incubated and washed as previously described (section 2.14.1). Once the filters were no longer giving more than 10 cps, as measured by a Geiger-Müller counter, they were exposed to Kodak autoradiography film for the time indicated.

Any plaques showing an enhanced positive signal with the PCR amplified apical meristem probe were removed with the sharp end of a Pasteur pipette and stored in 500  $\mu$ l of SM phage buffer (described in section 2.4) containing 1% (v/v) chloroform. This solution was left at room temperature for 1 h after which a 3  $\mu$ l aliquot was removed and dispensed onto a lawn of the appropriate strain of *E.coli*. The spots were allowed to dry for 30 min before being incubated at 37°C for 8-10 h. After this time, in the area where the aliquot of phage buffer was dispensed, a "giant plaque" could be seen, brought about by the presence of many phage particles concentrated into a relatively small area. This procedure was also performed with a number of plaques which initially did not show a signal with either probe. This procedure is referred to as "cold plaquing" and is described below.

## 2.20.2 Screening of 'Cold' Plaques (Scott et al., 1991)

It has been demonstrated that a percentage of phage plaques not showing a positive result with either probe may contain sequences represented at a very low level in the positive probe. For this reason a number of random 'cold' plaques were examined after first amplifying the inserts by producing "giant plaques" as described above. This amplified DNA was then screened as before. Any clones which displayed a positive result with the apical meristem probe were used to probe a northern blot as previously described (section 2.14.4).

#### 2.20.3 Production of a Subtracted Probe

The use of differential probes produces a high percentage of background positive plaques common to both species of probe. To eliminate these common clones a subtractive procedure was used incorporating biotin and avidin (a protein with four high affinity binding sites for biotin). Both the avidin and biotin were purchased from Gibco/BRL, UK Limited. A population of cDNA labelled with biotin, as previously described, was subjected to the following subtraction procedure.

The biotin labelled leaf cDNA was resuspended in H<sub>2</sub>O to a concentration of 0.5  $\mu$ g/ $\mu$ l. 2.5  $\mu$ g of this leaf cDNA was mixed with 25 ng of apical meristem PCR amplified cDNA (trace labelled with [ $\alpha$ -<sup>32</sup>P] dCTP) in 5  $\mu$ l of HS solution:

1.5 M NaCl
 50 mM HEPES (pH 7.5)
 10 μM EDTA (pH 8.0)
 0.2% (w/v) SDS

The cDNA in the HS solution was overlaid with 50  $\mu$ l of paraffin oil and incubated at 95°C for 2 min. The denatured DNA solution was then incubated at 65°C for at least 20 h. After this latter incubation the renatured cDNA mixture was diluted with 50  $\mu$ l of HE solution (HS solution minus the NaCl and SDS) and incubated at 55°C for 20 min to allow cDNA species to anneal completely. NaCl (2 M) was then added to a final concentration of 0.5 M to aid the binding of the avidin molecules to the biotin labelled cDNA. 5  $\mu$ g of avidin was then added and the solution was incubated at room temperature for 20 min. After this final incubation the solution was phenol extracted and ethanol precipitated as described in sections 2.8.3/2.8.4. The radiolabelled precipitated cDNA remaining was air dried and

## Chapter 2: Materials and Methods

resuspended in 5  $\mu$ l of HS solution. The above procedure was repeated up to six times. After every second round of subtraction a 1  $\mu$ l aliquot of precipitated cDNA was removed and subjected to 35 rounds of PCR amplification with primer 1 as described in section 2.15.

## Development of Methods to Produce PCR Amplified cDN. Libraries from Beta reiseris.

As described in section 1.10 the initial objective of this study was to make CDNA liberatics representing genes expressed in the shoot apex of *Beta valga*ris. Because of the very small amount of RNA likely to be obtained from this tissue PCR was employed to arabily cDNA prior to cloning. This section, therefore, describes the development of the technologies employed in the construction of PCR amplified cDNA libraries from small encourse of *Beta* valgaris tissue. To establish the conditions necessary for such library construction experiments were initially performed with the more abandant leaf RNA.

It is essential when synthesising cDNA that the RNA sample used is clean, and not degraded. Most commonly an absorption spectrum is used to demonstrate that the peak of absorbance is at a value of 260 nm and that the A260/280 ratio is within the departed most (between 1.8 and 2.0). To establish the degree of degradation of the RNA setundary of this RNA sample on a denaturing agarose get alongside RNA size markets is make partial Fig. 3.1 shows a 10 is sample of leaf CHAPTER 3 such a get alongside 0.24-9.5 kb badder RNA markers (Orboo/BRL). From the get in Fig. 3.1 is is apparent that the 185 and 255 fRNA bands have a molecular size of 1.5-1.7 Kb and 2.8-0.0 Kb respectively. These values are within the expected size ran **RESULTS** indicate that substantial degradation

# 3.1 Development of Methods to Produce PCR Amplified cDNA Libraries from *Beta vulgaris*.

As described in section 1.10 the initial objective of this study was to make cDNA libraries representing genes expressed in the shoot apex of *Beta vulgaris*. Because of the very small amount of RNA likely to be obtained from this tissue PCR was employed to amplify cDNA prior to cloning. This section, therefore, describes the development of the techniques employed in the construction of PCR amplified cDNA libraries from small amounts of *Beta vulgaris* tissue. To establish the conditions necessary for such library construction experiments were initially performed with the more abundant leaf RNA.

# 3.1.1 DIRNA ithesis kit supplied by Amersham. The values chosen were based on data from

It is essential when synthesising cDNA that the RNA sample used is clean, and not degraded. Most commonly an absorption spectrum is used to demonstrate that the peak of absorbance is at a value of 260 nm and that the A260/280 ratio is within the expected range (between 1.8 and 2.0). To establish the degree of degradation of the RNA separation of the RNA sample on a denaturing agarose gel alongside RNA size markers is more useful.

Fig 3.1 shows a 10  $\mu$ g sample of leaf RNA separated on such a gel alongside 0.24-9.5 kb ladder RNA markers (Gibco/BRL). From the gel in Fig 3.1 it is apparent that the 18S and 25S rRNA bands have a molecular size of 1.6-1.7 Kb and 2.8-3.0 Kb respectively. These values are within the expected size range and, therefore, indicate that substantial degradation of the leaf RNA sample shown has not occurred.

## 3.1.2 cDNA Synthesis

Having established that the RNA was sufficiently intact, it was possible to proceed with the production of double stranded cDNA using total RNA as indicated in section 2.17. It was decided not to enrich for poly  $(A^+)$  RNA here since that would have involved one further step in the method which could have led to loss of material. As the eventual aim of this study was to use small amounts of apical meristem RNA this step was seen as undesirable. The use of total RNA was not expected to present a problem since the cDNA synthesis procedure incorporated oligo dT as a primer for synthesis of the first strand of the cDNA, selecting for poly  $(A^+)$  RNA. Samples containing 20, 50 and 100 µg of total RNA were, therefore, used to establish the optimum RNA concentration for cDNA synthesis using the cDNA synthesis kit supplied by Amersham. The values chosen were based on data from Lu and Werner (1988) who stated that approximately 70 µg of total RNA gave optimum cDNA synthesis in their system.

Fig 3.2 shows an increase in radiolabelled cDNA synthesised with an increase in concentration of total leaf RNA from 20-50  $\mu$ g, but 100  $\mu$ g of total RNA gave no further increase in cDNA synthesised, and actually appeared to inhibit the process. Therefore, in all subsequent experiments 50  $\mu$ g of total RNA was used in the synthesis of cDNA.

## Fig 3.1

A comparison of the size of the *Beta vulgaris* leaf RNA (lane 1) with 0.24-9.5 kb RNA ladder markers (Gibco/BRL). The ladder is a series of six synthetic poly (A) tailed RNAs, five of which are shown here (lane M).



# Fig 3.2 Beta vulgaris Leaf cDNA Synthesised from Total Leaf RNA

Autoradiograph of a blot of samples of single and double stranded cDNA (internally radio-labelled with <sup>32</sup>P) synthesised from increasingly higher quantities of total RNA from *Beta vulgaris* leaf tissue (as described in section 2.17). The samples were separated on a 1.5% (w/v) denaturing agarose gel then transferred by blotting onto Hybond-N nylon membrane (Amersham) before being exposed to X-ray film for 48 h. The resultant autoradiograph is represented here.

Lane M,  $\lambda$  DNA, digested with *Hind* III and end labelled with [ $\alpha$ -<sup>32</sup>P] dCTP

- Lanes 1&2, double and single stranded cDNA respectively, prepared from 20 µg of Beta vulgaris leaf RNA.
- Lanes 3&4, double and single stranded cDNA respectively, prepared from 50 µg of Beta vulgaris leaf RNA
- Lanes 5&6, double and single stranded cDNA respectively, prepared from 100 µg of Beta vulgaris leaf RNA



Ka from all three samples into sine of the first and cDNA the reverse DNA had teen of DNA. This would nove figulitated lippings int the double samador dreed nucleotale pairs fits range was enabled bis range was enabled in the same sine size of a of sherease size size of a najor factor in this

clision was the fact that the size of cDNA produced by PCR is limited to 3 to 4 kb.

Employation of Leaf cDNA on a Separates G-50 Column

Since the process of PCR is known to preferentially amplify DNA of email size (a see losse pairs up to 500 bp), the lenf cDNA was passed dows a Scohades Q-50 column which would also serve to remove unincurpristed nucleondes from the synthesis reaction. A likes about from each of the 200 at fractions passed through the column which were

The size of the cDNA synthesised is another important factor in the production of representative cDNA libraries. In general the cDNA synthesised should range in size from a few hundred nucleotide pairs to 8-10 kbp, the bulk of which should lie between 1 and 2 kbp. Fig 3.2 shows the size of both the first strand and second strand cDNA from all three samples of leaf RNA. In each case there was no significant difference between the size of the first and second strand cDNA indicating that during synthesis of the first strand cDNA the reverse transcriptase enzyme had not continued on after the full length of the cDNA had been synthesised and started the transcription of a "snapped back" segment of DNA. This would have resulted in cDNA with a hairpin loop at one end, and would not have facilitated ligation of the linkers required for the cloning procedure. Fig 3.2 also shows that the double stranded cDNA produced in this experiment ranged in size from less than a hundred nucleotide pairs to 2 kbp, with the majority of the material in the range of 0.5 - 1 kbp. This range was smaller than expected but was within the right order of magnitude. Attempts to increase the size of the cDNA by altering the enzyme concentration and the timing of each of the steps in the synthesis procedure were not successful. Therefore, it was decided to proceed with the library construction using the sample shown in lane 3 of Fig 3.2. One major factor in this decision was the fact that the size of cDNA produced by PCR is limited to 3 to 4 kb.

## 3.1.3 Fractionation of Leaf cDNA on a Sephadex G-50 Column

Since the process of PCR is known to preferentially amplify DNA of small size (a few base pairs up to 500 bp), the leaf cDNA was passed down a Sephadex G-50 column which would also serve to remove unincorporated nucleotides from the synthesis reaction. A 10  $\mu$ l aliquot from each of the 200  $\mu$ l fractions passed through the column which were

observed to be radioactive (as measured by scintillation counting) was removed and analysed by electrophoresis through a 1.5% (w/v) agarose gel. The gel was subsequently dried in a Bio-Rad gel dryer at  $60^{\circ}$ C and exposed to X-ray film. Fig 3.3 shows the proportion of cDNA to unincorporated radiolabel within each of these aliquots. Fractions 5, 6 and 7 (lanes 1,2, and 3 of Fig 3.3) contained no unincorporated radiolabel and a large proportion of cDNA species of a size greater than 500 bp. The remaining fractions contained unincorporated nucleotides and a high proportion of very small cDNAs. Therefore the fractions shown in lanes 1,2, and 3 were pooled together and prepared for the ligation of the appropriate linkers.

## 3.1.4 Development of the techniques for PCR amplification of cDNA Suitable for Cloning.

## 3.1.4.1 Oligonucleotide Linkers

Linkers were designed (Fig 3.4a) which would enable PCR amplification of the cDNA ligated to those linkers. The oligonucleotides JB1 and JB2 were annealed to produce the required linkers. JB1 contained an *Eco* RI restriction site within its 29 nucleotide sequence to facilitate subsequent cloning of the cDNA into the chosen vector ( $\lambda$  ZAP II). The smaller oligonucleotide (JB2), was not phosphorylated at its 5' end, and would therefore not ligate to the 3' end of the double stranded cDNA. However, when annealed to JB1 the smaller 12-mer oligonucleotide was held in place by its complementary base pairing. JB1 would ligate to the 5' end of the double stranded cDNA since this cDNA was phosphorylated in the process of blunt-ending using the Klenow fragment of DNA polymerase I. The fact that

# Fig 3.3 Size Fractionation of Leaf cDNA on a Sephadex G-50 column.

A dried gel of 10  $\mu$ l aliquots from 200  $\mu$ l fractions of leaf cDNA from lane 3 in Fig 3.2 after being passed through a Sephadex G-50 column. The gel was dried at 80°C under vacuum after precipitation of the nucleotides on the gel with 7% TCA and then exposed to X-ray film for 24 h.

Lane M,  $\lambda$  DNA, digested with *Hind* III and end labelled with [ $\alpha$ -<sup>32</sup>P] dCTP Lanes 1-6, Fractions 5-10 from the Sephadex G-50 column. Size in bn 4.361 2,322 2,027

546



JB2 was not ligated to the cDNA meant that at 72°C this smaller oligonucleotide would melt away from its complement in JB1. The *Taq* Polymerase enzyme would fill in the single strand remaining after removal of JB 2 and would produce a full length complement to JB1. On denaturing at 94°C the double stranded cDNA would melt apart. At one end of each strand of cDNA would be a synthesised sequence complementary to JB1. Therefore, the JB1 oligonucleotide was able to act as a primer for each of the strands of denatured cDNA. The sequence of events in the PCR amplification procedure is shown in Fig 3.4b.

### 3.1.4.2 Ligation of Linkers to Control DNA

Before ligation of the linkers to the leaf cDNA it was essential to show that the linkers described above would ligate to blunt-ended double stranded DNA. Using  $\lambda$ /Hind III marker DNA that was not blunt-ended, and some that had been subjected to a blunt-ending reaction using the Klenow fragment of DNA polymerase I, four separate treatments were performed to test the suitability of the ligase enzyme, and to show that the oligonucleotide linkers would ligate to the marker DNA. The results are shown in Fig 3.5 and demonstrate that when no linkers were present the ligase enzyme was able to ligate the non-blunt-ended marker DNA. This demonstrated that the ligase enzyme was working. However, the efficiency of ligation of the blunt-ended marker DNA was lower than that for the non-blunt-ended DNA. The most interesting finding from Fig 3.5 was that in the presence of the JB linkers self ligation of the marker DNA was reduced substantially. This could have been caused by the linkers being ligated to the marker DNA and inhibiting self ligation. Due to the non-complementary arrangement of the overhang left by JB1, when annealed to JB2, ligation of the linkers to themselves should not occur, as is demonstrated in lane 4 of Fig 3.5.

## Fig 3.4 Oligonucleotide Linker Sequence and PCR Protocol Employed in this Study

a) shows the sequence of the two oligonucleotides used in this study and the position of the *Eco* RI restriction site

b) shows the sequence of events involved in the PCR protocol employed in this study.
3.4 i). Once the oligonucleotide linkers have been ligated to the double stranded cDNA of interest the PCR reaction is heated to 72°C. This allows the smaller linker (JB 2) to melt off and *Taq* polymerase fills in the gap left by this linker.
3.4 ii) The double stranded cDNA is then denatured at 94°C
3.4 iii) & iv) The larger oligonucleotide JB 1 acts as a primer for *Taq* polymerase to synthesise a new complementary strand of cDNA

a)	Eco RI JB1 5' ATGCCTAG <u>G/AAT</u>				
	JB2	3' AAATCGGAGTAT 5'			
b)	JB1 5'3' 5' 3' 5' 3' JB2	Double Stranded cDNA	3' 5' 3'		
3.4i		Heat to 72 <sup>0</sup> C for 5 min Double stranded cDNA	3' 5' 3'		
3.4ii		Denature at 94 <sup>o</sup> C single stranded cDNA			
3.4.iii	JB1 5'3' 5'	Cool to 68 <sup>0</sup> C for 1 min single stranded cDNA	JB1 primer 3'5' 3' 5'3'		
The JB1 oligonucleotide primer anneals to the JB1 complement produced in step 3.4i by the melting off of JB2 and filling in of the overhang by <i>Taq</i> polymerase					
3.4 iv	3'5' 3'	Heat to 72 <sup>o</sup> C for 1.1 min double stranded cDNA			

Repeat steps 3.4 ii) iii) and iv)for 35 cycles before finally heating to 72°C for 10 min

## Fig 3.5 Ligation of Linkers to Control DNA

Photograph of an ethidium bromide stained 1.5 % (w/v) agarose gel of ligation reactions using  $\lambda$ /Hind III DNA ( $\lambda$  DNA digested with Hind III) treated as follows;

Lane 1, untreated control  $\lambda$ /Hind III. i.e.  $\lambda$  DNA digested with *Hind* III.

Lane 2,  $\lambda$  DNA digested with *Hind* III and then ligated.

Lane 3,  $\lambda$  DNA digested with *Hind* III, blunt-ended and then ligated.

Lane 4, λ DNA digested *Hind* III, blunt-ended and ligated in the presence of an excess of JB1 and JB2 oligonucleotide linkers
size in bp

23,130 9.416 6,557 4,361 samples, even though the start 2,322 2.027 the contamination of the larger marke



that had been subjected to ligation in the whe'0.5 km DNA by electrophoresis, or lanation seems male nolifying the 4 kbp

## <u>3.1.4.3 PCR Amplification of λ Hind III marker DNA, Blunt-Ended</u> and Ligated to the JB Linkers

The results described in section 3.1.4.2 above indicate that the JB linkers were successfully ligated to the marker DNA, but as a definitive method of demonstrating this fact two marker DNA fragments (0.5 and 4 kbp in size) that had been subjected to ligation in the presence of the JB 1 and JB 2 linkers were isolated by the method described in section 2.8.6. Aliquots of these two samples were subjected to PCR using JB 1 as a primer, as described in section 2.18 and shown schematically in Fig 3.4b. Fig 3.6 shows that the 0.5 kbp DNA fragment (actually 622 bp after ligation of the primers) was preferentially amplified in both samples, even though the starting material should not have contained this lower sized marker when the 4 kbp marker DNA fragment was subjected to PCR. This may have been caused by the contamination of the larger marker DNA fragment with a small quantity of the 0.5 kbp fragment of DNA due to incomplete separation of the marker DNA by electrophoresis, or contamination at some other stage of the experiment. The latter explanation seems more likely since it is only the 0.5 kbp fragment of marker DNA, and not the 2.0 kbp nor 2.3 kbp fragment which is contaminating the sample produced after PCR amplifying the 4 kbp marker DNA sample. This demonstrates a point made earlier, that PCR preferentially amplifies small sized DNA, and any small sized DNA present in a sample would not be representatively amplified. Besides emphasising this important point, the experiment shown in Fig 3.6 also demonstrates that the system described in Fig 3.4 was feasible, since amplification of both of the marker DNA fragments isolated was observed.

## Fig 3.6 PCR Amplification of Control Marker DNA with JB1 and JB2 Oligonucleotide Linkers Ligated.

Photograph of an ethidium bromide stained 1.5% (w/v) agarose gel showing the products of 35 rounds of PCR amplification of aliquots of samples containing 0.5 and 4 kbp fragments of  $\lambda$  DNA digested with *Hind* III.

- Lane M, 1 kbp "ladder" marker DNA (Gibco/BRL). The bands of the ladder each contain 1-12 repeats of a 1,018 bp DNA fragment. In addition the ladder also contains additional DNA fragments that range from 75-1,636 bp
- Lane 1, DNA produced after PCR amplification of a 2.5 µl aliquot of 0.5 kbp marker DNA with the JB 1 and JB 2 linkers ligated
- Lane 2, DNA produced after PCR amplification of a 2.5 µl aliquot of a 1:10 dilution of the 0.5 kbp marker DNA with the JB 1 and JB 2 linkers ligated
- Lane 3, DNA produced after PCR amplification of a 2.5 µl aliquot of 4 kbp marker DNA with the JB 1 and JB 2 linkers ligated
- Lane 4, DNA produced after PCR amplification of a 2.5 µl aliquot of a 1:10 dilution of the 4 kbp marker DNA with the JB 1 and JB 2 linkers ligated

Size bp 1,636 1,018 506/517 396 344 298



came off the column later. The range of the size of the amplitude GDEAN was range and hundred bp to over 2 kbp (as measured against 1 kbp "ladder" marker DNA). The range and size of the cDNA in Figs 3.7 and 3.8 are very similar indicating that the amplification was accurate. Therefore the samples shown to contain a small propriation of small cDNA species (lanes 1,2 and 3 from Fig 3.8) were pouted and used for cloning into A ZAP II. The remaining 87.5 µl of leaf cDNA material from cach of these 3 fractions was stored for later amplification as required. Therefore, the results from Fig 3.8 demonstrate that from 50 µg of total leaf RNA an unlimited supply of cDNA was available.

#### 3.1.4.4 Ligation of Linkers to the cDNA

Having established that the linker ligation was successful with the control marker DNA, and that the linkers could be successfully used for the amplification of DNA by PCR using the protocol described in section 2.18, the next step was to ligate the linkers to the leaf cDNA. To eliminate any remaining cDNAs of a small size, the leaf cDNA ligation mix was passed through a Bio Gel A-150 column. A small aliquot was removed from each of the 100  $\mu$ l fractions from the column shown to be radioactive and subjected to electrophoresis and Southern blotting. Fig 3.7 shows that the DNA from fractions 6,7, and 8 from the column (lanes 1,2, and 3, in Fig 3.7) contained cDNA of an acceptable size distribution (150 bp - 6 kbp). Other fractions contained increasingly smaller cDNA. A further 2.5 µl aliquot was taken from each fraction shown in Fig 3.7 and amplified using the same protocol as for the control marker DNA. Each fraction contained amplifiable material, as shown in Fig 3.8, with a general trend towards a higher proportion of small cDNA species within the fractions that came off the column later. The range of the size of the amplified cDNA was from a few hundred bp to over 2 kbp (as measured against 1kbp "ladder" marker DNA). The range and size of the cDNA in Figs 3.7 and 3.8 are very similar indicating that the amplification was accurate. Therefore the samples shown to contain a small proportion of small cDNA species (lanes 1,2 and 3 from Fig 3.8) were pooled and used for cloning into  $\lambda$  ZAP II. The remaining 87.5 µl of leaf cDNA material from each of these 3 fractions was stored for later amplification as required. Therefore, the results from Fig 3.8 demonstrate that from 50 µg of total leaf RNA an unlimited supply of cDNA was available.

92

#### Fig 3.7 Ligation of Linkers to the Leaf cDNA

An autoradiograph of a blot of leaf cDNA that had been size fractionated through a Bio Gel A-150 column after ligation of the JB1 and JB2 oligonucleotide linkers. From each 100  $\mu$ l fraction collected from the column and shown to be radioactive (as measured using a Geiger-Müller counter), a 10  $\mu$ l aliquot was removed and separated on a 1.5% (w/v) agarose gel. The gel was blotted onto Hybond-N nylon and exposed to X-ray film for 48 h.

Lane M,  $\lambda$  DNA, digested with *Hind* III and end labelled with [ $\alpha$ -<sup>32</sup>P] dCTP Lanes 1-11, Fractions 6-16 from the Bio Gel A- 150 column



### Fig 3.8 PCR Amplification of Leaf cDNA

Photograph of an ethidium bromide stained 1.5% (w/v) agarose gel containing the products of PCR amplification of  $2.5 \mu$ l aliquots of fractions 6-16 from the Bio-Gel A-150 column containing linker-ligated leaf cDNA. The aliquots of leaf cDNA were subjected to 35 cycles of PCR using the JB 1 oligonucleotide primer and 10  $\mu$ l aliquots of these products were separated on an agarose gel.

Lanes 1-11, PCR amplified cDNA products obtained on subjecting 2.5 µl aliquots of solution from lanes 1-11 in Fig 3.7 to 35 cycles of PCR
Lane M, 1 kb "ladder" marker DNA (Gibco/BRL)



the DNA cluted from these strips screating as the method described to sociole 2.8.6. An post of each fraction from the get was extimined on a second 1.5% (w/v) against get Fig shows the size of the cDNA contained within each of the five series of get measured tent 1 kbp DNA. "ladder" markets. There is an obvious decrease in the size of the cDNA is fraction 2-4. There was no detectable DNA is fraction 1. The cDNA is fraction 2 flate that 3, 10) contained cDNA of a size greater than 300 bp extending to 1 kbp.

## 3.1.5 Development of the Cloning Procedure for PCR Amplified cDNA.

#### 3.1.5.1 Eco R1 Digestion of the Leaf cDNA

For the PCR amplified leaf cDNA to be inserted into the *Eco* RI digested  $\lambda$  ZAP II vector, the *Eco* RI restriction site within the JB linkers had to be digested. To demonstrate that the digestion had been successful digested PCR amplified leaf cDNA was self ligated using T4 DNA ligase. If the digestion was successful the leaf cDNA would have been able to ligate to itself, and thus there would have been an increase in the average size of the leaf cDNA. This is shown to have occurred in Fig 3.9. The leaf cDNA in lane 3 is of an overall larger size than that in lane 2 which was not subjected to *Eco* RI digestion.

#### 3.1.5.2 Removal of Small Sized cDNA Prior to Cloning

In a final attempt to eliminate small cDNA species from the PCR amplified leaf cDNA sample a final size fractionation experiment was performed. The *Eco* RI digested PCR amplified leaf cDNA was subjected to electrophoresis for 20 min before being sectioned into 4 mm wide horizontal strips by cutting the gel at right angles to the direction of migration, and the DNA eluted from these strips according to the method described in section 2.8.6. An aliquot of each fraction from the gel was examined on a second 1.5% (w/v) agarose gel. Fig 3.10 shows the size of the cDNA contained within each of the five strips of gel measured against 1 kbp DNA "ladder" markers. There is an obvious decrease in the size of the cDNA from fraction 2-4. There was no detectable DNA in fraction 1. The cDNA in fraction 2 (lane 2 of Fig 3.10) contained cDNA of a size greater than 300 bp extending to 1 kbp.

#### Fig 3.9 Eco RI Digestion of PCR Amplified Leaf cDNA

Photograph of an ethidium bromide stained 1.5% (w/v) agarose gel containing a sample of leaf cDNA digested with *Eco* RI restriction enzyme and subjected to ligation with T4 DNA ligase compared with leaf cDNA not digested with this *Eco* RI but subjected to the same ligation conditions.

Lane 1, λ DNA digested with *Hind* III
 Lane 2, Leaf cDNA not subjected to *Eco* RI digestion prior to incubation with T4 DNA ligase
 Lane 3, Leaf cDNA subjected to *Eco* RI digestion prior to incubation with T4 DNA ligase



## Fig 3.10 Removal of Small Sized Leaf cDNA Prior to Cloning

*Eco* RI digested PCR amplified leaf cDNA was subjected to size fractionation by separation through a 1.5% (w/v) agarose gel and then sectioning of that gel into 4mm strips at right angles to the direction of migration of the DNA. Aliquots of the DNA isolated from these strips were run on a second 1.5% (w/v) agarose gel stained with ethidium bromide, shown here.

## Lane M, 1kb "ladder" marker DNA(Gibco/BRL)

Lanes 1-4, DNA products isolated from sequential 4mm strips from the first agarose gel

This fraction was thought to be most suitable for the subsequent cloning procedure and was, therefore, used for lightion to the  $\lambda$  ZAP II vector.



with 250 ng of leaf PCR amplified shakes wish when meaningeon clones with a time of of Ix 10<sup>6</sup> clones within the library.

This series of experiments, therefore, destributing of that the techniques used have were successful in producing a PCR ampirited cDNA library from small quantities of Beta sulgaris leaf total RNA.

1.1.6 Characterisation of the Leaf PCR Apphilied cDWA Library

Since thermal cycling is able to produce large appoints of DNA from very south

This fraction was thought to be most suitable for the subsequent cloning procedure and was, therefore, used for ligation to the  $\lambda$  ZAP II vector.

## 3.1.5.3 Ligation of PCR Amplified cDNA to $\lambda$ ZAP II Phage Vector

Since the PCR amplified leaf cDNA was demonstrated to have successfully been digested with *Eco* RI it was subjected to ligation in the presence of the phage vector  $\lambda$  ZAP II (supplied by Stratagene as an *Eco* RI digested stock). According to the recommendation of Stratagene two separate ligations were performed with the leaf cDNA using 50 and 250 ng of DNA. The relevant control was also performed to ensure the vector was able to accept a test insert of DNA. To see how successful the ligation was, the vector was next packaged into its necessary protein coat as described in section 2.18.3. Each packaging reaction was diluted before plating out in the presence of X-Gal and IPTG to establish the titre of the library. Table 3.1 shows that in each case the ligation and packaging reactions were successful and that the greatest recombination frequency and titre were observed in the library produced with 250 ng of leaf PCR amplified cDNA with 80% recombinant clones and a titre of 1x 10<sup>6</sup> clones within the library.

This series of experiments, therefore, demonstrated that the techniques used here were successful in producing a PCR amplified cDNA library from small quantities of *Beta vulgaris* leaf total RNA.

#### <u>3.1.6</u> Characterisation of the Leaf PCR Amplified cDNA Library

Since thermal cycling is able to produce large amounts of DNA from very small

# amounts of starting material dess tha Table 3.1 and the possibility of contamination with

## Amount of DNA packaged No. of clones obtained Recombination frequency

cal RNA. For instan

1x10<sup>6</sup>

cDNA. It was also necessary to establish if the cDNA library was representative of 250 µg leaf PCR amplified cDNA

 $50 \ \mu g \ leaf \ PCR \qquad 1.5 \times 10^5 \qquad 60\%$ amplified cDNA abundance of c

DNAs in the population. Hence, the library was screened with probes for

80%

it is possible that the library

100  $\mu$ g test insert  $1 \times 10^8$  95% supplied by Strategene

Properties of cDNA libraries in  $\lambda$  ZAP II prepared from different Table 3.1 amounts of PCR amplified leaf cDNA compared with a test insert. No clones were obtained in a control packaging reaction with no insert present.

amounts of starting material (less than a few pg), and the possibility of contamination with foreign DNA cannot be dismissed (despite the rigorous controls used), it was essential to establish that the material cloned into the phage vector was PCR amplified *Beta vulgaris* leaf cDNA. It was also necessary to establish if the cDNA library was representative of transcripts present in the leaf RNA. For instance, it is possible that the library might be enriched in clones representing small transcripts because the PCR reaction favours small DNA fragments. Moreover, it is possible that PCR amplification distorts the relative abundance of cDNAs in the population. Hence, the library was screened with probes for known genes whose transcripts were assumed to be of different size and abundance in the leaf RNA.

#### 3.1.6.1 Examination of Random Plaques

To show that the PCR amplified leaf cDNA had inserted into the vector, as indicated by the blue/white selection, 10 plaques were taken at random (including 2 blue plaques as a control) and amplified using oligos PUC 1 and PUC 2 (as described in section 2.15) designed to amplify inserts from any vector related to the pUC plasmid vector. Fig 3.11 shows the DNA that was amplified from each of these random plaques. The blue plaques produced a DNA product of about 200 bp which is caused by the fact that the PUC 1 and PUC 2 primer oligonucleotides flank the multiple cloning site of pUC related vectors leaving 100 bp of vector DNA available to be amplified. Taking this into account the average size of the insert DNA from the remaining 8 plaques was between 500 and 600 bp and ranged up to 1.3 kbp, as measured against 1 kbp "ladder" DNA markers. Therefore the procedure used here was shown to produce inserts of a reasonable size distribution.

# Fig 3.11 Examination of Inserts from Random Plaques of the Leaf cDNA Library.

 $28 \ \mu$ l aliquots from the SM storage buffer of 10 random clones were subjected to PCR amplification using the PUC1 and PUC 2 oligonucleotides. 10  $\mu$ l aliquots of these products were separated on a 1.5% (w/v) agarose gel stained with ethidium bromide.

Lanes M, 1 kbp "ladder" marker DNA (Gibco/BRL)

- Lanes 1&10, DNA products isolated from plaques showing a blue colour when plated out in the presence of X-Gal and IPTG and subjected to PCR amplification with PUC 1 and PUC 2 oligonucleotide primers
- Lanes 2-9, DNA products isolated from random plaques showing a white colour when plated out in the presence of X Gal and IPTG and subjected to PCR amplification with PUC1 and PUC 2 oligonucleotide primers



used to screen the leaf library. These were the LHCP close, the or-tubulin close and enter

within the leaf library that were hybridising us a close of som offers, and the source of the library to be of leaf origin diemselves. They want this present as a present how been library which was expected from the nature of the close and the data from Cartage source of (1981) who reported that LHCP mRIVA was approximately 2% of the loss insertion of A<sup>+</sup>) RNA in illuminated per leaves. To earling the sumber of true 1962 cover we can library 14 madom pursitive point we LHCP closes were plated out individuality of the second sec

#### 3.1.6.2 Screening of the Leaf Library with Known DNA Probes

To show that the leaf cDNA was representatively amplified by PCR, and as such was representative of the transcripts present in the leaf RNA, three clones of plant origin were used to screen the leaf library. These were the LHCP clone, the  $\alpha$ -tubulin clone and clone PTA 71 (described in section 2.19.2). This screening would establish whether the leaf library contained inserts of plant origin, and that the particular species of cDNA were present in the library at the expected frequency.

To establish the frequency of gene transcripts complementary to the LHCP DNA probe in the *Beta vulgaris* leaf tissue, RNA isolated from the leaf tissue used to make the leaf library was fractionated on a formaldehyde-containing agarose gel (described in section 2.8.5), blotted as described in section 2.14.4 and hybridised to the LHCP cDNA of *Zea mays*. Fig 3.12 demonstrates that the *Zea mays* LHCP cDNA hybridises to a single transcript 1400 nucleotides long (as determined using RNA markers supplied by Gibco BRL) in *Beta vulgaris* leaf RNA, and is not detectable in the root RNA.

On screening 1,000 plaques from the leaf cDNA library with the Zea mays LHCP DNA probe it was observed (Fig. 3.13) that approximately 5% of the cDNA clones within the leaf library showed strong hybridisation. This demonstrated the presence of cDNA clones within the leaf library that were hybridising to a clone of leaf origin, and therefore, were likely to be of leaf origin themselves. They were also present at a relatively high frequency which was expected from the nature of the clone and the data from Cuming and Bennett (1981) who reported that LHCP mRNA was approximately 2% of the total translatable poly (A<sup>+</sup>) RNA in illuminated pea leaves. To estimate the number of true LHCP clones within the leaf library 14 random putative positive LHCP clones were plated out individually at a lower

# Fig 3.12 Northern Blot of *Beta vulgaris* Leaf and Root RNA Probed with the LHCP cDNA Probe

10  $\mu$ g of *Beta vulgaris* leaf RNA and 30  $\mu$ g of *Beta vulgaris* root RNA were separated on a 1.5% (w/v) denaturing agarose gel. The gel was blotted onto Hybon-N nylon membrane (Amersham) and hybridised with the radiolabelled *Zea mays* LHCP cDNA clone at 65°C. The nylon membrane was washed three times in 2X SSC, 0.1% (w/v) SDS at 65°C for 20 min before being exposed to X-ray film for 48 h. Lanes M,1 and 2 show the ethidium bromide stained gel prior to blotting and lanes 3 and 4 show the autoradiograph produced from the blotted gel after hybridisation to the LHCP cDNA clone of *Zea mays*.

- Lane M, 0.24-9.5 kb RNA ladder markers (Gibco/BRL). The ladder is a series of six synthetic poly (A) tailed RNAs, three of which are shown here
- Lane 1, 30 µg of annual Beta vulgaris root RNA

Lane 2, 10 µg of annual Beta vulgaris leaf RNA

- Lane 3 transcript of root RNA hybridising to the LHCP cDNA clone of Zea mays
- Lane 4, transcript of annual leaf RNA hybridising to the LHCP cDNA clone of Zea mays

The transcript hybridising to the LHCP clone in lane 3 is of 1.4 kb in size. The fact that no visible hybridisation of the LHCP clone is observed in the root RNA is not surprising since what was assumed to be 30  $\mu$ g of root RNA is clearly less than 10  $\mu$ g when compared with the leaf RNA. This is a common problem experienced with sugar beet root RNA and subsequent experiments should have been performed using greater quantities of root RNA. However, the lack of a signal in the lane containing the root RNA after hybridisation with the LHCP probe is indicative of the lack of transcripts encoding LHCP in the roots of *Beta vulgaris*.

where the problem of the second screen with the LHCP DNA probe and Fig 3.14 shows that 12 of the 14 putative positives gave a positive signal after a second screen. However after a third screen using the LHCP DNA probe, 8 of the patative size in bases M 1 2 3 4

the hybrid sation to leaf RA Since the leaf horn 2,370 2,370 1,350 2,370 1,350 2,370 1,350 2,370 1,350 2,370 1,350 2,370 1,350 2,370

Same 1,000 clones with the file
Basi fibrary likeles on of field, and it was monored field
Sweren, second way, or write
Sybridisation to the second system
Stranger in unaccome by
A.
C library in unaccome by

e engless poly (A\*), sunched

estable the Forary was higher or lower winning 1,000 cDNA clones from the his in Fig 3,16. Subjecting 4 of these

putative positive clones to a second resize of screening with the traduction CDMA crobe showed that all 4 were positive to tobulin cDNA clones (Fig 3.17). Therefore, the frequency of this cDNA clone within the leaf library is roughly if %. This is lower than for the LHCP clone, which is to be expected. The above evidence suggests that the leaf library contains CDNA representative of the menocripts present in leaf RNA. density than before. These 14 clones were subjected to a second screen with the LHCP DNA probe and Fig 3.14 shows that 12 of the 14 putative positives gave a positive signal after a second screen. However after a third screen using the LHCP DNA probe, 8 of the putative positives were still hybridising strongly to the probe (data not shown). The 8 positive clones are indicated in Fig 3.14. This reduced the estimated number of positive clones to around 3%. This is still a relatively high frequency and of the same order of magnitude as expected from the hybridisation to leaf RNA (Fig 3.12) and the data of Cuming and Bennett (1981).

Since the leaf library was prepared from total RNA, and not poly (A<sup>+</sup>)- enriched RNA, a ribosomal DNA clone (PTA 71) was used to probe the same 1,000 clones used in the LHCP experiment to establish the frequency of clones within the leaf library likely to be of ribosomal RNA origin. Fig 3.15 shows the result of this experiment, and it was observed that a number of areas looked to contain putative positive clones. However, second screening of several random plaques (data not shown) showed non-specific hybridisation to the ribosomal DNA clone. This demonstrated that the leaf cDNA library did not contain an unacceptably high proportion of cDNA clones originating from ribosomal RNA.

An  $\alpha$ -tubulin DNA probe was also used to probe the leaf library to investigate whether the abundance of the  $\alpha$ -tubulin cDNA clones within the library was higher or lower than that observed for the LHCP cDNA clones. On probing 1,000 cDNA clones from the leaf library 12 putative positives were observed as shown in Fig 3.16. Subjecting 4 of these putative positive clones to a second round of screening with the  $\alpha$ -tubulin cDNA probe showed that all 4 were positive  $\alpha$ -tubulin cDNA clones (Fig 3.17). Therefore, the frequency of this cDNA clone within the leaf library is roughly 1%. This is lower than for the LHCP clone, which is to be expected. The above evidence suggests that the leaf library contains cDNA representative of the transcripts present in leaf RNA.

# Fig 3.13 Screening of the PCR Amplified Leaf Library with the LHCP cDNA Clone from Zea mays

1000 clones from the PCR amplified leaf cDNA library were bound to Hybond-N nylon membrane filters (Amersham) and hybridised to the radiolabelled LHCP cDNA clone from Zea mays at  $65^{\circ}$ C for 24 h. The nylon membrane filters were washed three times in 2X SSC 0.1% (w/v) SDS for 20 min and exposed to X-ray film for 24 h. The resultant autoradiograph is shown here.



# Fig 3.14 Secondary Screening of Putative Positive LHCP Clones.

14 Separate Hybond-N nylon membrane filters (Amersham) were bound to plaques from 14 putative positive LHCP cDNA clones plated out at a relatively low density. The filters were probed with the radiolabelled LHCP cDNA clone from Zea mays at  $65^{\circ}$ C for 24 h. The filters were washed three times in 2X SSC 0.1% (w/v) SDS for 20 min before exposure to X-ray film for 24 h. The resultant autoradiograph is shown here. Those clones which were seen to be hybridising to the LHCP cDNA probe after a third round of screening are indicated with arrows.



# Fig 3.15 Screening of the Leaf cDNA Library with a Ribosomal DNA probe

1000 leaf PCR amplified cDNA clones were bound to Hybond-N nylon membrane (Amersham) and hybridised with the radiolabelled wheat ribosomal DNA probe PTA 71 at  $65^{\circ}$ C for 24 h. The nylon membrane filters were washed three times in 2X SSC 0.1% (w/v) SDS for 20 min before exposure to X-ray film for 24 h. The resultant autoradiograph is shown here.



# Fig 3.16 Screening of the Leaf cDNA Library with an α-Tubulin cDNA probe

1000 leaf cDNA clones were bound to Hybond-N nylon membrane filters (Amersham) and hybridised to the radiolabelled  $\alpha$ -tubulin cDNA probe at 65°C. The nylon membrane filers were washed three times in 2X SSC, 0.1% (w/v) SDS at 65°C for 20 min before exposure to X-ray film for 48 h. The resultant autoradiograph is shown here.



# Fig 3.17 Secondary Screening of Putative Positive α-Tubulin Clones.

Plaques from 4 putative positive  $\alpha$ -tubulin clones from the PCR amplified leaf cDNA library were plated out at a relatively low density and bound to Hybond-N nylon membrane (Amersham). The nylon membrane filters were hybridised to the radiolabelled  $\alpha$ -tubulin cDNA probe at 65°C for 24 h. The filters were washed three times in 2X SSC, 0.1% (w/v) SDS at 65°C for 20 min before exposure to X-ray film for 24 h. The resultant autoradiograph is shown here.

## Chapter 3: Results



## 3.2 Preparation of a PCR Amplified cDNA Library from Shoot Apical Meristem Tissue

Having developed a method of PCR amplification and cloning to produce what appears to be a representative cDNA library from 50  $\mu$ g of *Beta vulgaris* leaf RNA it was decided to proceed with the production of a similar library from shoot apical meristem tissue. Taking annual *Beta vulgaris* plants grown in SD conditions at a temperature of 20<sup>o</sup>C (for the reasons explained in section 1.10) 35 shoot apical meristems were successfully dissected for small scale RNA extraction (See section 2.16 and Jepson *et al.* 1991). To demonstrate that the material isolated did contain apical meristem tissue a further two samples were dissected and embedded in wax ready for sectioning as described in section 2.7.4. Fig 3.18 shows a sample of one of these sections examined under the light microscope. The material did contain apical meristem tissue, as well as a proportion of leaf primordium and stem pith tissue. This, however, was due to the need for a larger sample for sectioning than was used to isolate RNA. Therefore, the tissue used to isolate the RNA would have contained very little contaminating tissue although non-meristematic tissue would not necessarily be absent. This tissue would, however, be enriched in RNA from apical meristems.

## 3.2.1 Isolation of RNA from Annual Beta vulgaris Shoot Apical Meristem Tissue

The data presented in Fig 3.2 demonstrate that 50  $\mu$ g of total RNA was required for optimum cDNA synthesis. Shoot apical meristems were isolated from as many healthy, eight week old, annual, *Beta vulgaris* plants, that had experienced the desired growth conditions, as was possible at the one time. From the fifty plants chosen only 37 apical meristems were

# Fig 3.18Examination of a Section of Annual Beta vulgaris Shoot Apical MeristemTissue by Light Microscopy

The tissue was dissected from the sugar beet plant and treated as described in section 2.7.4. The photograph shows the dissected apical meristem tissue at a magnification of x40. The inner box indicates the tissue dissected for use in isolating RNA.

successfully isolated, two of which were sectioned for Fig 3 18. A shash alignet (1%) of the RNA isolated from this spacel meristers there was take on a 1.5% (new equated prior to establish the presence of RNA, and to determine whether this spacel RNA was despresses or not. Fig 3.19 shows that RNA was successfully isolated been this tester, and the at dis and appear degraded. The Scale bands are then 1 mm<sup>-164</sup>.



meristen tissue.

U2.3 Removal of Unincorporated Radiolabelled Nucleother

As was described for the leaf cDNA library (section 5.1.2) the uniterrelated radiolabelled nucleonides were removed from the cDNA reaction hits by passing it decough a Septiadex G-50 column. An autoratiograph of the dried gel of 10 µl alsoute rakes from 200 µl fractions run through the externa is shown in Fig 3.21. Fractions 6.7, and 5 from the
successfully isolated, two of which were sectioned for Fig 3.18. A small aliquot (5%) of the RNA isolated from this apical meristem tissue was run on a 1.5% (w/v) agarose gel to establish the presence of RNA, and to determine whether this apical RNA was degraded or not. Fig 3.19 shows that RNA was successfully isolated from this tissue, and that it did not appear degraded. The visible bands are those of rRNA.

#### 3.2.2 Preparation of Annual Shoot Apical Meristem cDNA

The remaining 95% of the RNA extracted from the dissected apices was used immediately for the preparation of double stranded cDNA. Aliquots of the cDNA produced were analysed in the same manner as for the leaf cDNA, and Fig 3.20 shows the cDNA produced from the apical tissue to be smaller than the expected size range. The majority of the cDNA was within the range 0.2-2 kbp but, as was shown in section 3.1, the small size of the cDNA should not pose a problem. The dried gel in this figure shows that there was a large amount of unincorporated dNTPs remaining after the cDNA synthesis reaction was completed. This indicates that less than 50 µg of total RNA was isolated from the apical meristem tissue.

#### 3.2.3 Removal of Unincorporated Radiolabelled Nucleotides

As was described for the leaf cDNA library (section 3.1.2) the unincorporated radiolabelled nucleotides were removed from the cDNA reaction mix by passing it through a Sephadex G-50 column. An autoradiograph of the dried gel of 10 µl aliquots taken from 200 µl fractions run through the column is shown in Fig 3.21. Fractions 6,7, and 8 from the

# Fig 3.19 RNA Isolated from Annual *Beta vulgaris* Shoot Apical Meristems Exposed to Non-Floral Inductive Conditions (Short Days and Warm Temperatures)

The RNA from 35 annual *Beta vulgaris* warm, short day treated apical meristems was isolated as described in section 2.16 and Jepson *et al.* (1991). Of the 10  $\mu$ l of RNA solution obtained 0.5  $\mu$ l was run on a 1.5% (w/v) agarose gel. The bands visible in the figure are those of the 25S and 18S ribosomal RNA.



## Fig 3.20 Annual Beta vulgaris Apical Meristem cDNA

An autoradiograph of a 1.5% (w/v) agarose gel dried under vacuum at 80°C after TCA precipitation of the radiolabelled polynucleotides and exposure to X-ray film for 24 h. The dried gel represents the single and double stranded cDNA synthesised from an aliquot of 95% of the RNA isolated from 35 dissected apical meristems of annual *Beta vulgaris* plants grown under warm short day conditions.

- Lane M,  $\lambda$  DNA, digested with *Hind* III and end labelled with [ $\alpha$ -<sup>32</sup>P] dCTP
- Lane 1, single stranded apical meristem cDNA
- Lane 2, double stranded apical meristem cDNA



unincorporated radiolabelled dCTP

# Fig 3.21 Size Fractionation of Apical Meristem cDNA through a Sephadex G-50 Column.

An autoradiograph of a 1.5% (w/v) agarose gel dried under vacuum at  $80^{\circ}$ C after TCA precipitation of the radiolabelled nucleotides and exposure to X-ray film for 24 h. 10 µl aliquots of cDNA from each of the 200 µl fractions passed through a Sephadex G-50 column which were shown to be radioactive (as measured using a Geiger-Müller counter) were separated through the agarose gel.

Lanes 1-9, 10 µl aliquots from 200 µl fractions 6-14 taken from the Sephadex G-50 column.

Lane M,  $\lambda$  DNA digested with *Hind* III and end labelled with [ $\alpha$ -<sup>32</sup>P] dCTP

column (lanes 1,2, and 3 of Fig. 3.21) contained cDNA, with no containable amincorporated radiolabel, and the least small stend cDNA. These terms transcore we therefore, pooled for ligation to the JB linkers.



column (lanes 1,2, and 3 of Fig. 3.21) contained cDNA with no contaminating unincorporated radiolabel, and the least small sized cDNA. These three fractions were, therefore, pooled for ligation to the JB linkers.

#### 3.2.4 Amplification of Apical Meristem cDNA After Ligation of Linkers

Since it was important not to lose any of the cDNA material pooled from the three fractions mentioned in the previous section, and since the preparation of the leaf library had demonstrated that the method employed in this study worked satisfactorily, the next few steps were monitored using a Geiger -Müller counter to establish the presence or absence of radiolabelled cDNA, thereby retaining as much cDNA as possible. Fractions 6,7 and 8 (lanes 1,2,and 3 of Fig 3.21) were subjected to ligation conditions in the presence of 5 µg of annealed JB1 and JB2. The material from this ligation reaction was passed down a Bio Gel A-150 column to remove the unincorporated radionucleotides and the unligated linkers which may interfere with the PCR process. Using a Geiger-Müller counter it was established that fractions 4-7 passed through the Bio-Gel column contained a peak of radioactivity consistent with the presence of cDNA free from contaminating radiolabel and small sized cDNA. A small aliquot (2.5 µl) was removed from each of the fractions 4-7 plus fractions 3 and 8, which may have contained a small amount of apical cDNA material useful for cloning. Fig 3.22 shows that each of the 6 fractions contained material amplifiable by PCR. Therefore the ligation was shown to be successful in this experiment. Fig 3.22 shows the cDNA amplified in all 6 fractions to be of the right size range, with an acceptable proportion of small cDNA species. Lane 1 of Fig 3.22 demonstrates that contamination of the PCR reagents with foreign DNA had not occurred since when no apical cDNA was present, no

#### Fig 3.22 PCR Amplification of Apical Meristem cDNA

Photograph of an ethidium bromide stained 1.5% (w/v) agarose gel of apical meristem cDNA. Fractions 6,7 and 8 from the Sephadex G-50 column from Fig 3.21 were subjected to ligation conditions and passed down a Bio-Gel A-150 column to remove unincorporated radionucleotides and the unligated linkers. 2.5  $\mu$ l aliquots from fractions 3-8 passed through the Bio-Gel A-150 column (described in section 3.2.4) were subjected to 35 cycles of PCR using the JB1 oligonucleotide primers and an aliquot of the products of these reactions was run on a 1.5% (w/v) agarose gel.

Lane M, 1kb "ladder" marker DNA (Gibco/BRL)
Lane 1, The product of 35 cycles of PCR in the absence of any apical cDNA
Lanes 2-7, The product of PCR amplification of 2.5 µl aliquots from fractions 3-8 of the Bio-Gel A-150 column

letectable amount of cDNAs was amplified. The peopled 98's proplifted artical cDNA fixen ractions-3-7 (lanes 2-6 of Fig 3.22) was digented with free 11



best recombination frequence, of approximately 20%. The library prepared from this factor was also shown to comain 1.4 x 10<sup>6</sup> recombinant clones.

3.2.6 Amplification of Inserts from the Apical Metisters (UNA Library

To show that the PCR emplified apical meristem cONA library prepared must 2019g of cDNA contained inserts of an acceptable size distribution it condon plaques (7 white and 1 blue) were amplified using PUC 1 and 2 as described in section 2.15. Fig 3.24 shows that the average size of these inserts was 600 bp with a range from less than 160 bp to 1.6 kbp. detectable amount of cDNA was amplified. The pooled PCR amplified apical cDNA from fractions 3-7 (lanes 2-6 of Fig 3.22) was digested with *Eco* RI.

#### 3.2.5 Removal of Small Sized cDNA Prior to Cloning

To eliminate the possibility of including a large proportion of small cDNAs in the apical meristem library, fractions 3-7 (lanes 2-6 of Fig 3.22) were subjected to a further size fractionation as described in section 3.1.5.2. Fig 3.23 shows the separation of the apical meristem cDNA by this method (described in section 3.1.5.2) to have successfully produced fractions of cDNA containing progressively smaller sized material. The *Eco* RI digested apical meristem cDNA shown in lane 2 of Fig 3.23 was cloned into the  $\lambda$  ZAP II vector. Using 50 and 250 ng of this digested cDNA ligations were performed in the presence of  $\lambda$  ZAP II. The ligated material was packaged, as previously explained in section 2.18.3, and serial dilutions of the packaged material were spread on to a lawn of X-L1 Blue *E. coli* cells in the presence of X-Gal and IPTG. Table 3.2 shows that 250 ng of apical cDNA gave the best recombination frequency, of approximately 80%. The library prepared from this material was also shown to contain 1.4 x 10<sup>6</sup> recombinant clones.

## 3.2.6 Amplification of Inserts from the Apical Meristem cDNA Library

To show that the PCR amplified apical meristem cDNA library prepared from 250 ng of cDNA contained inserts of an acceptable size distribution 8 random plaques (7 white and 1 blue) were amplified using PUC 1 and 2 as described in section 2.15. Fig 3.24 shows that the average size of these inserts was 600 bp with a range from less than 100 bp to 1.6 kbp.

## Fig 3.23 Removal of Small Sized Apical Meristem cDNA Prior to Cloning

Photograph of an ethidium bromide stained 1.5% (w/v) agarose gel of PCR amplified apical meristem cDNA from fractions 4-7 of the Bio-Gel A-150 column after being pooled and digested with *Eco* RI and then size fractioned as follows: The pooled digested apical meristem cDNA was subjected to electrophoresis and sectioned into four sequential 4mm strips at right angles to the direction of migration of the DNA. Aliquots of the cDNA recovered from these strips were run on a 1.5% (w/v) agarose gel

Lane M, 1 kb "ladder" marker DNA (Gibco/BRL)

Lanes 1-4, The DNA isolated from the first four strips of the 1.5% (w/v) agarose gel used to size fractionate the cDNA



## Table 3.2

Amount of DNA packaged	No. of clones obtained	Recombination frequency
250 ng apical meristem PCR amplified cDNA	1.4x10 <sup>6</sup>	80%
50 ng apical meristem PCR amplified cDNA	2.3x10 <sup>3</sup>	45%
100 ng test insert supplied by Strategene	3.4x10 <sup>7</sup>	92%

Table 3.2Properties of cDNA libraries in  $\lambda$  ZAP II prepared from differentamounts of PCR amplified apical meristem cDNA compared with a test insert control.

## Fig 3.24 Isolation of Random Inserts from the Apical Meristem cDNA Library.

Photograph of an ethidium bromide stained 1.5% (w/v) agarose gel of the inserts from 8 random apical meristem cDNA clones (1 blue and 7 white). 28 µl aliquots of SM phage buffer (described in section 2.4) containing 8 random plaques from the apical meristem cDNA library were subjected to 35 cycles of PCR using the PUC 1 and PUC 2 oligonucleotide primers. The DNA products of PCR amplification of these aliquots are shown here.

Lane M, 1 kb "ladder" marker DNA (Gibco/BRL)

- Lane 1, The DNA product of 35 rounds of PCR amplification using oligonucleotides PUC 1 and PUC 2 of a plaque showing a blue colour when plated out in the presence of X-Gal and IPTG
- Lanes 2-8, The DNA product of 35 rounds of PCR amplification using oligonucleotides PUC 1 and PUC 2 of random plaques showing a white colour when plated out in the presence of X-Gal and IPTG

Having successfully cloned PCR amplified cDNA prepared from Berg valgaris spical meristem RNA it was rext necessary to show that the cloned inserts were of show order



the PCR employed pical meristem RNA then n woeld upped A library contains a pup by the work of a ta pea leaf not aling process in flight by the leaf ected in the spical get from Fig 3.25 is 30 up then the

northern blot in Fig 3.25 shows there to be been selected between the level of LFICP transcripts in leaf and apical meristion MNA.

One thousand clones from the spical meristern cDNA library were proved with the LHCP cDNA from Zea mays. It was found that the clones hybridising to the LHCP cDMA probe were present at a lower abundance in the apical meristern library than in the built library. Fig 3.26 shows that of 1,000 clones screened only four of the clones from the apical library were giving a positive signal. These four clones were plaque profiles a screened a second time. Fig 3.27 shows three of the four (all but the bottom ket panel a trade)

Having successfully cloned PCR amplified cDNA prepared from *Beta vulgaris* apical meristem RNA it was next necessary to show that the cloned inserts were of plant origin.

#### 3.2.7 Screening the Apical Meristem cDNA Library with Known cDNA Probes

As with the leaf library, it was important to investigate whether the PCR amplified apical meristem cDNA library was representative of transcripts in the apical meristem RNA. If the amount of leaf RNA in lane 8 of Fig 3.25 is assumed to be 10  $\mu$ g then it would appear that the apical meristem RNA used to prepare the apical meristem cDNA library contains a lower proportion of LHCP transcripts than the leaf RNA. This is backed up by the work of Cuming and Bennett (1981) who state that the level of LHCP transcripts in pea leaf not exposed to light was 0.16%. Since LHCP is involved in the light harvesting process in photosynthesis, and since the apical meristem tissue is shaded from the light by the leaf primordia and mature leaves, fewer transcripts for LHCP would be expected in the apical tissue. However, if the fluorescence from the ethidium bromide stained gel from Fig 3.25 is to be believed and the amount of leaf RNA present is actually nearer to 30  $\mu$ g then the northern blot in Fig 3.25 shows there to be little difference between the level of LHCP transcripts in leaf and apical meristem RNA.

One thousand clones from the apical meristem cDNA library were probed with the LHCP cDNA from Zea mays. It was found that the clones hybridising to the LHCP cDNA probe were present at a lower abundance in the apical meristem library than in the leaf library. Fig 3.26 shows that of 1,000 clones screened only four of the clones from this sample of the apical library were giving a positive signal. These four clones were plaque purified and screened a second time. Fig 3.27 shows three of the four (all but the bottom left panel in Fig

Fig 3.25 Northern Blot of Warm, Short Day *Beta vulgaris* Annual andBiennial Apical RNA from Meristem Tissue compared with RNA Isolated from Annual Mature Leaf and Root Tissue Probed with the LHCP cDNA Probe of *Zea mays*.

a) Photograph of an ethidium bromide stained 1.5% (w/v) agarose gel containing 2.2 M formaldehyde. The gel contains differing amounts of annual and biennial apical meristem RNA alongside 10  $\mu$ g\* of annual leaf RNA and 30  $\mu$ g (see Fig 3.12) of annual root RNA. All

RNA was isolated from plant tissue grown under short day conditions at 20°C. The 25S and 18S rRNA are indicated.

Lanes 1-3, 2, 10 and 5 µg of biennial apical RNA respectively

Lane 4,  $30 \mu g$  of annual root RNA

Lanes 5-7, 10, 5 and 2 µg of annual apical RNA respectively

Lane 8,  $10 \mu g^*$  of annual leaf RNA

b) An autoradiograph showing the resultant hybridisation of the Zea mays LHCP cDNA probe to the RNA from the above figure after transfer to Hybond-N nylon membrane. Lanes 1-3, Transcripts from 2, 10 and 5  $\mu$ g of biennial RNA hybridising to the

radiolabelled LHCP cDNA clone of Zea mays

- Lane 4, Transcript from 30 µg of root RNA hybridising to the radiolabelled LHCP clone from Zea mays
- Lanes 5-7, Transcripts from 10, 5 and 2 µg of annual apical RNA hybridising to the radiolabelled LHCP clone from Zea mays
- Lane 8, Transcript from 10 µg\* of annual leaf RNA hybridising to the radiolabelled LHCP clone from Zea mays

\*The amount of leaf RNA in lane 8 was determined as 10  $\mu$ g according to the A<sub>260</sub> reading but the fluorescence from the ethidium bromide stained gel indicates the amount to be closer to 30 $\mu$ g.



## Fig 3.26 Screening the Apical Meristem Library with the LHCP Probe

Autoradiograph of 1,000 cDNA clones from the PCR amplified apical meristem cDNA library plated out and bound to Hybond-N nylon membrane. The nylon membrane filters were hybridised with the radiolabelled LHCP cDNA clone from Zea mays at 65°C for 24 h. The filters were washed three times in 2X SSC, 0.1% (w/v) SDS at 65°C for 20 min before being exposed to X-ray film for 24 h.



# Fig 3.27 Secondary Screening of Putative LHCP Clones from the Apical Meristem cDNA Library.

Autoradiograph of 4 separate putative LHCP clones from the apical meristem cDNA library plated out at a relatively low density and bound to separate Hybond-N nylon membrane filters. The nylon membrane filters were hybridised to the radiolabelled Zea mays LHCP cDNA probe at 65°C for 24 h. The filters were washed three times in 2X SSC, 0.1% (w/v) SDS at 65°C for 20 min before being exposed to X-ray film for 24 h.



Y-	
cDNA probe % of clones hybridising to the probe under	
stringent hybridisation conditions	
LHCP 0.3%	
$\alpha$ -tubulin deside the class probes Table 0.7%	
G PAL-2 <0.1%	
ALS <0.1%	
a determined that the corresponding tracersize used he sector is additional to	

3.27) to be true positive clones for LHCI Table 3.3 ybridistion conditions employed in this

ndance in apical RNA

 Table 3.3
 Frequency of clones in the PCR amplified apical meristem cDNA

 library hybridising to several DNA probes

Although the apical meristent oDNA library was found to contain clones which is hybridised to cDNA probes of plant origin, it was torrettent to establish this the cloned inserts within the apical meristem cDNA library were of breat subjects solved meristers origin. A small sample of apical meristem cDNA clones was, therefore, hybridised to single stranded cDNA prepared from apical meristem RNA as described in section 2.17. These clones from the apical meristem cDNA library were plated out at a low density (50-60 clones per plate) to enable an accurate assessment of the hybridisation frequency. Fig 3.28 shows hybridisation of these clones to the apical meristem prote. Approximately 75% of the clones screened from the apical meristem library with the apical meristem single stranded cDNA, hybridised to that probe ander releavely strangent washing conditions. Since only 80% of the

3.27) to be true positive clones for LHCP under the hybridisation conditions employed in this study.

Using the  $\alpha$ -tubulin, G-PAL 2 and ALS cDNA clones (described in section 2.19.2) as examples of clones encoding different sized mRNAs which are assumed to be expressed at different levels in the RNA of apical tissue from sugar beet, 1,000 apical meristem cDNA clones were probed with these three cDNA probes. Table 3 shows the results of these probing experiments and demonstrates that after a second round of screening with all three probes levels of representitiveness ranged from less than 0.1% to 0.7% of clones in the library. These data suggest that the corresponding transcripts would be present at relativelylow abundance in apical RNA.

# 3.2.8 Screening the Apical Meristem cDNA Library with Single Stranded Apical Meristem cDNA

Although the apical meristem cDNA library was found to contain clones which hybridised to cDNA probes of plant origin, it was important to establish that the cloned inserts within the apical meristem cDNA library were of *Beta vulgaris* apical meristem origin. A small sample of apical meristem cDNA clones was, therefore, hybridised to single stranded cDNA prepared from apical meristem RNA as described in section 2.17. These clones from the apical meristem cDNA library were plated out at a low density (50-60 clones per plate) to enable an accurate assessment of the hybridisation frequency. Fig 3.28 shows hybridisation of these clones to the apical meristem probe. Approximately 75% of the clones screened from the apical meristem library with the apical meristem single stranded cDNA, hybridised to that probe under relatively stringent washing conditions. Since only 80% of the

# Fig 3.28 Screening of the Apical Meristem cDNA Library with Single Stranded Apical Meristem cDNA

Autoradiograph of two Hybond-N nylon membrane filters of plaque lifts from two separate plates, each containing 50-60 plaques, after hybridisation to radiolabelled single stranded apical meristem cDNA. The filters were hybridised with the single stranded cDNA at  $65^{\circ}$ C for 24 h and then washed three times at  $65^{\circ}$ C in 2X SSC, 0.1% (w/v) SDS for 20 min before being exposed to X-ray film for 48 h.



128

clones in the library were recombinant (Table 3.2) this seems a high frequency of complementary sequences. Furthermore, it is important to note that a proportion of the clones could have been represented at too low a frequency in the apical meristem cDNA probe for them to show a signal under these conditions (so called 'cold' plaques, Scott *et al.*, 1991).Thus 75% was considered an acceptable figure indicating the presence of a high frequency of inserts derived from apical meristem RNA within the apical meristem cDNA library.

## 3.3. Methods of Isolating Apex-Specific cDNA clones from the Apical Meristem cDNA Library.

The annual apical meristem cDNA library has been shown to contain inserts of varying size range and to represent certain specific cDNA clones of plant origin at an acceptable level. It has also been shown to contain a substantial amount of inserts complementary to single stranded cDNA prepared from the tissue used to produce the library. Therefore, it was assumed that the library derived from this apical meristem material was suitable for screening to isolate apex-specific cDNA clones.

# 3.3.1 Differential screening

The principles of differential screening were explained in section 1.10. In this study the positive probe was the PCR amplified apical meristem cDNA used to prepare the apical meristem cDNA library. The negative probe was the PCR amplified cDNA used in the preparation of the leaf cDNA library. Both of these were available in large amounts and so many thousands of clones from the apical meristem cDNA library could be screened for the presence of apex-specific cDNA clones.

Initially 100,000 clones of the apical meristem cDNA library were plated out and duplicate plaque lifts were taken from each plate as described in section 2.14.1. Fig 3.29 shows the result of a typical differential screen from this study. The arrows indicate putative apex-specific clones. Since only 38 putative apex-specific clones were isolated from this initial study a further 100,000 clones were differentially screened in the same way. A total of 64 putative apex-specific clones were chosen for further study from the 200,000 clones screened. This was a disappointingly low number of clones isolated from such a large number of clones screened.

## 3.3.2 Secondary Screening of Putative Apex-Specific cDNA Clones

Of the 64 clones subjected to a secondary screen, by the "giant plaque" method described in section 2.20.1, only 8 of them appeared to be hybridising more strongly to the apical cDNA probe than the leaf cDNA probe (Fig 3.30). None of these putative positive clones were hybridising specifically to the apical cDNA probe. This, again was a disappointing result since what was initially only a very few promising clones suddenly diminished to none that were apex-specific, and only 8 that could possibly be enhanced in the apical meristem cDNA probe.

# Fig 3.29 Differential Screening of the PCR Amplified Apical Meristem cDNA Library

Autoradiograph of duplicate Hybond-N nylon membrane filters prepared from a typical sample of apical meristem cDNA clones plated out at a density of approximately 500 plaques per plate. One filter from each plate was hybridised to annual leaf PCR amplified cDNA radiolabelled to a specific activity of  $10^8$  cpm per µg of DNA and the other filter from the same plate was hybridised to annual apical PCR amplified cDNA radiolabelled to a specific activity of  $10^8$  cpm per µg of DNA radiolabelled to a specific activity of  $10^8$  cpm per µg of DNA radiolabelled to a specific activity of  $10^8$  cpm per µg of DNA. All filters were hybridised at  $65^{\circ}$ C for 24 h before being washed three times in 2X SSC, 0.1%(w/v) SDS at  $65^{\circ}$ C and exposed to X-ray film for 24 h. The arrows in b) indicate clones that were putatively apex-specific.

a) LEAF



b) APEX

## Fig 3.30 Secondary Screening of Putative Apex-Specific cDNA Clones.

Autoradiograph of duplicate Hybond-N nylon filters containing 64 putative apexspecific "giant plaques". The plaques were prepared by spotting an aliquot of SM phage buffer, (described in section 2.4) containing an agar plug within which was hopefully a single bacteriophage plaque, onto a lawn of *E.coli* cells and incubating at  $37^{\circ}$ C for 24 h. One filter was hybridised to the radiolabelled PCR amplified annual leaf cDNA whilst the other filter was hybridised to the radiolabelled PCR amplified apical cDNA. Both hybridisations were performed at  $65^{\circ}$ C and in each case the probes used were labelled to a specific activity of  $10^{8}$  cpm per µg of DNA. The filters were then washed three times at  $65^{\circ}$ C in 2X SSC, 0.1% (w/v) SDS for 20 min before being exposed to X-ray film for 24 h.

The arrow indicates the most likely apex-specific cDNA clone (number 26)

a) LEAF b) APEX



The previous reachs were neared to be the second as a size were and a to be 201000 clones screened none were seen werein. If the here where were reached here spectements and expression after recordary screening, the mass of the term were reached by a mind out to be expressed most strongly in the leaf means if were down the marriers as use the cold plaque method of Scon *et al.* (1991) to see if approximation there are the solution of clones (200) then the method of Scon *et al.* (1991) to see if approximation members of clones (200) there the method of Scon *et al.* (1991) to see if approximation members of clones (200) there the method of Scon *et al.* (1991) to see if approximation members of clones (200) there this implified spical members and the rest of the leaf of a place members of clones (200) there this library, which did not hybridise to either the leaf of a place members of clones (200) there this screened by the "giant planue" method described in section 2.20 1. Fig. 3.12 shows that es day

## 3.3.3 Investigation of Putative Positive Clones to Establish the Abundance of their Transcripts Within Leaf and Apical Meristem RNA

The most promising of the 8 clones still giving a differential signal when hybridised to leaf and apical cDNA (clone 26) was used to probe a northern blot of RNA of *Beta vulgaris* tissue isolated from annual and biennial apical meristems and mature leaves. Fig 3.31 shows that clone 26 hybridised most strongly to mature leaf RNA (apparently to rRNA) and gave an almost equal signal in the biennial and annual apex RNA. Although there is evidence of degradation of the RNA in this blot (Fig 3.31) the hybridisation of the leaf RNA indicates that the cDNA is not apex-specific. Hybridisation of similar northern blots to inserts of several other putative positive clones gave no indication of apex-specificity (data not shown). These results provide further evidence indicating that the small number of putative apex-specific clones initially isolated did not contain any true apex-specific clones.

## 3.3.4 'Cold' Plaque Screening of the Apical Meristem Library.

The previous results were not very promising, as they indicated that of the 200,000 clones screened none were apex specific. Of the few which were putatively apex-enhanced in expression after secondary screening, the most promising clone (clone 26) turned out to be expressed most strongly in the leaf tissue. It was decided, therefore, to use the cold plaque method of Scott *et al.* (1991) to see if apex-specific clones could be obtained from the PCR amplified apical meristem cDNA library. A significant number of clones (200) from this library, which did not hybridise to either the leaf or apical meristem cDNA probes, were screened by the "giant plaque" method described in section 2.20.1. Fig 3.32 shows that of the

# Fig 3.31 Northern Blot of Annual and Biennial Apical Meristem RNA and Annual Leaf RNA Probed with Clone 26

a) Photograph of an ethidium bromide stained 1.5% (w/v) agarose gel containing 2.2 M formaldehyde. The gel contains 10 µg of annual and biennial apical meristem RNA alongside 5µg of annual leaf RNA.

Lane 1, 10 µg of biennial apical meristem RNA

Lane 2,  $10 \,\mu g$  of annual apical meristem RNA

Lane 3,  $5 \mu g$  of annual leaf RNA

b) Autoradiograph showing hybridisation of the cDNA insert of clone 26 to the RNA from the formaldehyde gel in a) after blotting onto a Hybond-N nylon membrane filter. The membrane was hybridised to the radiolabelled clone 26 cDNA probe at 65°C and then washed at 65°C in 2X SSC, 0.1% (w/v) SDS; 0.2X SSC, 0.1% (w/v) SDS; and 0.1X SSC, 0.1% (w/v) SDS, each wash was for a duration of 20 min. The filter was exposed to X-ray film for 48 h.

Lane 4,	Transcripts from a 10 $\mu$ g sample of biennial apical meristem RNA hybridising
	to the radiolabelled clone 26 cDNA probe
Lane 5,	Transcripts from a 10 $\mu$ g sample of annual apical meristem RNA hybridising
	to the radiolabelled clone 26 cDNA probe
Lane 6,	Transcript from a 5 $\mu$ g sample of annual leaf RNA hybridising to the
	radiolabelled clone 26 cDNA probe



a)
## Fig 3.32 'Cold' Plaque Screening of the Apical Meristem cDNA Library

Autoradiograph of duplicate Hybond-N nylon membrane filters of over 200 plaques which were white in the presence of X-Gal and IPTG but which were seen not to hybridise to either the apical meristem or the leaf cDNA probes. Aliquots of the SM phage buffer (described in section 2.4) from over 200 of these 'cold' plaques were spotted onto a lawn of *E.coli* cells and incubated for 24 h. Duplicate plaque lifts were made of each of four plates. One copy of the plaque lifts from each plate was hybridised to the PCR amplified annual *Beta vulgaris* leaf cDNA probe whilst the other copy from each plate was hybridised to the PCR amplified annual apical meristem cDNA probe. Both cDNA probes were radiolabelled to a specific activity of  $10^8$  cpm per µg of DNA and were hybridised to the nylon membrane filters at  $65^{\circ}$ C for 24 h. The filters were washed three times at  $65^{\circ}$ C in 2X SSC, 0.1% (w/v) SDS for 20 min before being exposed to X-ray film for 48 h.



described above cDNA spinies a sequences would The subtractive with 100 fold le with biotin, or h

bound to the say







APEX



parries for as hybridized 200 'cold' plaques isolated none hybridised more strongly to the apical meristem cDNA probe. Hence, there was no evidence that the 'cold' plaque method would reveal any significant number of apex-enhanced (let alone apex-specific) cDNA clones, and so this method was therefore, not pursued.

TORGANDER DELADERS

3.3.5 Development of Methods for Subtractive Screening of the Apical Meristem <u>cDNA Library.</u> (based on the method of Sive and St John, 1988)

Because of the lack of encouraging results obtained by differential screening, described above, it was decided that a more positive set of results could be obtained using a subtractive approach (see section 1.10 and 2. 20.3). The hope was that in removing as many cDNA species as possible common to both the apical and leaf cDNAs, apex-specific cDNA would be more abundant. This cDNA enriched for apex-specific, or at least apex-enhanced, sequences would then facilitate the isolation of corresponding clones from the apical library. The subtractive procedure employed in this study involved the incorporation of biotin labelled dATP into PCR amplified leaf cDNA (driver cDNA). This leaf cDNA was incubated with 100 fold less PCR amplified apical meristem cDNA before removal of cDNA labelled with biotin, or bound to a biotin labelled cDNA fragment, using streptavidin.

It is important when attempting this procedure to ensure that the negative (or driver) DNA is sufficiently biotin labelled to enable the subtraction of common cDNA species when bound to the streptavidin protein. This protein body has four high affinity binding sites for biotin and, therefore, any cDNA that is labelled with the biotin moiety, or that has hybridised to cDNA labelled with biotin, should be removed by this step.

### 3.3.5.1 Incorporation of Biotin Labelled dATP into PCR Amplified Leaf cDNA

Two methods of biotin labelling were available to use in this study;

1) nick translation of the driver cDNA incorporating biotin labelled dATP into the reaction mixture;

2) photolabelling of the driver cDNA with photoactive biotin using a high intensity ultra-violet light source.

Both procedures were equally efficient at biotin labelling the driver cDNA, but the nick translation process was cheaper, safer to the user and any fellow workers, and allowed the incorporation of radiolabelled nucleotides into the cDNA. This last point is important because it enables the user to monitor the presence of the driver cDNA whilst establishing the necessary conditions for subtraction. The leaf cDNA nick translated in the presence of both radiolabelled dCTP and biotin labelled dATP was passed through a Sephadex G-50 column to eliminate unincorporated biotin labelled dATP (which would hinder the removal of the driver cDNA by competing for the binding sites of the streptavidin). 10 µl aliquots were removed from each of the 200 µl fractions passed through the column containing traces of radiolabel (as measured with a Geiger-Müller counter) and run on a 1.5 % (w/v) agarose gel. The gel was Southern blotted and first exposed to X-ray film before being stained (using the blue-GENE kit supplied by Gibco-BRL) to test for the presence of biotin. Fig 3.33 b) shows that the nick translation procedure efficiently incorporated radiolabelled dCTP into fractions 4-8 from the G-50 column. Very little unincorporated dCTP was present in any of the fractions from this column (as measured using a Geiger-Müller counter) indicating that the incorporation of biotin labelled dATP was also likely to be efficient. Fig 3.33 a) shows that those fractions which appeared to contain radiolabelled cDNA also contained biotin-

## Fig 3.33 Incorporation of Biotin Label in to the PCR Amplified Leaf cDNA

A blot of 10  $\mu$ l aliquots from 200  $\mu$ l fractions of PCR amplified leaf cDNA passed through a Sephadex G-50 column. The cDNA was labelled with both biotin-7-dATP and [ $\alpha$ -<sup>32</sup>P] dCTP by nick translation. The leaf cDNA was separated on a 1.5% (w/v) agarose gel and transferred to a Hybond-N nylon membrane filter. The filter was stained for the presence of biotin using the blue-GENE non-radioactive detection system before being exposed to Xray film for 24 h.

a) Shows the incorporation of biotin into the leaf cDNA after staining with the blue-GENE kit from Gibco/BRL.

Lanes 1-6, Biotin-7-dATP labelled PCR amplified leaf cDNA from fractions 3-8 of the Sephadex G-50 column.

b) Shows the incorporation of  $[\alpha - {}^{32}P]$  dCTP into the leaf cDNA after exposure of the blot to X-ray film for 24 hours

Lanes 1-6, Radiolabelled PCR amplified leaf cDNA from fractions 3-8 of the Sephadex G-50 column labelled cDNA. Therefore, it was decided that fractions 3-8 from the Q-30 column (lanes 1-0 in Figs 3.33a and b) should be used to establish the efficiency of removal of each DNA by complavidin. a) 1 - 2 - 3 - 4 - 5 - 4

2.3.5.2 Removal of

Since the dri to show that the size biotin-labelled, ici-<sup>2</sup> conditions described subjected to a phorne is a protein, and serstreptavidin would in

b) optavion. The season of run on a 1.5 % (w) equivalent amount was Southern blot indicates that no d labelled driver cD the conditions use



1 2 3 4 5 6



any cDNA beametres use oution was removed and III marker DNA and as this experiment. This ad sourc to X-ray film and fore, or sumably on highle in. This decompliates that filment. labelled cDNA. Therefore, it was decided that fractions 3-8 from the G-50 column (lanes 1-6 in Figs 3.33a and b) should be used to establish the efficiency of removal of such DNA by streptavidin.

#### 3.3.5.2 Removal of Biotin Labelled Leaf cDNA Using Streptavidin

Since the driver cDNA was trace labelled with radioactive dCTP it was not difficult to show that the streptavidin was able to remove the driver cDNA very efficiently. The biotin-labelled,  $[\alpha^{-32}P]$  labelled cDNA was incubated with the streptavidin under the conditions described in section 2.20.3. After this incubation the reaction mixture was subjected to a phenol/chloroform extraction as described in section 2.8.3. Since streptavidin is a protein, and since phenol is able to remove protein from an aqueous solution, the streptavidin would have been taken up by the phenol, along with any cDNA bound to the streptavidin. The aqueous phase from this phenol/chloroform extraction was removed and run on a 1.5 % (w/v) agarose gel along with radiolabelled  $\lambda/Hind$  III marker DNA and an equivalent amount of biotin-labelled driver cDNA as was used in this experiment. This gel was Southern blotted. Fig 3.34 shows the resultant blot after exposure to X-ray film and indicates that no detectable radiolabelled driver cDNA (and therefore, presumably no biotin labelled driver cDNA) remained after binding with the streptavidin. This demonstrates that the conditions used to remove biotin-labelled cDNA were very efficient.

# Fig 3.34 Removal of Biotin Labelled Leaf cDNA using Streptavidin

Autoradiograph of a blot of biotin labelled,  $[\alpha - {}^{32} P]$  dCTP labelled cDNA remaining after binding to steptavidin compared with an equivalent amount of biotin labelled,  $[\alpha - {}^{32} P]$  dCTP labelled leaf cDNA not incubated in the presence of streptavidin.

Leaf cDNA internally labelled by nick translation using  $[\alpha - {}^{32}P]$  dCTP and biotin-7dATP was incubated in the presence (lane 1) or absence (lane 2) of streptavidin in 0.5 M NaCl at room temperature for 20 min. After phenol/chloroform extraction the aqueous phase was removed and run on a 1.5% (w/v) agarose gel. The size markers (lane M) were  $\lambda$  DNA digested with Hind III and end labelled with  $[\alpha - {}^{32}P]$  dCTP



140

## 3.3.5.3 Establishing the Number of Rounds of Subtraction Required to Remove the Driver cDNA

Although the data in Fig 3.34 show there was no visible radiolabelled cDNA after subtraction, there only needs to be a very small amount of DNA available for PCR to amplify that material many thousand fold. Therefore, to test the efficiency of the subtraction procedure the PCR amplified biotin-labelled leaf driver cDNA was subjected to four rounds of mock-subtraction (i.e. subtraction in the absence of any apical cDNA). After each round of mock-subtraction a small aliquot (10 µl) was removed from the aqueous phase left after phenol/ chloroform extraction, and subjected to PCR as described in section 2.15. The products of amplification were barely visible on an agarose gel and so the agarose gel was blotted and probed with radiolabelled PCR amplified leaf cDNA. Fig 3.35 shows the result of this blot and demonstrates that amplification of samples from the first three rounds of mock subtraction produced enough cDNA to be detected with a radiolabelled leaf cDNA probe. After four rounds of mock-subtraction PCR amplification of an aliquot of the supernatant remaining after phenol /chloroform extraction did not give rise to a detectable amount of amplifiable cDNA. Therefore, in all subsequent experiments at least four rounds of subtraction were performed, in the hope that this would remove enough driver cDNA to allow the isolation of apex-specific cDNA.

#### 3.3.5.4 Subtraction of Leaf Driver cDNA with Apical Meristem cDNA

Since the mock-subtraction procedure was shown to be successful in the previous experiments subtraction with apical meristem cDNA present was performed. Using 100 fold

# Fig 3.35 Establishing the Number of Rounds of Subtraction Required for Efficient Removal of the Driver cDNA

Autoradiograph of a Southern blot of cDNA prepared by PCR amplification of 10  $\mu$ l aliquots taken from the aqueous phase remaining after each round of mock-subtraction (using leaf cDNA in the absence of any apical meristem cDNA). Samples taken from up to four rounds of mock-subtraction were amplified by 35 cycles of PCR and the products were separated on a 1.5% (w/v) agarose gel and transferred to Hybond-N nylon membrane. The nylon membrane was hybridised with [ $\alpha$ -<sup>32</sup>P] dCTP labelled PCR amplified leaf cDNA at 65°C for 24 h. The nylon membrane was washed three times in 2X SSC, 0.1% (w/v) SDS for 20 min before exposure to X-ray film for 24 h.

Lane 1-4, Hybridisation of the radiolabelled leaf cDNA probe to DNA prepared by PCR amplification of aliquots of cDNA taken from the 1st to 4th rounds of mock-subtraction respectively.

Lane 5, Hybridisation of the radiolabelled leaf cDNA probe to 25 ng of PCR amplified leaf cDNA as a control



## Fig 3.36 Subtraction of Leaf Driver cDNA with Apical Meristem cDNA

pelana to a subset

Autoradiograph of a Southern blot of PCR amplified cDNA prepared by subtraction of apical meristem cDNA with 100 fold excess of driver PCR amplified leaf cDNA. The apical meristem cDNA was subtracted against PCR amplified leaf cDNA for five successive rounds. An aliquot of the cDNA remaining after each round of subtraction was subjected to 35 rounds of PCR amplification. A sample of each of these amplified products was run on a 1.5% (w/v) agarose gel and blotted onto Hybond-N nylon membrane. The nylon membrane was hybridised at 65°C with [ $\alpha$ -<sup>32</sup>P] dCTP labelled PCR amplified leaf cDNA for 24 h and washed three times in 2X SSC, 0.1% (w/v) SDS for 20 min before being exposed to X-ray film for 48 h. The position of  $\lambda$ /*Hind* III DNA markers (taken from the ethidium bromide stained gel prior to blotting and probing) are indicated.

- Lane 1, resultant hybridisation of the radiolabelled leaf cDNA probe to 100 ng of PCR amplified apical meristem cDNA
- Lane 2, resultant hybridisation of the radiolabelled leaf cDNA probe to 100 ng of PCR amplified leaf cDNA
- Lane 3-7, resultant hybridisation of the radiolabelled leaf cDNA probe to the cDNA synthesised after 35 rounds of PCR amplification from aliquots of cDNA remaining after 1-5 rounds of subtraction with 100 fold excess of driver leaf cDNA.





143

more driver cDNA than apical cDNA five rounds of subtraction were performed as described in section 2.20.3. An aliquot of the aqueous phase from each of these rounds of subtraction was amplified and the product subjected to electrophoresis through a 1.5% (w/v) agarose gel before Southern blotting and probing with radiolabelled PCR amplified leaf cDNA. Fig 3.36 shows that in every case the aliquot of aqueous phase removed from the subtraction mix contained enough cDNA to allow amplification of that cDNA. Also the product from each round of subtraction hybridised to the driver leaf cDNA, when used to probe the Southern blot of these products ( as shown in Fig 3.36). This experiment was repeated and the same result was obtained. This indicates that, under the conditions employed, it was not possible to obtain a subtracted fraction containing purely apex-specific sequences. Any sequences that did remain after extensive subtraction still showed hybridisation to leaf cDNA. It is likely that 35 rounds of PCR was too stringent for the detection of apex-specific cDNA and fewer cycles should have been performed. It is possible then that the cDNA produced from the aqueous phase of the subtracted cDNA fractions after fewer cycles of PCR would have been enhanced for apex-specific cDNA, and as such would have proved more useful in the isolation of such cDNA clones.

study of meristematic development since very little is known of the molecular mechanisms involved in the transition to flowering. The introduction of molecular biological techniques has enabled scientists to look at differences in the generates of various organisms, and this has led to the isolation of many important genes.

The results from the previous section show that from small amounts of sugar beet total RNA representative cDNA libratics can be prepared using the presenced shows in Fig. 3.4. This protocol provides a simple and rapid proteduce for the construction of such CHAPTER 4

Froduced from very small among the first state is significant. This contacts in the First

DISCUSSION

Fogure targe another of poly (A) and (I an

### 4.1 Development of the PCR Cloning Technique

The ultimate aim of this study was to establish the nature of the agent responsible for floral induction upon exposure to low temperature treatment in sugar beet (*Beta vulgaris*). Since it is known that the annual sugar beet plant possesses a dominant gene (B) conferring annuality upon the beet plants it was assumed that the agent responsible for induction of flowering upon exposure to low temperatures may be the B gene, or some product thereof. As vernalization is perceived by the shoot apical meristem it seemed reasonable to investigate the genes of the annual sugar beet shoot apical meristem and look for apex-specific cDNA clones. Any cDNA clones isolated would be further characterised to see if any may be involved in the low temperature induction of flowering, and could possibly represent the B gene. The isolation of apex-specific cDNAs would prove a useful tool in the study of meristematic development since very little is known of the molecular mechanisms involved in the transition to flowering. The introduction of molecular biological techniques has enabled scientists to look at differences in the genomes of various organisms, and this has led to the isolation of many important genes.

The results from the previous section show that from small amounts of sugar beet total RNA representative cDNA libraries can be prepared using the protocol shown in Fig 3.4. This protocol provides a simple and rapid procedure for the construction of such libraries. The fact that good quality RNA, suitable for efficient cDNA synthesis, can be produced from very small amounts of plant tissue is significant. This contrasts with the PCR cloning protocols designed for the isolation of genes expressed in mammalian systems which require large amounts of poly  $(A^+)RNA$  (Timblin *et al.*, 1990). Furthermore, this procedure is straightforward when compared with other systems in that precipitation steps are kept to a minimum, thus no carriers are required. Complex cloning strategies are also minimised (Belyavsky *et al.*, 1989). The work of Domec *et al.* (1990) demonstrates that it is possible to clone cDNA synthesised from total RNA by PCR but more complex amplification procedures are involved since the linkers used to flank each end of the double stranded cDNA are different from each other; therefore, two PCR primers are required for the amplification of cDNA using this protocol.

The cDNA prepared from the sugar beet tissue was smaller than would have been expected, and several attempts to increase the size of the cDNA were unsuccessful. Therefore, it was decided to use this cDNA for cloning. Since PCR amplification is limited to a size of 3 to 4 kbp (Erlich *et al.*, 1991) any gain in cDNA size may have been lost subsequently due to the PCR amplification of that cDNA. If the full sequence of any gene was desired a clone could have been isolated from a sugar beet genomic library probed with the cDNA clone of interest. The small size of the starting cDNA is represented in the size of the inserts isolated from random recombinant cDNA clones from both the apical meristem and leaf cDNA libraries. For the reasons stated above, however, this was not seen as a problem.

On screening the leaf library with the LHCP cDNA clone (Fig 3.13) it was observed after secondary screening that the number of clones hybridising to the probe was of the expected order of magnitude. The work of Cuming and Bennett (1981) on the expression of LHCP in *Pisum sativum* demonstrated that LHCP mRNA is present at an abundance of 2% of the poly (A<sup>+</sup>) RNA in photosynthetically active tissue and 0.16% in etiolated leaf tissue. Since the apical meristem of sugar beet is shaded from the light by the surrounding leaves this tissue may be assumed to be less photosynthetically active than the leaf tissue, and as such resemble etiolated leaves. The data in the previous chapter (section 3.1.6.2) on this subject show that after secondary screening LHCP clones were represented at a level of 3% in the leaf cDNA library and 0.3% in the apical cDNA library (section 3.2.7). This is within the expected order of magnitude and is considered to be a good indication that the two libraries were representative of the RNA transcripts used to make the double stranded cDNA, and as such representative of cDNA libraries prepared by conventional techniques (Huynh *et al.*, 1985).

Since both libraries were prepared from total RNA, which removes one very wasteful step from the cloning strategy, a ribosomal DNA clone was used to demonstrate the lack of an excessively high number of cDNA clones originating from ribosomal RNA (section 3.1.6.2; section 3.2.7). For both libraries it was observed that eliminating the step of enriching for poly (A<sup>+</sup>)RNA had not been to the detriment of the cDNA libraries. Using this evidence, and further data from screening the apical meristem cDNA library with α-tubulin, G-PAL 2, and ALS cDNA probes, it was assumed that the leaf and apical meristem cDNA libraries prepared by this PCR cloning technique were representative of the initial RNA transcripts used in their construction. They were, therefore, considered to be significantly different from each other to indicate that the cDNA amplified from the leaf and apical meristem cDNA species were not contaminated with cDNA species from each other.

## 4.2 Techniques for the Isolation of Apex-Specific cDNA Clones

Using PCR amplified cDNA for the differential screening of the apical meristem cDNA library led to the isolation of several putative positive apex-specific cDNA clones. This process was very labour intensive and eventually, from differentially screening 200,000 apical meristem cDNA clones only 64 appeared to give an indication that they may be more highly expressed in the apical meristem cDNA probe. On secondary screening of these 64 putative positive clones 8 still appeared to be more highly represented in the plus (apical meristem cDNA) probe than in the minus (leaf cDNA) probe. However, no single cDNA clone stood out as a true apex-specific clone, being represented only in the plus cDNA probe.

Although the cloning procedure used here has overcome the need for large quantities of RNA as the starting material for cloning, there is still a need for large amounts of RNA to be available for the characterisation of interesting cDNA clones. It may be the case that the PCR amplification procedure misrepresents the cDNA species in a population enough to disrupt the analysis of cDNA libraries prepared from that material. If a low abundance cDNA species is represented at twice its usual level in the plus probe this may lead to that cDNA species being represented at the same level in the plus and minus differential probes. For this reason PCR is not recommended as the best method for preparing cDNA probes that are to be used in differential screening. The data in Chapter 3 indicate that misrepresentation should not be a problem for this study, but the possibility must not be ruled out.

Subsequent characterisation of clones of interest was hindered by a lack of annual, warm, short day apical RNA available from sugar beet plants of the desired age (8-10 weeks). For enough RNA for one northern blot at least 100 apices must be isolated and the

RNA successfully retrieved from them. This compares with the 35 apices isolated for the cloning procedure. When using the more abundant leaf RNA for northern analysis of cDNA clones of interest there is a good chance of success since large quantities of RNA (mg amounts) may be isolated at any one time from a small number of leaves. One of the major stumbling blocks in this study was the apparent degradation of apical meristem RNA used in the analysis of the 8 putative positive cDNA clones (data not shown). This is most likely to be caused by the action of a ribonuclease so RNA must be prepared under conditions where the ribonuclease activity is eliminated. Equipment used for the isolation and analysis of RNA was therefore sterilised, gloves were worn and DEPC was used to inhibit the enzymes action. Several attempts were made to isolate non-degraded apical meristem RNA. On only two occasions was apical RNA that appeared to be non-degraded successfully isolated (Fig 3.25 and 3.31). However, on probing the northern blot in Fig 3.31 with the radiolabelled clone 26 cDNA probe the apparently intact RNA gave what appeared to be a non-specific signal. This could not have been due to the radiolabelled probe hybridising to a multigene family since there appeared to be no distinct bands of RNA transcripts hybridising to the probe but a smeared signal. However, it is more likely that a degree of degradation had occurred in the transfer of the RNA to the nylon membrane. Therefore, in this study it was not possible to show that all of the putative positive clones were false positives but one which appeared most likely to be an apex-specific clone was seen to be expressed more highly in the leaf RNA. Although several repeat northern blots were performed they all gave the same smeared result and so were not included in the results section. To repeat this northern blotting procedure with apical meristem RNA would require growing annual sugar beet plants under warm, short day conditions for 8-10 weeks. A maximum of 150 plants may be grown up at any one time, and usually 1/3 of them fall foul to pest infestation or waterlogging. From the

remaining healthy plants 80-90% of the apical meristems were usually successfully isolated. Since this operation is very tedious and painstaking, not surprisingly some of the apices were lost. From these apices it was possible to isolate enough RNA for 1-2 northern blots. If this apical meristem RNA was degraded by the ribonuclease enzyme, or was not precipitated in the presence of 2 M LiCl then a further 8 weeks must pass for the experiment to be repeated. Therefore, the conclusion here is that the difficulty of the experiments has meant that characterising a large number of putative positive apex-specific cDNA clones has not been possible. The limited amount of data obtained during this study on the characterisation of putative positive cDNA clones did, however, indicate very strongly that none of the putative positive clones isolated from the apical meristem library was enhanced in the apical meristem and, as shown by secondary differential screening, none was apex-specific.

As previously mentioned the two cDNA libraries appear to represent different populations of cDNA species yet no apex-specific cDNA clones were isolated. This may be because the cDNA used for differential screening was prepared by PCR amplification after the initial amplification of the cDNA used to make the libraries. PCR is a very sensitive procedure so a small amount of contaminating cDNA would be amplified a million fold, leading to results which may be misrepresentative of the original population. To try to eliminate contamination as a problem for this study separate equipment was used when handling PCR amplified cDNA material, solutions used in the PCR reaction were aliquotted into small amounts so that carryover from previously used solutions did not occur, and all reactions were carried out using sterile equipment. Therefore, contamination should have been eliminated but cannot be ruled out.

The susceptibility of the PCR reaction to contamination may explain to a degree the reason for the subtraction of apex-specific cDNA species with the leaf driver cDNA being

unsuccessful. In the absence of any apical meristem cDNA 4 rounds of mock-subtraction were seen to reduce the PCR amplified cDNA present in the final aqueous phase to an amount undetectable via Southern blotting and probing with a radiolabelled leaf cDNA probe. Therefore, the subtraction procedure was, in essence, working satisfactorily. However, upon addition of apical cDNA at a concentration 100 X lower than the driver DNA, even after 5 rounds of subtraction, sufficient cDNA remained to hybridise to a leaf cDNA probe (Fig 3.36) after PCR amplification. There are several reasons why this may have occurred. First of all, contaminating cDNA from either the leaf or apical meristem cDNA populations may have entered into the subtraction system at any point in the procedure. If this was the case several rounds of subtraction should have removed the contaminating cDNA species. Secondly, the hybridisation conditions used during subtraction may have been too stringent for common cDNA species to hybridise. This should not have been the case since the hybridisation conditions used during subtraction were the same as those used in differential screening of the apical meristem cDNA library. Since both the apical meristem and leaf cDNA species hybridised to the apical meristem cDNA clones studied under these conditions, it would be expected that any common cDNA species from the apical and leaf cDNA probes in the subtraction system would have hybridised to each other, if not after the first round of subtraction then after the second or third. The use of excess driver leaf cDNA in the subtraction procedure could have been a problem in that a majority of common cDNA species would not have bound to their complement in the apical cDNA population. Since after each round of subtraction the remaining cDNA is amplified, and since some of the leaf cDNA species should have been removed from the subtraction set-up, the number of apexspecific cDNA species within the subtraction should gradually increase and out number the common cDNA species in the subtracted cDNA population. The third possibility would be

that the streptavidin protein was not binding to the driver cDNA. This was demonstrated not to be the case in Fig 3.34 and 3.35. Everything, therefore, points to contamination of the PCR amplifiable subtracted cDNA samples with either leaf or apical cDNA species. This material could have been introduced into the subtraction procedure after binding of the driver cDNA to the streptavidin and, as such, would not have been exposed to any form of subtraction. If this contamination was repeated at the same point after each round of subtraction then a subtracted apex-specific cDNA probe would never have been produced by this method.

Considering all of these drawbacks with the characterisation of cDNA clones isolated from the apical meristem cDNA library it cannot be said that the apical meristem cDNA library did not contain clones specific to the apical meristem but, using the methods employed in this study, it was not possible to isolate cDNA clones specific to the apical meristem. Using similar PCR protocols Meeks-Wagner et al. (1989), and Medford et al. (1991) have utilised plant systems that give rise to large quantities of shoot meristem material to isolate genes expressed in the shoot tips of tobacco and cauliflower respectively. For the study of tobacco a thin cell layer system developed by Tran Thanh Van et al. (1985) was used to isolate large quantities of bud tissue from plant material induced and not induced to flower. This work has enabled the isolation of genes expressed upon the induction of flowering in the cells of buds induced to flower in an in vitro cell system. Therefore, this study did not concentrate on purely meristematic tissue but involved the whole of the developing floral bud. The work of Medford et al. (1991) utilised the unusual growth habit of cauliflower, a member of the Brassica family. This plant is unusual in that the head of the cauliflower is a proliferation of shoot apical meristems. This allows for the isolation of large quantities of apical meristem tissue for cloning and subsequent characterisation of any clones of interest. This was of great advantage to Medford et al. (1991) in that cDNA libraries were

constructed by the conventional methods (Huynh *et al.*, 1985) which meant that they could be sure that their libraries were representative of the transcripts contained within the starting RNA material. PCR was only used to increase the quantity of cDNA used for differential screening. Several genes shown by northern blotting to be highly expressed in the shoot tips of cauliflower were introduced into *Arabidopsis* plants to study their expression (using the GUS assay) in a plant which does not develop in the unusual manner of the cauliflower plant. None of these genes was shown to be truly specific to the apical meristem but were also expressed in the shoots, roots, seedlings or flowers. Medford *et al.* (1991) also observed that one gene, termed meri-5, was expressed in different areas of the transformed *Arabidopsis* plant as the plant aged. Therefore, it may be the case that the apical tissue used in this study of sugar beet should have been of a younger age to allow the isolation of genes specific to the apical meristem. The isolation of meristematic genes by Medford *et al.* (1991) will enable future workers to assess the possibility that plants of a younger age may be more useful in the study of genes expressed specifically in the apical meristem tissue of flowering plants.

#### 4.3 Concluding Remarks

The introduction of molecular biological techniques to the process of understanding the mechanisms of flowering has advanced our knowledge of the genes involved in the morphology of flowering considerably. We do not know the mechanistic basis for fating of cells or tissues; however, in the flower organ formation we are beginning to understand how fate is acquired (Coen and Meyerowitz, 1991). Homeotic mutants have allowed the identification of gene products that appear to be transcriptional regulatory factors that directly, or indirectly instruct the cells of an organ primordium as to what organ to form. Whether or not this type of molecular mechanism is employed in the initiation of floral development is not known (McDaniel, 1992).

There are still those who believe that the florigen/antiflorigen theory holds true. The difficulty many scientists have in accepting this theory is that it assumes one process is acting in the same manner in all flowering plants. The hypothesis that the same factor(s) is (are) involved in the initiation of flowering in all flowering plant species has been criticised by many workers (reviewed in Bernier, 1988) who have observed that graft transmissible substances do not initiate flower development in all non-induced graft partners. It is not possible to obtain graft partnerships between every species of flowering plant and so only a limited number of plant species has been studied for the effects of graft transmission from induced to non-induced graft partner. However, of those species studied a minority have demonstrated the ability of non-related induced flowering plants to influence non-induced graft partners to flower. Therefore this unifying theory can only be applied to a select few species. The possibility that the concentration of nutrients in the apex increasing in low temperatures has something to do with the initiation of flowering is an assumption that many workers are considering (Salisbury, 1963; Atherton et al., 1987; Bernier, 1988). Since a substantial amount of data indicates that an increase in sugar content of low temperature treated apical meristems does occur, and under these conditions floral induction is observed, there is reason for investigating this hypothesis further. The need for an increase in nutrients is understandable if the apical meristem is to divide rapidly, and differentiation is to occur. For this to be the only factor involved in the switch from vegetative to floral induction appears improbable since these data only apply to flowering plants requiring vernalization

and the process of flowering is known to be influenced by many different factors (intrinsic and extrinsic). Many inhibitors of nucleic acid synthesis have been shown to inhibit flowering, which implies that this process is in some way linked to the regulation of flowering. It may be that the inhibition of nucleic acid synthesis disrupts housekeeping processes within the plant, and as such inhibits all further development of the plant. The belief from these studies is that floral induction requires the synthesis of new mRNAs specific to the process of flowering. Several workers have shown that under conditions necessary for floral induction in many plants nucleic acid, and protein synthesis is altered (reviewed by Bernier, 1988). Very few, however, have demonstrated that the new products are related to the development of flowers, and not related to other processes within the plant which change under inductive conditions. For example, under conditions of low temperature most species synthesise proteins associated with cold tolerance. These may play some part in the ability of plants to flower under the correct, subsequent, photoperiodic conditions, but serve a more general purpose in permitting the growth of the plant concerned under very low temperature conditions. Nucleic acid species known to be synthesised in LDP under LD conditions may well be associated with initiation of flowering, but to date it has not been possible to state that any one gene is associated solely with the induction of flowering, and not with some other associated process, such as an increased rate of photosynthesis. The association of a NB with the induction of flowering under SD conditions has been studied and shows that there are proteins synthesised during SDs which are not simply associated with the change in daylength, but are involved in the perception of photoperiodic induction of flowering (Lay-Yee et al., 1987). Therefore the implication that simple nutrient shifts are responsible for the change from vegetative to floral development does not take into account all of the evidence to date. There appears to be a lot more involved in this developmental

change than a hormonal, or nutrient shift. One explanation of the molecular mechanisms involved in vernalization and its effects on flowering may be that induction by low temperatures brings about changes in membrane permeability to a hormone previously synthesised under non-inductive conditions overcoming some barrier to the flowering process. However, at the same time the concentration of nutrients within the apical meristem must be sufficient for the increased cell division that is about to take place to commence. This would explain the observation of several workers that the removal of root tissue is detrimental to the flowering process. Alternatively, it may be the case that vernalization acts to allow the phosphorylation of a protein already present in the apical meristem cells of the plant and this phosphorylation alters the properties of the protein significantly to allow its action to be perceived by the apical meristem cells and induce the change from vegetative to floral development. This is all speculation as, to date, nobody has succeeded in isolating a gene associated with floral induction, neither has anybody succeeded in isolating the hormone(s) proposed by so many workers so long ago to be involved in the process of floral induction.

The idea that GA is involved in the induction of flowering is an old one, and it has been demonstrated that external applications of GA in some form or other do cause stem elongation in many flowering plants. However, for GA to be the hormone implicated in the florigen/antiflorigen theory is not acceptable since GA is only effective when applied to rosette LDP and a few other LD species. SDP, in general, do not respond to applications of GA. Again, however, there are exceptions as *Chrysanthemum morifolium* is a SDP which is affected by application of GA causing elongation of its stem without exposure to SD conditions. The general consensus is that GA plays a role in stem elongation, and can account for part of the vernalization requirement of many flowering plants, but GA alone is not the agent responsible for the transition from vegetative to floral development.

Although the methods used in this study have not proved to be successful in isolating any apex-specific cDNA clones they have raised some interesting questions. Does the apical meristem of sugar beet possess apex-specific mRNAs? The data of Medford et al. (1991) indicates that this should be the case although few other workers have yet managed to demonstrate this in other plant species. Work with Pharbitis nil (Keith Edwards, personal communication) has proved fruitless in the search for apex-specific cDNA clones, adding to the evidence that the isolation of apex-specific cDNA clones is a difficult process when tissue is limiting. The introduction of PCR has increased the success of attempts to clone cDNA from small amounts of both plant and animal tissue, but the characterisation of cDNA clones from libraries produced in this way still requires the isolation of large quantities of RNA from such tissues. Therefore, should methods of enhancing for apex-specific cDNA clones be explored? Subtractive screening and cloning have proved useful in the isolation of cDNA clones specific to certain tissues (Sive and St John, 1988) but were not successful in this study, possibly due to the sensitivity of PCR. It is hoped that the work of Medford et al. (1991) will lead to a better understanding of the expression of genes in the apical meristem of plant shoots, allowing the field of floral induction to be studied with a greater in depth knowledge of the subject.

Over the past decade advances in the field of plant molecular biology indicate that isolating a gene (or genes) for a process whereby the gene product is not known may be best accomplished using a method of gene-tagging. The most extensively studied method is that of transposon-tagging which is a process originally described for work with *Drosophila* (Bingham *et al.*, 1981). A few years later gene tagging experiments were reported in plants, the first of which was the successful cloning of the *bz* locus of *Zea mays* using a transposable element known as Ac (Activator) (Fedoroff et al., 1984).

Transposable elements, or transposons, are mobile segments of DNA which inactivate, or alter gene expression by insertion into the genetic locus. Transposon-induced mutations have been described in a few plant species and a few plant transposable elements have been cloned (mostly from Zea mays). All of the early transposable elements isolated from Zea mays were associated with characteristics of the Zea mays kernel which is easily scored. This is an important factor in the isolation of genes by gene-tagging.

In Zea mays some elements are capable of self encoded excision and transposition. Such elements are termed autonomous. Ac is an autonomous transposable element isolated from Zea mays and is capable of catalysing its own excision and transposition. Another element, Ds, (which resembles the terminal sequence of the Ac element) is non-autonomous and requires an active Ac element in the genome before it will move. As has been demonstrated for Zea mays it is useful to choose a target gene that can be recognised at the seed or seedling level. In a genetic cross to attempt to isolate a gene of interest one parent plant would carry an active transposon and be homologous for the wild type of the target gene while the other carries the homozygous stable recessive allele of the target gene. Thus, in the F1 progeny a wild type phenotype would be expected. If the sample size is large enough a few exceptional progeny not exhibiting the wild type phenotype may be found, suggesting that something has occurred to alter the wild type allele of the heterozygote. This phenotype may reflect the insertion of the active transposon into the wild type allele of the target gene.

To establish that a transposable element has caused a mutation of the target gene the transposon must be used as a hybridisation probe to detect pieces of DNA harbouring the transposon.

158

It is also possible to introduce transposable elements into plant species by transformation as a viable alternative to isolating endogenous transposons. This would require the ability to stably transform the plant species of interest which, to date, has not proved possible for sugar beet, but such species as tobacco (Baker *et al.*, 1986), *Arabidopsis* and carrot (Van Sluys *et al.*, 1987), have been stably transformed with transposable elements and interesting genes isolated by gene-tagging. Recently a male sterility gene of *Arabidopsis* was isolated using the *Zea mays Enhancer-inhibitor* transposable element system (Aarts *et al.*, 1993) and a *Petunia* flower colour gene was isolated with the use of the *Zea mays* transposable element *Ac* (Chuck *et al.*, 1993).

This all bodes well for the field of plant molecular biology as it seems to be that many new and interesting genes are being isolated by gene-tagging but the procedure is limited to those few species known to contain transposable elements, or easily transformed with such elements. To apply this methodology to sugar beet would require stable transformation. Also, since we would be looking for a phenotypic change involving flowering the progeny of any experiment would need to be grown up for a long period of time, taking up a large amount of space, to isolate a few putative mutants tagged by the transposon at the gene of interest. Using a model system like *Arabidopsis* to look for genes involved in the induction of flowering using gene-tagging may be more appropriate and successful since this plant species is easily transformable with transposable elements and has a rapid generation time. It is quite probable that genes involved in the induction of flowering in *Arabidopsis* are also involved in the same process in sugar beet. If gene-tagging can prove successful in isolating such genes from some other plant species these genes may be used to probe a sugar beet library provided the conservation of floral inductive genes is high between highly unrelated plant species.

# REFERENCES

Aarts, M.G.M., W.G. Dirkse, W.J. Stiekema and A. Pereira (1993). Transposon-tagging of a male sterility gene in *Arabidopsis*. Nature 363: 715-717

Abegg, F.A. (1936). A genetic factor for the annual habit in beets and linkage relationship. J. Ag. Res. 53: 493-511

Akowitz, A. and L.Manuelidis (1989). A novel cDNA/PCR strategy for efficient cloning of small amounts of undefined RNA. Gene 81: 295-306

Al Mallah, M.K., M.R.Davey and E.C.Cocking (1989). A new approach to the nodulation of non-legumes by *Rhizobia* and the transformation of cereals by *Agrobacterium* using enzymatic treatment of root hairs. International Journal of Genetic Manipulation of Plants 5: 1-7

Arzee, T., J.Gressel and E.Galun (1970). Flowering in *Pharbitis*. The influence of actinomycin D on growth, incorporation of nucleic acid precursors and autoradiographic patterns. In, Cellular and Molecular Aspects of Floral Induction. (Ed. G.Bernier): 93-107. Longman, London.

Atherton, J.G., D.J.Hand and C.A.Williams, (1987). Curd initiation in the cauliflower (*Brassica oleracea* var Botrytis L.). In, Manipulation of Flowering (Ed. J.G.Atherton): 133-145. Butterworths, London.

Baker, B. J. Schell, H. Loerz and N. Fedoroff (1986). Transposition of the maize controlling element "Activator" in tobacco. Proc. Natl. Acad. Sci. USA. 83: 4844

Baldev, B. (1959). In vitro responses of growth and development in *Cuscuta reflexa* Roxb. **Phytomorphology 9: 316-319** 

Belyavsky, A., T. Vinogradova and K.Rajewsky (1989). PCR based cDNA library construction: general cDNA libraries at the level of a few cells. Nucl. Acids Res. 16: 2919-2932

Benton, W.D. and R.W.Davies (1977). Screening  $\lambda$  gt recombinant clones by hybridisation to single plaques *in situ*. Science 196: 180-182

Bernier, G., J-M.Kinet and R.M.Sachs (1981). The initiation of flowers. In, The Physiology of Flowering Vol 1: 70-126. CRC, Boca Raton.

Bernier, G. (1988). The control of floral evocation and morphogenesis. Ann. Rev. Plant Physiol. Plant Mol. Biol. 39: 175-219

Bingham, P.M., R. Levis and G.M. Rubin (1981). Cloning of DNA sequences from the white locus of *Drosophila melanogaster* by a novel and general method. Cell. 25: 693

Boll, W., J-I.Fujisawa, J.Niemi and C.Weissmann (1986). A new approach to high sensitivity differential hybridisation. Gene 50: 41-53

Borthwick, H.A. and M.W.Parker (1938 a). Effectiveness of photoperiodic treatments of plants of different age. Bot. Gaz. 100: 245-249

Borthwick, H.A. and M.W.Parker (1938 b). Photoperiodic perception in biloxi soy beans. Bot. Gaz. 100: 374-387

Borthwick, H.A., S.B.Hendricks, M.W.Parker, E.H.Toole and V.K.Toole (1952). A reversible photoreaction controlling seed germination. **Proc. Natl. Acad. Sci. USA. 38: 662-666** 

Campbell, G.K.G. (1953). Selection of sugar beet for resistance to bolting. Publications of the Institute of International Research Better. XVI Winter Congress: 5

Danyluck, J. and F.Sarhan (1990), Differential mRNA muscription during the induction to

Chailakhyan, M.Kh. (1937) Concerning the hormonal nature of plant development. C.R. (Donkl) Acad. Sci. URSS 3:442

Chailakhyan, M.Kh. (1938). Motion of blossom hormone in girdled and grafted plants. C.R. (Donkl) Acad. Sci. URSS 18: 607-612

Choroboczek, E. (1934). A study of the ecological factors influencing seed stalk development in beets (*Beta vulgaris*). Agri. Exp. Sta. Mem. 154: 1-84

conting amplification A sal Righton 188-427

Chouard, P. (1960). Vernalization and its relations to dormancy. Ann. Rev. Plant Physiol. 11: 191-238

Chuck, G., T. Robbins, C. Nijjar, E. Ralston, N. Courtney-Gutterson and H. K. Dooner (1993). Tagging and cloning of a Petunia flower color gene with the maize transposable element *Activator*. Plant Cell 5: 371-378

Cocking, E.C. (1990). All sorts of genetic manipulation. In. Genetic Engineering of Crop Plants. (Ed. G.W.Lycett and D.Grierson): 1-12. Butterworths, London.

Coen, E.S. and E.M.Meyerowitz (1991). the war of the whorls. Genetic interactions controlling flower development. Nature 353: 31-37

Cuming, A.C. and J.Bennett (1981). Biosynthesis of the light harvesting chlorophyll *a/b* binding protein. Control of mRNA activity by light. Eur. J. Biochem. 118: 71-80

Curth, P. (1960). Der ubergang in die reproductive phase bei dir zuckerrübe in abhängigkeit von verschiednen umweltfactoren. Biet Zur Rübenforschung nr. 4 Deutsch Akademie der landwirtschaftswissenschaften zu Berlin.

Curtis, O.F. and H.T.Chang (1930). The relative effectiveness of the temperature of the crown as contrasted with the rest of the plant upon flowering of celery plants. Am. J. Bot. 17: 1047-1048

Curtis, G.J. and K.G.Hornsey (1964). Graft transmission of the flowering stimulus from a wild *Beta* species to a line of beet selected for resistance to bolting. Nature 203: 201-202

Danyluck, J. and F.Sarhan (1990). Differential mRNA transcription during the induction to freezing tolerance in spring and winter wheat. Plant Cell Physiol. 31: 609-619

De Block, M., J.Botterman, M.Vandeweile, J.Dockx, C.Thoen, V.Gosselé, N.R.Movva, C.Thompson, M.Van Montague and J.Leemans (1987). Engineering herbicide resistance in plants by expression of a detoxifying enzyme. EMBO J. 6: 2513-2518

Domec, C., B.Garbay, M.Fournier and J.Bonnet (1990). cDNA library construction from small amounts of unfractioned RNA: Association of cDNA synthesis with polymerase chain reaction amplification. Anal. Biochem. 188: 422-426

Drews, G.N. and R.B.Goldberg (1989). Genetic control of flower development. Trends in Genetics 5: 256-261

Ecker, J.R. and R.W.Davis (1986). Inhibition of gene expression in plant cells by expression of antisense RNA. Proc. Natl. Acad. Sci. USA. 83: 5372-5376

Erlich, H.A., D.Gelfand and J.J.Sninsky (1991). Recent advances in the polymerase chain reaction. Science 252: 1643-1650

Fedoroff, N.V., D.B. Furtek and O.E. Nelson (1984). Cloning of the bronze locus in maize by a simple and generalizable procedure using the transposable controlling element Activator (Ac). Proc. Natl. Acad. Sci. USA. 3825

Fife, J.M. and C.Price (1953). Bolting and flowering of sugar beets in continuous darkness. Plant Physiol. 28: 4755-480

Gardner, F.P. and W.E.Loomis (1953). Floral induction and development in orchard grass. Plant Physiol. 28: 201-217

Garner, W.W. and H.A.Allard (1920). Effect of length of day on plant growth. J. Agri. Res. 18: 553-606

Gaskill, J.O. (1957). A preliminary report on the use of gibberellic acid to hasten reproductive development in sugar beet seedlings. J. Am. Soc. for Sugar Beet Technol. 9: 521-528

Gassner, G (1918). Beiträge Zur physiologischen charakteristik sommer-und-winterannualler Gewächse insbesondere der Getreidepflazen. Zeit. Bot. 10: 417

Gonthier, R., A.Jacqmard and G.Bernier, (1987). Changes in cell cycle duration and growth fraction in the shoot meristem of *Sinapis* during floral transition. **Planta 170: 55-59** 

Gregory, F.G. and O.N.Purvis (1936). Devernalization of winter rye by high temperature. Nature 138: 1013

Gressel, J., J.Zilberstein, J.A.Porath and D.Arzee (1980). Demonstration with fibre illumination that *Pharbitis* plumules also perceive flowering photoinduction. In, Light and

Flowering Processes. (Ed. D.Vince-Prue, B.Thomas and K.Cockshull): 257-275. Academic Press, London.

Gubler, U. and B.J.Hoffman (1983). A simple and very efficient method for generating cDNA libraries. Gene 25: 263-269

Hamner, K.C. (1940). Interrelation of light and darkness in photoperiodic induction. Bot. Gaz. 101: 658-687

Heery, D.M., F.Gannon and R.Powell (1990). A simple method for subcloning DNA fragments from gel slices. Trends in Genetics 6: 173

Heide, O.M. (1973). Environmental control of bolting and flowering in red garden beets. Scientific Reports of the Agricultural University of Norway 52: 1-17

Hughes, M.A. and R.S.Pearce (1988). Low temperature treatment of barley plants causes altered gene expression in shoot meristems. J. Exp. Bot. 39: 1461-1467

Hussey, G. (1963). Growth and development in young tomato II. The effect of defoliation on the development of the shoot apex. J. Exp. Bot. 14: 326-333

Huynh, T.V., R.A. Young and R.W.Davis (1985). Construction and screening of cDNA libraries in λgt 10 and λgt 11. In, DNA Cloning Volume 1: A Practical Approach (Ed. D.M.Glover): 79-88. IRL Press

Jaffe, M.J., K.A.Bridle and J.Kopcewicz (1987). A new strategy for the identification of native plant photoperiodically regulated flowering substances. In, Manipulation of Flowering. (Ed. J.G.Atherton): 279-287. Butterworths, London.

Jepson, I., J.Bray, G.I.Jenkins, W.Schuch and K.J.Edwards (1991). A rapid procedure for the construction of PCR amplified cDNA libraries from small amounts of plant tissue. **Plant** Mol. Biol. Reporter 9: 132-138

Kelly, A.J., M.T.Zagotta, R.A.White, C.Chang and D.R.Meeks-Wagner (1990). Identification of genes expressed in the tobacco shoot apex during the floral transition. Plant Cell 2: 963-972 Khudairi, A-K. and K.C.Hamner (1954). The relative sensitivity of Xanthium leaves of different ages to photoperiodic induction. Plant Physiol. 29: 251-257

Kilppart, J.H (1857). An essay on the origin of growth, disease, varieties, etc within the wheat plant. Ohio state Bd. Agri. Ann. Rep. 12, 562.

Kohli, R.K., N.Sawhney and S.Sawhney (1980). Photo-induced changes in proteins associated with floral induction in *Amaranthus*. Plant & Cell Physiol. 21: 1483-1490

Lane, H.C., H.M.Cathey and L.T.Evans (1965). The dependance of flowering in several long-day plants on the spectral composition of light extending the photoperiod. Am. J. Bot. 52: 1006-1014

Lang, A. (1952). Physiology of flowering. Ann. Rev. Plant Physiol. 3:256-306

Lang, A. (1965). Physiology of flower initiation. In, Encyclopedia Plant Physiol. Vol 15, Part 1. (Ed. W.Ruhland): 1379-1536. Springer Verlag, Berlin

Lasa, J.M. and C.Pérez Peña (1978). Light intensity and photoperiod in sugar beet, *Beta* vulgaris L. An. Aula. Dei. 14: 181-187

Law, C.N. (1987). The genetic control of daylength response in wheat. In, Manipulation of Flowering. (Ed. J.G.Atherton): 225-239. Butterworths, London.

Lay-Yee, M., R.M.Sachs and M.S.Reid (1987). Changes in cotyledon mRNA during floral induction of *Pharbitis nil* cv. violet. **Planta 171: 104-109** 

Lexander, K. (1975). Bolting susceptability of sugar beet (*Beta vulgaris*) in relation to contents of sulfhydryls and disulfides and to protein composition of membranes. **Physiol. Plant. 33:142-150** 

Lexander, K. (1985). *BETA VULGARIS*. In, Handbook of Flowering Vol 2 (Ed. A.H. Halvey) : 24-32. Boca Raton;CRC.

Lexander, K. (1987). Characters related to the vernalisation requirement of sugar beet. In, Manipulation of Flowering (Ed. J.G.Atherton): 147-158. Butterworths, London. Lu, X. and D.Werner (1988). Construction and quality of cDNA libraries prepared from cytoplasmic RNA not enriched in poly(A)<sup>+</sup>RNA. Gene 71: 157-164

Magara, J. (1960). Recherches sur le déterminism de l'elongation et de la floraison dans le genre *Beta*. Thése. Paris

Magara, J. (1964). Discussion sur les relations entre élongation caulinaire et initiation florale chez la betterave bisannualle *Beta vulgaris* L. Ann. Physiologie Vég. 6: 243-250

Martinez-Zapater, J.M., and C.R.Somerville (1990). Effect of light quality and vernalization on late flowering mutants of *Arabidopsis thaliana*. Plant Physiol. 92: 770-776

McDaniel, C.N. (1985). Competence, determination, and induction in plant development. In, **Pattern Formation (Ed. G.M.Malaakinski): 393-412** MacMillan: N.Y.

McDaniel, C.N. (1991). Early events in flowering. A black box with a bright future. Plant Cell 3: 431-433

McDaniel, C.N. (1992). Induction and determination: developmental concepts. Flowering Newsletter 14: 3-6

McDaniel, C.N., R.W.King and L.T.Evans (1991). Floral determination and *in vitro* floral differentiation in isolated shoot apices of *Lolium tumulentum* L. Planta 185: 9-16

Medford, J.I., J.S.Elmer and H.J.Klee (1991). Molecular cloning and characterisation of genes expressed in shoot apical meristems. Plant Cell 3: 359-370

Meeks-Wagner, D.R., E.S.Dennis, K.Tran Than Van and W.J.Peacock (1989). Tobacco genes expressed during *in vitro* floral initiation and their expression during normal plant development. **Plant Cell 1: 25-35** 

Melchers, G. and A.Lang (1948). Die physiologie der blüten bildung. (Ein Übersichtsbericht). Biol. Zbl. 67: 105-174

Meltzer, S., D.M.Majewski and K.Apel (1990). Early changes in gene expression during the transition from vegetative to generative growth in the long-day plat *Sinapis alba*. **Plant Cell 2: 953-961** 

Meza-Basso, L., M.Alberdi, M.Raynal, M-L.Ferrero-Cadinanos and M.Delseny (1986). Changes in protein synthesis in rapeseed (*Brassica napus*) seedlings during low temperature treatment. **Plant Physiol. 82: 733-738** 

Moskov, B.S. (1936). Role of leaves in photoperiodic reaction of plants. Bull. Appl. Bot. Genet. and Plant Breeding. A17: 25

Murneek, A.E. and R.O.Whyte (1948). Vernalization and Photoperiodism. Chronica Botanica. Waltham, Mass.

Murfet, I.C. (1977). Environmental interaction and the genetics of flowering. Ann. Rev. Plant Physiol. 28: 253-278

Murfet, I.C. (1989). Flowering genes in *Pisum*. In **Plant Reproduction, From Floral Induction to Pollination. (Ed. E.Lord and G.Bernier): 10-18**. American Society of Plant Physiologists, Rockville.

Murfet, I.C. and J.B.Reid (1987). Flowering in *Pisum*: Gibberellins and the flowering genes. J. Plant Physiol. 127: 23-29

Nanda, K.K., H.N.Kirrishnamoorthy, T.A.Anuradha and K.Lal (1967). Floral induction by gibberellic acid in *Impatiens balsamina*, a qualitative short day plant. **Planta 76: 367-370** 

Napp-Zinn, K. (1985). ARABIDOPSIS THALIANA. In, Handbook of Flowering, Vol 1 (Ed. A.H.Halvey): 429-503. Boca Raton, CRC

Naylor, A.W. (1941). Effects of nutrition and age upon rate of development of terminal staminate inflorescences of *Xanthium pennsylvanicum*. Bot. Gaz. 103: 342-353

Newberry, J., G.A.Todd, I.Goodwin and B.V.Ford-Lloyd (1989). Designer beet-The impact of genetic engineering on sugar beet. British Sugarbeet Review 57: 41-45

Nougaréde, A., J.Rembur, D.Francis and P.Rondet (1989). Ageing of *Silene coeli-rosa* L. shoot apex under non inductive conditions: changes in morphology, mitotic index, and polypeptide composition. **Protoplasma 153: 30-36** 

Pharis, R.P. and R.W. King (1985). Gibberellins and reproductive development in seed

#### plants. Ann. Rev. Plant Physiol. 36: 517-568

Poa, C-I. and P.W.Morgan (1986). Genetic regulation of development in Sorghum bicolor. II. The effect of MA<sub>3</sub><sup>R</sup> allele mimicked by GA<sub>3</sub>. Plant Physiol. 82: 581-584

Pollock, C.J. and C.F.Eagles (1988). Low temperature and the growth of plants. In, Plants and Temperature (Ed. S.P.Long and F.I.Woodward): 157-180. Company of Biologists, Cambridge.

Purvis, O.N. (1939). Studies in vernalization of cereals. V. The inheritance of spring and winter habit in hybrids of Petkus rye. Ann. Bot. III: 315-325

Purvis, O.N. (1948). Studies in vernalization XI. The effect of date of sowing and of excising the embryo upon the response of Petkus rye to different periods of vernalization treatment. Ann. Bot. XII: 183-206

Radley, M.E. (1975). Sugar beet, growth substances and bolting. In, Whitingham, C.P., Botany Dept. Rothamsted Report for 1974 Part 1.

Reid, J.B. (1986). Gibberellin mutants. In, A Genetic Approach to Plant Breeding. Plant Gene Research Vol 3. (Ed. P.J.King and A.D.Blonstein): 1-34. Springer-Verlag, Vienna.

Réide, G.P. (1975). In, Genetic Manipulations with Plant Material. (Ed. L.Ledoux): 183. Plenum press, New York.

Ross, C. (1970). Antimetabolite studies and the possible importance of leaf protein synthesis during induction of flowering in the cocklebur. In, Cellular and Molecular Aspects of Floral Induction. (Ed. G.Bernier): 139-151. Longman, London.

Sachs, R.M. and W.P.Hackett (1983). Source sink relationships and flowering. In, Strategies of Plant Reproduction. (Ed. W.J.Meudt): 263-272. Allenhead, Osmund.

Saiki, R.K., S.Scharf, F.Faloona, K.B.Mullis, G.T.Horn, H.A.Erlich and N.Arnheim (1985). Enzymatic amplification of  $\beta$ -globin genomic sequences and restriction site analysis for diagnosis of sickle cell anemia. Science 230: 1350-1354

Sakar, S., (1958). Versuche zur physiologie der vernalization. Biol. Zbl. 77: 1-49

Salisbury, F.B. (1955). The dual role of auxin in flowering. Plant Physiol. 30: 327-334

Salisbury, F.B. (1963). The Flowering Process. Pergamon Press, Oxford.

Sambrook, J., E.F.Fritsch and T.Maniatis (1988). Molecular Cloning: A Laboratory Manual, 2nd Ed. Cold Spring Harbor Laboratory Press, New York.

Schöffl, F. (1988). Genetic engineering strategies for the manipulation of the heat shock response. Plant, Cell and Environment 11: 339-343

Schultz, E.A., F.B.Pickett and G.W.Haughn (1991). The FLO 10 gene product regulates the expression domain of homeotic genes AP 3 and PI in Arabidopsis flowers. Plant Cell 3: 1221-1237

Schwabe, W.W. (1987). The flowering problem. In, Manipulation of Flowering, (Ed. J.G.Atherton): 3-12. Butterworths, London.

Scott, R., R.Hodge, W.Paul and J.Draper (1991). The molecular biology of anther differentiation. Plant Sci. 80: 167-191

Sive, H.L. and T. St John (1988). A simple subtractive hybridisation technique employing photoactivatable biotin and phenol extraction. Nucl. Acids Res. 16: 10937

Snyder, F.W. and S.H.Wittwer (1959). Some effects of gibberellin on stem elongation and flowering in sugar beets. J. Am. Soc. of Sugar Beet Technol. 10: 553-561

Sommer, H., J.P.Beltrán, P.Huijser, H.Pape, W-E. Lönnig, H.Saedler and Z.Schwartz-Sommer (1990). *Deficiens*, a homeotic gene involved in the control of flower morphogenesis in *Antirrhinum majus*; the protein shows homology to transcription factors. **EMBO J. 9**: **605-613** 

Southern, E. (1975). Detection of specific sequences among DNA fragments separated by gel electrophoresis. J. Mol. Biol. 98: 503

Sprent, J.I. (1989). Tansley review  $N^{\underline{O}}$ . 15. Which steps are essential for the formation of functional legume nodules ? New Phytol 111: 129-153

Steeves, T.A. and I.M.Sussex (1989). Patterns in Plant Development. 2nd Ed. Cambridge University Press, Cambridge.

Stout, M. (1945). Translocation of the reproductive stimulus in sugar beets. Bot. Gaz. 107: 86-95

Stout, M. (1959). Some effects of gibberellic acid on the physiology of sugar beets. J. Am. Soc. of Sugar Beet Technol., 10: 305-309

Sussex, I.M. (1989). Developmental programming of the shoot meristem. Cell 56: 225-229

Terry, N. (1968). Developmental physiology of sugar beet I. The influence of light and temperature on growth. J. Exp. Bot. 19: 795-811

Timblin, C., J.Battey and W.M.Kuehl (1990). Application of PCR technology to subtractive cDNA cloning: Identification of genes expressed specifically in murine plasmacytoma cells. Nucl. Acids Res. 18: 1587-1593

Tran Thanh Van, K., P.Toubart, A.Cousson, A.G.Darvill, D.J. Gollin. P.Chelf and P.Albersheim (1985). Manipulation of the morphogenic pathways of tobacco explants by oligosaccharins. Nature 314: 615-617

Van Sluys, M.A., J.Temple and N.Fedoroff (1987). Studies on the introduction and mobility of the maize Activator element in Arabidopsis thaliana and Daucus carota. EMBO J. 6: 3881-3890

Vince-Prue, D. (1975). Photoperiodism in Plants. Chapter 7, Vernalization: 262-291. McGraw Hill, London.

Wareing, P.F. and I.D.J. Phillips (1978). The Control of Growth and Differentiation in Plants, 2nd Ed. Pergamon, Oxford.

Warm, E. (1984). Changes in the composition of *in vitro* translated mRNA caused by photoperiodic flower induction of *Hyoscyamus niger*. Physiol. Plant. 61: 344-350

Wellensiek, S.J. (1958). Vernalization and age in *Lunaria biennsis*. K. Ned. Acad. Wetsenchap Proc. C61: 561

Wellensiek, S.J. (1964). Dividing cells as a prerequisite to vernalization. Plant Physiol. 39: 832-835

Williams, C.A. and J.G.Atherton (1990). A role for young leaves in vernalisation of cauliflower: I. Analysis of leaf development during curd induction. Physiol. Plant. 78: 61-66

Withrow, A., R.T.Withrow and J.P.Bieble (1943). Inhibiting influence of leaves on photoperiodic response of noble spinach. Plant Physiol. 18: 294-298

Zeevart, J.A.D. (1958). Flower formation as studied by grafting. Mededel Landbouwhogescool 58: 88

Zeevart, J.A.D. (1962). DNA multiplication as a requirement for expression of floral stimulus in *Pharbitis nil*. **Plant Physiol. 37: 296-304** 

Zeevart, J.A.D. (1963). Climatic control of reproductive development. In, Environmental control of Plant Growth. (Ed. L.T.Evans): 298-310. Academic Press, New York.

Zeevart, J.A.D. (1976). Physiology of flower formation. Ann. Rev. Plant Physiol. 27: 321-348

Zeevart, J.A.D., J.M.Brede and L.B.Cetas (1977). Translocation patterns in *Xanthium* in relation to long-day inhibition of flowering. Plant Physiol. 60: 747-753

Zeevart, J.A.D. and A.Lang (1962). The relationship between gibberellin and floral stimulus in *Bryophyllum daigremontiantum*. Planta 58: 531-542



171